

SCIENTIFIC JOURNAL OF WOOD TECHNOLOGY



ZNANSTVENI ČASOPIS ZA PITANJA DRVNE TEHNOLOGIJE

Populus spp.

UDK 674.031.677.7 ISO: Drv. Ind. CODEN: DRINAT JCR: DRVNA IND ISSN 0012-6772



since 1913





# SCIENTIFIC JOURNAL OF WOOD TECHNOLOGY

Znanstveni časopis za pitanja drvne tehnologije

# PUBLISHER AND EDITORIAL OFFICE

Izdavač i uredništvo

University of Zagreb Faculty of Forestry and Wood Technology Sveučilište u Zagrebu Fakultet šumarstva i drvne tehnologije www.sumfak.unizg.hr

## CO-PUBLISHER / Suizdavač

Hrvatska komora inženjera šumarstva i drvne tehnologije

**FOUNDER** / Osnivač Institut za drvnoindustrijska istraživanja, Zagreb

#### EDITOR-IN-CHIEF

Glavna i odgovorna urednica Ružica Beljo Lučić

ASSISTANT EDITOR-IN-CHIEF

Pomoćnik glavne urednice Josip Miklečić

#### EDITORIAL BOARD / Urednički odbor

Vlatka Jirouš-Rajković, Hrvatska Iva Ištok, Hrvatska Zoran Vlaović, Hrvatska Andreja Pirc Barčić, Hrvatska Nikola Španić, Hrvatska Miljenko Klarić, Hrvatska Tomislav Sedlar, Hrvatska Maja Moro, Hrvatska Matija Jug, Hrvatska Ivana Perić, Hrvatska Christian Brischke, Germany Zeki Candan, Turkey Julie Cool, Canada Katarina Čufar, Slovenia Lidia Gurau, Romania Vladislav Kaputa, Slovak Republic *Robert Nemeth, Hungary* Leon Oblak. Slovenia Kazimierz Orlowski, Poland Hubert Paluš, Slovak Republic Marko Petrič, Slovenia Jakub Sandak, Slovenia Jerzy Smardzewski, Poland Aleš Straže, Slovenia Eugenia Mariana Tudor, Austria

#### **PUBLISHING COUNCIL**

Izdavački savjet president – predsjednik izv. prof. dr. sc. Vjekoslav Živković

prof. dr. sc. Ružica Beljo Lučić, prof. dr. sc. Vladimir Jambreković, Fakultet šumarstva i drvne tehnologije Sveučilišta u Zagrebu; dr. sc. Dominik Poljak, Drvodjelac d.o.o.; Silvija Zec, dipl. ing. šum., Hrvatska komora inženjera šumarstva i drvne tehnologije

#### **TECHNICAL EDITOR**

Tehnički urednik Zoran Vlaović

# ASSISTANT TO EDITORIAL OFFICE

Pomoćnica uredništva Dubravka Cvetan

## LINGUISTIC ADVISERS

Lektorice English – engleski Maja Zajšek-Vrhovac, prof.

**Croatian** – hrvatski Zlata Babić, prof.

The journal Drvna industrija is a public scientific journal for publishing research results on structure, properties and protection of wood and wood materials, application of wood and wood materials, mechanical woodworking, hydrothermal treatment and chemical processing of wood, all aspects of wood materials and wood products production and trade in wood and wood products. The journal is published quarterly and financially supported by the Ministry of Science and Education of the Republic of Croatia

Časopis Drvna industrija javno je znanstveno glasilo za objavu rezultata istraživanja građe, svojstava i zaštite drva i drvnih materijala, primjene drva i drvnih materijala, mehaničke i hidrotermičke obrade te kemijske prerade drva, svih aspekata proizvodnje drvnih materijala i proizvoda te trgovine drvom i drvnim proizvodima. Časopis izlazi četiri puta u godini uz financijsku potporu Ministarstva znanosti i obrazovanja Republike Hrvatske.

# Contents Sadržaj

#### CIRCULATION: 400 pieces

**INDEXED IN:** Science Citation Index Expanded, Scopus, CAB Abstracts, Compendex, Environment Index, Veterinary Science Database, Geobase, DOAJ, Hrčak, Sherpa Romeo

#### MANUSCRIPTS ARE TO BE SUBMITTED by the link http://journal.sdewes.org/drvind

CONTACT WITH THE EDITORIAL e-mail: editordi@sumfak.hr

SUBSCRIPTION: Annual subscription is 55 EUR. For pupils, students and retired persons the subscription is 15 EUR. Subscription shall be paid to the IBAN HR0923600001101340148 with the indication "Drvna industrija".

PRINTED BY: DENONA d.o.o., Getaldićeva 1, Zagreb, www.denona.hr

DESIGN: Bernardić Studio

#### THE JOURNAL IS AVAILABLE ONLINE: https://drvnaindustrija.com

COVER: Cross-sectional microscopic view of Populus spp., xylotheca of Institute for Wood Science, University of Zagreb Faculty of Forestry and Wood Technology

DRVNA INDUSTRIJA · VOL. 74, 4 · P. 393-504 · WINTER 2023. · ZAGREB EDITORIAL COMPLETED 1. 12. 2023.

#### NAKLADA: 400 komada

ČASOPIS JE REFERIRAN U: Science Citation Index Expanded, Scopus, CAB Abstracts, Compendex, Environment Index, Veterinary Science Database, Geobase, DOAJ, Hrčak, Sherpa Romeo

ČLANKE TREBA SLATI putem poveznice http://journal.sdewes.org/drvind

KONTAKT S UREDNIŠTVOM: e-mail: editordi@sumfak.hr

PRETPLATA: Godišnja pretplata za pretplatnike u Hrvatskoj i inozemstvu iznosi 55 EUR. Za đake, studente i umirovljenike 15 EUR. Pretplata se plaća na IBAN HR0923600001101340148 s naznakom "Drvna industrija".

TISAK: DENONA d.o.o., Getaldićeva 1, Zagreb, www.denona.hr

DESIGN: Bernardić Studio

**ČASOPIS JE DOSTUPAN NA** INTERNETU: https://drvnaindustrija.com

NASLOVNICA: Mikroskopska slika poprečnog presjeka drva Populus spp., ksiloteka Zavoda za znanost o drvu, Sveučilište u Zagrebu Fakultet šumarstva i drvne tehnologije

DRVNA INDUSTRIJA · VOL. 74, 4 · STR. 393-504 · ZIMA 2023. · ZAGREB REDAKCIJA DOVRŠENA 1. 12. 2023.

# **ORIGINAL SCIENTIFIC PAPERS**

Izvorni znanstveni radovi
The Effect of Hot and Cold Check Tests on Surface Roughness and Glossiness in Varnished Wood Material
Utjecaj testa vruće i hladne provjere na hrapavost i sjaj površine
lakiranog drva
Abdi Atilgan, Sait Dundar Sofuoglu
Wood Species Image Classification Using Two-Dimensional Convolutional Neural Network
Klasifikacija vrsta drva prema slikama uz pomoć dvodimenzionalne
konvolucijske neuronske mreže
Kenan Kilic, Kursat Kilic, Brian Bino Sinaice, Ugur Ozcan
Modeling of Color Design on Furniture Surfaces with CNC Laser Modification
Modeliranje dizajna boje površine namještaja CNC laserskom modifikacijom
Cebrail Açık
Timber Strength Grading as Necessary Basis for Structural Design in Ex-YU Region: Part 1
Ocjenjivanje drva prema čvrstoći kao nužna osnova za projektiranje
konstrukcija na području bivše Jugoslavije: dio 1.
Tatjana Kočetov Mišulić, Aleksandra Radujković, Zdravko Popović, Ksenija Hiel
Comparison of Reaction Wood and Normal Wood of Some Commercial Tree Species
Usporedba reakcijskoga i normalnog drva nekih komercijalnih vrsta drva
Sevinç Kaz, Saim Ateş, Tuba Külçe
Thermal Modification Intensity of Heat-treated Poplar Wood Part 1: Characterization and Predication of Surface Layer Intenzitet toplinske modifikacije topolovine. Dio 1: Karakterizacija i predikcija površinskog sloja Yang Li, Tao Yao, Yong Zhu, Shengquan Liu, Zuju Shu, Redžo Hasanagić,
Leila Fathi, Demiao Chu
dryno-nlastičnih kompozita
Sefa Durmaz, Ugur Aras, Erkan Avci, Yusuf Ziya Erdil, Ilkay Atar, Hulya Kalaycioglu
Bio-Durability and Engineering Characteristics of Heat-Treated Poplar Wood
Biološka trajnost i tehnička svojstva toplinski modificirane topolovine
Seyedeh Masoomeh Zamani, Reza Hajihassani, Saman Ghahri
Influence of Pre-Impregnation Acetic Anhydride and Heating Time on Mechanical and Physical Properties of Wood-Plastic Composites
Utjecaj predimpregnacije anhidridom octene kiseline i vremena
zagrijavanja na mehanička i fizička svojstva drvno-plastičnih kompozita
Seyyed Khalil Hosseinihashemi, Mohammad-Ali Akhoundi, Younes Shirmohammadli,
Nadir Ayriimis
Effect of Epoxy Resin Reinforcement on Screw Withdrawal Strength of Fiberboard and Particleboard Used in Furniture Industry Utjecaj ojačanja epoksidnom smolom na izvlačnu silu vijka u ploča
vlaknatica i iverica koje se rabe u proizvodnji namještaja
Mesut Uysal, Cagatay Tasdemir, Dogan Memis

## 

Abdi Atilgan<sup>1</sup>, Sait Dundar Sofuoglu<sup>2</sup>

# The Effect of Hot and Cold Check Tests on Surface Roughness and Glossiness in Varnished Wood Material

# Utjecaj testa vruće i hladne provjere na hrapavost i sjaj površine lakiranog drva

## **ORIGINAL SCIENTIFIC PAPER**

**Izvorni znanstveni rad** Received – prispjelo: 16. 1. 2023. Accepted – prihvaćeno: 22. 8. 2023. UDK: 620.179.118; 620.191.5; 630\*84 https://doi.org/10.5552/drvind.2023.0085 © 2023 by the author(s). Licensee University of Zagreb Faculty of Forestry and Wood Technology. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license.

ABSTRACT • In this study, specimens from Scots pine (Pinus sylvestris L.), Anatolian chestnut (Castanea sativa Mill.) and Eastern beech (Fagus orientalis Lipsky) tree species, prepared according to ISO 3129, were conditioned according to TS ISO 13061-1. Cellulosic varnish, water-based varnish, and glass polish varnish were applied to their surfaces in accordance with ASTM-D 3023 principles. In the study, roughness (TS 2495 EN ISO 3274 and TS EN ISO 21920-3) and gloss (according to TS EN ISO 2813) values of the samples were calculated after hot and cold check test. Varnished test specimens prepared in 100 mm  $\times$  100 mm  $\times$ 10 mm dimensions were first kept in drying oven at (50±5) °C for 1 hour, then were kept in conditioning room for 1 hour, and then at (-20±2) °C for 1 hour, according to ASTM D1211-97. These processes were accepted as one cycle, and tests were continued until 15 cycles. Afterward, glossiness was measured as perpendicular and parallel to fibers at 60° with a gloss measurement device, and surface roughness values of Ra and Rz were determined with a surface roughness measuring device. According to the results obtained, Scots pine (Pinus sylvestris L.), Eastern beech (Fagus Orientalis Lipsky), and Anatolian chestnut (Castanea sativa Mill.) varnish-coated wood material surfaces all showed a decrease in gloss values after hot-cold shock effect. While an increase occurred in Rz values of roughness for all wooden surfaces, Ra values roughness increased for Scots pine and chestnut and decreased for eastern beech. Gloss and roughness values of surface-treated wood materials against changing weather conditions can determine usefulness of the surface material used. The findings obtained in this study can be useful to manufacturers who use wooden products in the design of marine vehicles and those who export furniture to countries in different climatic conditions.

KEYWORDS: wood material; varnish; surface roughness; glossiness; hot and cold-check test

**SAŽETAK** • U ovom su istraživanju uzorci drva bora (<u>Pinus sylvestris</u> L.), anatolskog kestena (<u>Castanea sativa</u> Mill.) i bukve (<u>Fagus orientalis</u> Lipsky) pripremljeni prema ISO 3129 i kondicionirani prema TS ISO 13061-1, a na njihovu su površinu, prema ASTM D 3023, naneseni nitrocelulozni lak, vodeni lak i lak visokog sjaja. U istraživanju su izračunane vrijednosti hrapavosti (TS 2495 EN ISO 3274 i TS EN ISO 21920-3) i sjaja (TS EN ISO 2813) uzoraka nakon testa vruće i hladne provjere. Lakirani uzorci dimenzija 100 mm × 100 mm ×10 mm najprije su jedan sat sušeni pri (50±5) °C, zatim su jedan sat kondicionirani pri sobnim uvjetima i jedan sat pri (-20±2) °C, prema ASTM D1211-97. Ti procesi sušenja i kondicioniranja čine jedan ciklus, a ispitivanje se sastojalo od

<sup>&</sup>lt;sup>1</sup> Author is researcher at Afyon Kocatepe University, Afyon Vocational School, Department of Design, Afyon, Turkey.

<sup>&</sup>lt;sup>2</sup> Author is researcher at Kutahya Dumlupinar University, Simav Technology Faculty, Department of Wood Works Industrial Engineering, Kutahya, Turkey.

15 ciklusa. Nakon toga izmjeren je sjaj okomito i paralelno s vlakancima drva pod kutom od 60° u odnosu prema mjernom uređaju te su određeni parametri hrapavosti površine Ra i Rz. Prema dobivenim rezultatima, svi su lakirani uzorci drva bora (<u>Pinus sylvestris</u> L.), bukve (<u>Fagus orientalis</u> Lipsky) i anatolskog kestena (<u>Castanea sativa</u> Mill.) nakon testa hladne provjere pokazali smanjenje vrijednosti sjaja. Međutim, na svim površinama uzoraka drva zabilježeno je povećanje Rz vrijednosti hrapavosti, dok su se Ra vrijednosti hrapavosti povećale na borovini i kestenovini, a smanjile na bukovini. Vrijednosti sjaja i hrapavosti površinski obrađenih drvnih materijala mogu odrediti njihovu uporabljivost u promjenjivim okolišnim uvjetima. Rezultati dobiveni u ovom istraživanju bit će korisni proizvođačima koji upotrebljavaju drvo u proizvodnji brodova ili izvoze namještaj u zemlje s drugačijim klimatskim uvjetima.

KLJUČNE RIJEČI: drvni materijal; lak; hrapavost površine; sjaj; test vruće i hladne provjere

#### **1 INTRODUCTION**

#### 1. UVOD

Wood is a renewable material very important for human beings, who have used it in art and architecture from the past to the present. Today, protecting the properties of wood from external factors and providing hygienic conditions has become one of the most critical problems (Atilgan, 2022). In the use of wood material after it is finished, top surface materials such as paint, varnish and impregnation applied to wood surfaces are required to maintain its strength and aesthetic properties (Goktas *et al.*, 2006).

Wood material might be easily deformed by some external factors. Polymer structures such as lignin and cellulose, which make up the wood material, are affected by ultraviolet rays like other organic polymer structures. Negative changes occur in almost all physical and mechanical properties of wood material due to exposure to outdoor conditions without any protective treatment (Pelit and Korkmaz, 2019). Protective treatment is necessary for the long-term and efficient use of wood and wood-based materials for protection. In wooden materials, these processes include impregnation, varnishing and painting (Vardanyan et al., 2015). In the protection process, a protective layer is created, using materials with the feature of layering to protect furniture and decoration elements against physical, mechanical, and chemical effects, outdoor conditions, and biological pests. It is applied in the form of covering wood material surfaces (Sonmez, 2000). In order to extend the aesthetic and economic life of the surfaces, paints and varnishes are the most used materials in creating a protective layer with liquid surface treatments (Kurtoglu, 2000). The most inconvenient factors for wood material in outdoor conditions are temperature, humidity, different wavelengths of sunlight, and UV radiation, and changing of these conditions at certain times of the day according to the season cause negative effects on the wood material (Feist and Hon, 1984). Varnish types and varnishing techniques should be chosen appropriately. The tests applied to determine the resistance of paint and varnish layers against external effects were applied to determine the performance

of varnish systems or to develop products. Today, many surface treatment materials and many application methods have been developed for surface treatments (Ozdemir, 2003). Surface treatment and selection of a protective layer are very important for a long-term use of wood material (Ulay ve Budakci, 2015). There are many studies in the literature about varnish, which is one of the wood finishing materials used in the woodworking and furniture industry. The highest hardness value on varnished surfaces was obtained with polyester varnish. The glossiest surfaces were obtained with polyurethane varnish (Sonmez, 2000; Cakicier, 1994). When Scots pine and chestnut tree species were used in outdoor conditions after impregnation and varnishing, higher hardness values were obtained with polyurethane varnish than with synthetic varnish in both tree species (Peker, 1997). When examining the effect of layer thickness on wood varnishes, it was seen that the third layer varnish applications caused an increase in glossiness (Budakci, 1997). It was observed that the glossiness of waterbased varnishes was lower than that of solvent-based varnishes (Yakin, 2001). Wood species were found to be insignificant for the glossiness of different varnish layers, while the varnish effect was significant (Budakci, 2003). The layer increase in varnish applications negatively affects the flexibility properties of the varnish layer. The layer increases the surface tension and causes exfoliation (Cakicier, 2007). Gurleyen et al. (2017) analyzed glossiness values on parquets with UV varnish applied according to different varnish layers produced from limba, sapele, chestnut, and iroko woods (Gurleyen et al., 2017). Dongel et al. (2008) investigated the glossiness effect of dry heat on solid wood and wood-based flooring materials. Eastern beech solid parquet covered with polyurethane parquet varnish, laminated parquet covered with UV-curable polyurethane varnish, and laminate parquet with highdensity fiberboard (HDF) in the middle layer were used as test samples (Dongel et al., 2008). In addition, Gurleyen et al. (2017) determined the glossiness values in single and double UV system parquet varnish layers applied to rowan (Sorbus L.) wood (Gurleyen et al., 2017).

In the work of Sonmez and Kesik (1999), test samples were prepared by using beech (Fagus orientalis L.), pine (Pinus sylvestris L.) and oak (Quercus petreae L.) wood, and cellulosic, polyurethane, and acrylic varnishes were applied to the surface of sample panels. Samples were first kept at -18 °C for one hour and then removed and kept at 50 °C for one hour. These processes were accepted for one cycle and tests continued for 20 cycles. Tests were carried out according to ASTM D-1211. There was no visible failure on the sample panels. However, all the sample panels lost some hardness. On the other hand, in the case of glossiness measurement, the glossiness of oak panels coated with cellulosic varnish was increased, but the glossiness of other panels was decreased. (Sonmez and Kesik, 1999). It was also reported that, when Gubas solid wood material was subjected to the hot-cold check test applied to its surface varnish (acidic hardener and cellulosic), the surface with cellulosic varnish exhibited fracture, cracking, and color change (Yolanda, 1998).

In his study, Altiparmak (2017) observed the deformation of the surfaces after the hot-cold shock effect of yacht varnish, polyurethane varnish, and epoxy varnish applied on limba, chestnut, and sapele woods. No cracking or surface deformation was observed on the panel surfaces in the 20-cycle cold-check test (Altiparmak, 2017). In the study of Budakci et al. (2010), cellulosic, polyurethane and acrylic varnishes were applied on oriental beech, yellow pine and sessile oak, and color changes were determined after the effects of hot and cold shocks after accelerated aging effects (Budakci et al., 2010). It was stated that, in the hotcold aging tests of the varnish layers, the samples should be kept at temperatures of -20 and 50 °C for 1 h each, and this process should be considered as 1 period; the performance of the layers with no degradation in 10 periods should be considered as sufficiently good, while layers with no degradation in 25 periods should be considered as high-performance (Payne, 1965).

In some areas where wood materials with wood finishing are used, sudden temperature changes occur, which can cause the painted and varnished surface to lose its properties. It is important to determine the effect of these changes on the wood species and varnishes used in places with sudden climate changes. However, when the literature is examined, few studies can be found. Based on the data obtained from the surface roughness and glossiness change on varnished surfaces that may be exposed to the effects of sudden climatic changes (hot and cold), it will be possible to select the appropriate wood type and varnish for the place of use.

Wood species of Scots pine (*Pinus sylvestris* L.), Anatolian chestnut (*Castanea sativa* Mill.), and Eastern beech (*Fagus orientalis* L.) and Cellulosic, waterbased, and glass polish varnishes, which are used in the woodworking industry and manufacture of various marine vehicles, were used in the present study. These tree species and varnishes, which are frequently used, can be deformed in environments where they are exposed to sudden temperature changes (especially marine vehicles, yachts, etc.). The aim of this study was to determine the effect of the Hot and Cold-Check Test on the roughness and gloss changes on the surfaces and to specify the most suitable wood type and varnish.

# 2 MATERIALS AND METHODS

### 2. MATERIJALI I METODE

In this study, Scots pine (*Pinus sylvestris* L.), Anatolian chestnut (*Castanea sativa* Mill.), and Eastern beech (*Fagus orientalis* L.) tree species, which are widely used in the woodworking industry and yacht decoration applications, were preferred.

Cellulosic varnish, water-based varnish and glass polish varnish were used in the experiments. The varnishes used in the experiments were obtained from Çaglayan Wood Products and Hardware Industry Afyonkarahisar/Turkey (Table 1).

Timber used in the preparation of the samples were obtained by random selection from companies in Afyonkarahisar/Turkey. Experiment samples were prepared from sapwood not damaged by insects and fungi, free of knots, cracks, and inclusions, with no discoloration and a smooth fiber structure, according to ISO 3129 (ISO 3129, 2019).

According to ISO 3129, 3 wood species, three varnish types, and five replications for each parameter, i.e. 45 test samples were prepared. Measurements were made from 3 different points on each test sample, and 135 data were obtained for each type of test. The test samples were prepared in the dimensions of 100 mm  $\times$  100 mm  $\times$  10 mm (longitudinal direction  $\times$  radial direction  $\times$  tangential direction) in Afyon Kocatepe Uni-

Table 1	Characteristics	of varnishes us	ed
Tablica	1. Svojstva prin	mijenjenih lakov	va

Type of varnish	Solid ratio, %	Density at 20 °C, g/cm <sup>3</sup>	Viscosity at 20 °C, s
Vrsta laka	Udio suhe tvari, %	<i>Gustoća pri</i> 20 °C, g/cm <sup>3</sup>	Viskoznost pri 20 °C, s
Cellulosic / nitrocelulozni lak	34-40	0.92-0.99	100-110 (DIN6 mm)
Water-based / vodeni lak	32-38	1.02-1.04	30-40 (DIN4 mm)
Glass polish varnish / lak visokog sjaja	47-51	0.92-0.94	90-140 (DIN6 mm)

Peremeter / Payametar	Coded levels / Odabrane razine					
r ar ameter / r ur umetur	Level 1	Level 2	Level 3			
Wood species	Scots pine	Eastern beech	Anatolian chestnut			
vrsta drva	borovina	bukovina	drvo anatolskog kestena			
Type of varnish	Cellulosic	Water-based	Glass polish varnish			
vrsta laka	nitrocelulozni lak	vodeni lak	lak visokog sjaja			
Hot and cold shock test	Before	After				
test vruće i hladne provjere	prije	poslije				

 Table 2 Assignment of levels to factors (parameters used)

 Tablica 2. Dodjeljivanje razina faktorima (korišteni parametri)

versity Afyon Vocational School Furniture and Decoration Workshop. They were conditioned at temperatures of (20±2) °C and (65±5) °C, with a moisture content (MC) of about 12 %. (TS ISO 13061-1, 2021). The surfaces of the samples were sanded with 80 and 120 grit sandpapers, respectively. The varnishing process of the test samples was carried out by the principles specified in ASTM-D 3023 (ASTM-D3023, 2017). The dust on the surface was cleaned with compressed air. The varnishes were applied in two layers at 2 (27.5 bar) atm pressure (DYO, 1990) with an air gun with a tip opening of 1.8 mm, perpendicular and parallel to the fibers, at 125 g/m<sup>2</sup>. After 1 coat of application, the samples were sanded with 400-grit sandpaper. After sanding, the varnish was applied again. Table 2 gives the parameters and levels used in the application.

The experimental process of the study is given in Figure 1.

This test is applied to wood material that has been painted and varnished according to ASTM-D 1211 by exposing it to shock, heat and cold. Its purpose is to determine the varnish layer performance in sudden temperature changes that occur in natural climatic conditions. Hot-cold tests indicate the flexibility of transparent varnish layers on wooden surfaces or their resistance to sudden temperature changes (Sonmez and Kesik,1999). In this context, surface-treated test specimens prepared in dimensions of 100 mm  $\times$  100 mm  $\times$ 10 mm were exposed to cold check test for periods of 1 h (50±5) °C, 1 h laboratory conditions, and 1 h (-20±2) °C, for 15 cycles according to ASTM D 1211-97 (2001) standard.

Surface Roughness Tester Time TR220 (Time Group Inc., China) type surface roughness measurement equipment was used for the determination of surface roughness values via a contact stylus trace method. The sampling length was taken as 0.8 mm. The stylus probe speed was chosen as 10 mm/min, the diameter of the measurement needle was 5  $\mu$ m, and the needle tip was 90°. Care was taken to have a measurement environment around (18–22) °C and without vibrations. Perpendicular and parallel glossiness measurement of the fibers was made. The tool was calibrated before measurement, and calibration was checked at established intervals. The roughness of the wood material after the effect of hotcold shock was determined according to TS 2495 EN ISO 3274 and TS 6212 EN ISO 4288.



Figure 1 Schematic representation of experimental design Slika 1. Shematski prikaz eksperimenta

In the study, glossiness measurements were carried out using a Glossmeter at 60°. Not many errors were observed when measuring at 60° on matte and glossy surfaces (Ozen and Sonmez, 1990; Ordu and Sofuoglu, 2016). Glossiness measurements were carried out using Konica Minolta Multi Gloss 268 Plus (Glossmeter) gloss meter according to the principles specified in TS 4318 EN ISO 2813.

For the evaluation of the data, the Minitab 19 statistical software program was used. The values of the factor effects of wood species, varnish types, and hotcold shock test were determined using the analysis of one-way variance (ANOVA) and the Tukey procedure. The differences in the means were accepted at a significance of P < 0.05.

## **3 RESULTS AND DISCUSSION**

### 3. REZULTATI I RASPRAVA

Experiments were carried out to determine the effect of wood species, varnish types, and hot-cold shock test on roughness parameters (Ra, Rz) and surface glossiness of the samples. Roughness parameters and glossiness values measured on surfaces are given in Table 3.

Since the *P* value is greater than 0.05 (*P*=0.102) according to Figure 2, it is seen that the values of the Ra measurement show a normal distribution at the 95 % confidence level. Analysis of variance results for Ra is given in Table 4.

According to the results of a one-way analysis of variance for Ra (at 95% confidence level, Table 4),



Figure 2 Normality graphs for Ra Slika 2. Grafovi normalne distribucije za Ra

Table 3	Roughness	parameters (	(Ra and R	z) and glos	s values	measured	on surfaces	
Tablica	3. Paramet	ri hrapavosti	(Ra i Rz)	i vrijednos	ti sjaja i	zmjerene r	na površinam	18

<b>Wood species</b> Vrsta drva	Varnish / Lak	Varnish / LakHot and cold shock testTest vruće i hladne provjere		Rz, μm	<b>Glossiness</b> Sjaj
	Cellulosic	Before / prije	0.475	1.595	87.753
	nitrocelulozni lak	After / poslije	0.489	1.854	77.007
Scots pine	Water-based	Before / prije	0.959	4.147	44.827
borovina	vodeni lak	After / poslije	1.169	4.408	43.507
	Glass polish varnish	Before / prije	0.118	0.742	91.520
	lak visokog sjaja	After / poslije	0.175	0.771	85.927
	Cellulosic	Before / prije	0.526	2.047	81.513
	nitrocelulozni lak	After / poslije	0.465	2.337	63.593
Eastern beech	Water-based	Before / prije	1.092	5.347	36.713
bukovina	vodeni lak	After / poslije	0.948	5.600	36.447
	Glass polish varnish	Before / prije	0.217	1.049	89.953
	lak visokog sjaja	After / poslije	0.159	1.090	84.233
	Cellulosic	Before / prije	0.676	2.337	72.053
	nitrocelulozni lak	After / poslije	0.716	2.433	70.40
Anatolian chestnut	Water-based	Before / prije	1.105	5.131	45.047
drvo anatolskog kestena	vodeni lak	After / poslije	1.287	6.309	34.540
	Glass polish varnish	Before / prije	0.106	0.626	92.160
	lak visokog sjaja	After / poslije	0.140	0.962	81.253

Source / Izvor	DF	Adj SS	Adj MS	F Value	P Value
Wood species / vrsta drva	2	0.04470	0.02235	0.12	0.887
Error	15	2.77871	0.18525		
Total	17	2.82341			
Varnish / lak	2	0.6724	1.33622	132.77	0.000
Error	15	0.1510	0.01006		
Total	17	2.8234			
Hot and cold check test test vruće i hladne provjere	1	0.00417	0.004171	0.02	0.880
Error	16	2.81924	0.176203		
Total	17	2.82341			

 Table 4 Results of one-way analysis of variance for Ra

 Tablica 4. Rezultati jednosmjerne analize varijance za Ra

there is a significant difference since P=0.000 for the varnish type. It is seen that tree species (P=0.887>0.05) and hot-cold test (P=0.880>0.05) did not make any significant difference. Tukey test was applied for the varnish type, and it was seen that each varnish type formed a separate group (Table 5).

Table 5 Results of Tukey test for RaTablica 5. Rezultati Tukey testa za Ra

Varnish	N	Mean	Grouping
Lak	1.4	Srednja vrijednost	Grupiranje
2	6	1.0933	А
1	6	0.5578	В
3	6	0.1525	С

Figure 3 presents the main effects plot in terms of *Ra* of wood species, varnish types and hot-cold check test.

According to the main effect plot created for Ra, it was observed that the highest roughness value occurred in Anatolian chestnut tree species, while lower and close values were obtained in other tree species. In terms of varnish type, the Ra roughness value from the lowest is listed as glass polish, cellulosic and water-based. Although the values were close to each other before and after the test in terms of hot and cold-check test, there was a slight increase in the roughness value after the test.

When the interaction graph is examined in terms of  $R_a$  (Figure 4), it is seen that all three tree species have similar results in terms of applied varnish. The lowest roughness value was obtained in glass polish, followed by cellulosic and synthetic varnish, respectively. When the wood type and hot-cold test interaction were examined, a slight increase was observed in the roughness value after the test in Scots pine and Anatolian chestnut tree species, while a decrease occurred in Eastern beech tree species. When the interaction between the varnish type and the hot-cold test was examined, no change was observed in the post-test  $R_a$  value in cellulosic and glass lacquer varnish types, while an increase occurred in the water-based varnish type after the cold-hot test.

Since the *P* value is less than 0.05 (P=0.017) according to Figure 5, it is seen that the values in the *Rz* 



**Figure 3** Main effects plot in terms of *Ra* of wood species, varnish types, and hot-cold check test **Slika 3.** Dijagram glavnih učinaka na *Ra* s obzirom na vrstu drva, vrstu laka i test vruće i hladne provjere



**Figure 4** Interactions of wood species, varnish, and hot-cold check test of *Ra* **Slika 4.** Interakcije vrsta drva, laka i testa vruće i hladne provjere za *Ra* 



**Figure 5** Normality graphs for *Rz* **Slika 5.** Grafovi normalne distribucije za *Rz* 

measurement did not show normal distribution at the 95 % confidence level. The analysis of variance results for Rz is given in Table 6.

According to Table 6, there is a significant difference for Rz as P=0.000 for the varnish type according to the results of a one-way analysis of variance at a 95% confidence level. It is seen that tree species (P=0.792>0.05)

and hot-cold test (P=0.743>0.05) did not make any significant difference. Tukey test was applied for the varnish type in terms of Rz, and it was seen that each varnish type formed a separate group (Table 7).

According to the main effect plot created for Rz, it was observed that the lowest roughness value occurred in Scots pine tree species, while higher and

<b>Tablica 6.</b> Rezultati jednosmjerne analize varijance za <i>Kz</i>						
Source / Izvor	DF	Adj SS	Adj MS	F Value	P Value	
Wood species / vrsta drva	2	1.909	0.9543	0.24	0.792	
Error	15	60.316	4.0211			
Total	17	62.225				
Varnish / lak	2	58.347	29.1736	112.85	0.000	
Error	15	3.878	0.2585			
Total	17	62.225				
Hot and cold check test	1	0.4303	0.4303	0.11	0.743	
test vruće i hladne provjere						
Error	16	61.7946	3.8622			
Total	17	62.2249				

 Table 6 Results of one-way analysis of variance for Rz

 Tablica 6. Rezultati jednosmjerne analize varijance za R.



**Figure 6** Main effects plot in terms of *Rz* of wood species, varnish types, and hot-cold check test **Slika 6.** Dijagram glavnih učinaka na *Rz* s obzirom na vrstu drva, vrstu laka i test vruće i hladne provjere

close values were obtained in other tree species. In terms of varnish type, the Ra roughness value from the lowest is listed as glass polish, cellulosic and waterbased. Although the values were close to each other before and after the test in terms of the hot and coldcheck test, there was a slight increase in the Ra roughness value after the test.

When the interaction graph is examined in terms of Rz, it is seen that all three tree species have similar results in terms of the applied varnish, as in the Ra

Varnish	N	Mean	Grouping
Lak		Srednja vrijednost	Grupiranje
2	6	5.157	Α
1	6	2.107	В
3	6	0.8733	C

**Table 7** Results of Tukey test for *Rz***Tablica 7.** Rezultati Tukey testa za *Rz* 

value. The lowest roughness value was obtained in glass polish, followed by cellulosic and synthetic varnish, respectively. When the interaction between the tree species and the hot-cold test was examined, there was a slight increase in the roughness value after the test in all three tree species. When the interaction between the varnish type and the hot-cold test was examined, it was observed that the Ra value of cellulosic and glass varnish, varnish types, increased slightly after the test, while an increase was observed in the water-based varnish type after the cold-hot test. As a result of sudden temperature changes, roughness parameters (Ra and Rz) increased in Scots pine and Anatolian chestnut tree species for all three varnish types. The literature states that the molecular sizes of water-based varnishes are smaller than those of solvent-based systems; therefore, they penetrate more into the cavities of the wood material, thus giving thin layers (Sonmez et al., 2004).



Figure 7 Interactions of wood species, varnish, and hot-cold check test of RzSlika 7. Interakcije vrste drva, laka i testa vruće i hladne provjere za Rz



Figure 8 Normality graphs for glossiness Slika 8. Grafovi normalne distribucije za sjaj

Table 8 Results of one-way analysis of variance for glossiness Tablica 8. Rezultati jednosmjerne analize varijance za sjaj

Source / Izvor	DF	Adj SS	Adj MS	F Value	P Value
Wood species / vrsta drva	2	153.6	76.78	0.15	0.862
Error	15	7649.2	509.94		
Total	17	7802.7			
Varnish / lak	2	7223.3	3611.66	93.50	0.000
Error	15	579.4	38.63		
Total	17	7802.7			
Hot and cold check test test test vruće i hladne provjere	1	230.4	230.4	0.49	0.495
Error	16	75.72.4	473.3		
Total	17	7802.7			

The low layer thickness reveals the effect of the texture of the wood material and may cause higher roughness (Table 3).

The normality graph for glossiness is given in Figure 8.

Since the *P* value is less than 0.05 (P=0.010) according to Figure 8, it is seen that the values in the glossiness measurement did not show normal distribution at the 95 % confidence level. Variance analysis results for glossiness are given in Table 8.

Based on Table 8 (according to the results of a oneway analysis of variance at a 95 % confidence level), there is a significant difference in terms of varnish type for glossiness as P=0.000. It is seen that tree species (P=0.862>0.05) and hot-cold test (P=0.495>0.05) did not make any significant difference. Tukey test was applied for the varnish type in terms of glossiness, and it was seen that each varnish type formed a separate group (Table 9).

According to the main effect plot created for glossiness (Figure 9), the highest glossiness value occurred in Scots pine tree species, while lower and close values were obtained in other tree species. Glossiness value in terms of varnish type is listed from the lowest to the highest as water-based, cellulosic, and glass polishes. Although the values were close to each other before and after the test in terms of hot and cold-check test, there was a decrease in the glossiness value after the test.

When the interaction graph is examined in terms of glossiness, it is seen that all three tree species have similar results according to the varnish applied. The lowest glossiness value was obtained in water-based varnish in every tree species, followed by cellulosic and glass varnish, respectively. Considering the wood type and the interaction of the hot-cold test, a slight decrease in the post-test glossiness value occurred in each tree species. When the interaction between the varnish type and the hot-cold test was examined, a decrease in the glossiness value after the test occurred in each type of varnish applied. Glossiness values were listed from the highest to the lowest according to the varnish type applied as glass polish, cellulosic and water-based. This sequence did not change before and after the hot-cold test.

 Table 9 Results of Tukey test for glossiness

 Tablica 9. Rezultati Tukey testa za sjaj

Varnish	N	Mean	Grouping
Lak		Srednja vrijednost	Grupiranje
3	6	87.41	А
1	6	75.33	В
2	6	40.18	С



Figure 9 Main effects plot in terms of glossiness of wood species, varnish types, and hot-cold check test Slika 9. Dijagram glavnih učinaka na sjaj s obzirom na vrstu drva, vrstu laka i test vruće i hladne provjere



Figure 10 Interactions of wood species, varnish, and hot-cold check test of glossiness Slika 10. Interakcije vrsta drva, laka i testa vruće i hladne provjere za sjaj

Kesik (1999) determined the effect of the hot-cold shock test on hardness and glossiness in Scots pine (Pinus sylvestris L.), Eastern beech (Fagus orientalis L.) and oak (Quercus petreae L.) tree species with cellulosic, polyurethane and acrylic varnish. According to the results of the experiment, it was determined that after the process, the hardness and glossiness values of all varnish types decreased. Ozalp et al. (2009), stated that hardness and glossiness values decreased in solid wood material kept at 100, 150 and 200 °C for 4 and 6 hours. The comparison, made in terms of the aging factor, showed that the highest total color change value was obtained in the samples without the hot-cold test, and the lowest in the samples with the hot-cold test. Similarly, the highest glossiness value was found in the samples without the hot-cold test and the lowest in the samples with the hot-cold test. It was reported in the literature that the varnish layers prepared for wooden surfaces lose their glossiness in outdoor weather conditions (Akpinar, 2012). Glossiness values also differ according to varnish types (Saygin, 2016). It was observed and found in the literature that a decrease in glossiness values occurs when wood is exposed to hot or cold external effects. As a result of sudden temperature changes, the glossiness decreased for each tree species and each varnish type. It was stated that the reasons for the decrease in the value of wood could lie in the formation of hemicelluloses and extractives, decomposition of products resulting from the effect of high heat or lignin polymerization reactions formed as a result of heat (Kamperidou et al., 2013; Barcík et al., 2015). In their study, they reported that the natural glossiness value of some trees was affected negatively after the increase in temperature and that the decrease in the glossiness value of the wood material depending on the temperature occurred as a result of chemical changes in glucose, hemicellulose and lignin

(Esteves *et al.*, 2008). The glossiness values differed according to the varnish type, and this result was also supported by the literature (Sonmez, 1989). The highest glossiness value was determined in the samples without the hot-cold test and the lowest in the samples with the hot-cold test (source). The literature states that the molecular sizes of water-based varnishes are smaller than those of solvent-based systems. Therefore, they penetrate more into the cavities of the wood material, thus creating thin layers (Sonmez *et al.*, 2004). The low layer thickness reveals the effect of the texture of the wood material and the lowest glossiness value is obtained in water-based varnish. Water-based varnishes give the closest glossiness values to unvarnished samples (So-fuoglu and Aykac, 2020).

# 4 CONCLUSIONS

#### 4. ZAKLJUČAK

Cellulosic varnish, water-based varnish and glass polish parquet varnish were applied to the surfaces of Scots pine, Eastern beech and Anatolian chestnut. Glossiness and roughness measurements were performed to detect the deformation on the surface during the hot-cold test process applied to the surfaces. The results show as follows:

The highest glossiness value was observed in Scots pine trees, while lower and close values were obtained in other tree species.

In terms of varnish type, the glossiness value was listed from the lowest as water-based (40.18 gloss), cellulosic (75.33 gloss) and glass polish (87.41 gloss).

Although the values were close to each other before and after the test in terms of hot and cold-check test, there was a decrease in the glossiness value after the test.

It was observed that the highest average roughness value (Ra) occurred in Anatolian chestnut trees, while lower and close values were obtained in other tree species.

In terms of varnish type, *Ra* roughness value was listed from the lowest as glass polish (0.1525  $\mu$ m), cellulosic (0.5578  $\mu$ m) and water-based (1.0933  $\mu$ m).

Although the pre-test and post-test values were close to each other in terms of hot and cold-check test, there was a slight increase in the roughness value after the test.

It can be said that the changes in glossiness and roughness that occur on the surface of the varnished wood material after the hot-cold shock test do not affect the useful functions and surface properties of the products but are only affected in terms of aesthetics.

It can be predicted that such surface changes may occur in the three types of varnish applied to all three tree species, the furniture exported to countries with different climatic conditions, and the surface treatment material used on sea vehicles such as yachts/boats.

#### **5 REFERENCES**

#### 5. LITERATURA

- Akpinar, E., 2012: The effect of stem height on the physical and mechanical properties of European larch (*Larix decidua* Mill.) wood grown in Trabzon-Macka region. MSc Thesis, Karadeniz Technical University, Graduate School of Natural and Applied Sciences Trabzon, Turkey.
- 2. Altiparmak, M., 2017: Comparison of the performance of various varnishes used in wooden yachts on different wood species. MSc Thesis, Mugla Sitki Kocman University, Graduate School of Natural and Applied Sciences, Mugla, Turkey.
- Aykac, S.; Sofuoglu, S. D., 2020: Investigation of the effect of varnish types on surface properties used in bamboo wooden material. Journal of Polytechnic, 24 (4): 1353-1363.
- Atilgan, A., 2022: The use of resin laque as a top surface material on wood material and investigation of its antimicrobial effect. PhD Thesis, Gazi University Graduate School of Natural and Applied Sciences, Ankara, Turkey.
- Barcík, Š.; Gašparík, M.; Razumov, E. Y., 2015: Effect of temperature on the color changes of wood during thermal modification. Cellulose Chemistry and Technology, 49 (9-10): 789-798.
- Budakci, M., 1997: Effects of the resistance to adhesion surface of layer thickness glossy and hardness at the wood's varnishes. MSc Thesis, Gazi University Graduate School of Natural and Applied Sciences, Ankara, Turkey.
- Budakci, M., 2003: Design and production of a new adhesion testing device and its utilization with testing of wood varnishes. PhD Thesis, Gazi University Graduate School of Natural and Applied Sciences, Ankara, Turkey.
- Budakci, M.; Sevim Korkut, D.; Esen, R., 2010: The color changes on varnish layers after accelerated aging through the hot and cold-check test. African Journal of Biotechnology, 9 (24): 3595-3602.
- Çakicier, N., 1994: The effects of varnishes treated on wood surfaces in the colour of wood stains. MSc Thesis, Gazi University Graduate School of Natural and Applied Sciences, Ankara, Turkey.
- Cakicier, N., 2007: Changes due to weathering of surface finishing layers of wood. PhD Thesis, Istanbul University Graduate School of Natural and Applied Sciences, Istanbul, Turkey.
- Dongel, N.; Kureli, I.; Sogutlu, C., 2008: The effect of dry heat for the colour and gloss changes on the wood and wood based floor covering materials. Journal of Polytechnic, 11 (3): 255-263.
- Esteves, B.; Velez Marques, A.; Domingos, I.; Pereira, H., 2008: Heat-induced colour changes of pine (*Pinus pinaster*) and eucalypt (*Eucalyptus globulus*) wood. Wood Science and Technology, 42: 369-384.
- Feist, W. C.; Hon, D. N. S., 1984: Chemistry of weathering and protection, The Chemistry of Solid Wood. American Chemical Society, Chapter 11, pp. 401-451.
- 14. Goktaş, O.; Mammadov, R.; Duru, M. E.; Baysal, E.; Çolak, A. M.; Ozen, E., 2006: Development of new environment friendly natural-colored preservatives for wood surface dying derived from different tree and herbaceous plant extracts and determination of their color parameters. Ekoloji, 15 (60): 16-23 (in Turkish).
- Gurleyen, T.; Ayata, U.; Gurleyen, L.; Esteves, B.; Sivrikaya, H.; Can A., 2017: The determination of colour and

glossiness values on the parquets that underwent single and double layered UV varnishing system treatment. In: Proceedings of 2<sup>nd</sup> International Conference on Material Science and Technology in Cappadocia (IMSTEC'17), Nevsehir, Turkey, pp. 408-412.

- Gurleyen, T.; Ayata, U.; Gurleyen, L.; Esteves, B.; Çakicier, N., 2017: Determination of color, glossiness and hardness values on single and double layer uv system parquet varnish layers applied to rowanberry (*Sorbus* L.) Wood. In: Proceedings of 5<sup>th</sup> International Symposium on Innovative Technologies in Engineering and Science (ISITES2017), Baku – Azerbaijan, pp. 1327-1336.
- 17. Kesik, H. I., 1999: The Tests with accelerating aging effects on varnish layers. MSc Thesis, Gazi University, Graduate School of Natural and Applied Sciences, Ankara, Turkey.
- Kurtoglu, A., 2000: Agaç malzeme yuzey islemleri, Genel bilgiler, Cilt I, İstanbul University Publications, Istanbul University Publication no: 4262 (in Turkish).
- Kamperidou, V.; Barboutis, I.; Vasileiou, V., 2013: Response of colour and hygroscopic properties of Scots pine wood to thermal treatment. Journal of Forestry Research, 24: 571-575. https://doi.org/10.1007/s11676-013-0389-y
- Ordu, M.; Sofuoglu, S. D., 2016: Comparison of natural color and brightness value for the various types of wood. Ejovoc, 6 (4): 43-51.
- Ozalp, M., 2008: The investigation of borax pentahydrate influences with double components in varnish applications of wood materials. Wood Research, 53 (4): 121-128.
- 22. Ozdemir, T., 2003: The Investigations of varnishes' features at some tree species grown in Turkey. PhD Thesis, Karadeniz Technical University Graduate School of Natural and Applied Sciences, Turkey.
- Ozen, R.; Sonmez, A., 1990: Ahşap yüzeyler için hazırlanan verniklerin fiziksel, kimyasal ve mekanik etkilere karşı dayanaklıkları, Doğa Türk Tarım ve Ormancılık Dergisi, 1428-1436 (in Turkish).
- 24. Payne, H. F., 1965: Organic Coating Technology, Library of Congress Catalog, 1: 390.
- 25. Peker, H., 1997: Efficiency of treatment chemicals materials to varnishes used on surfaces of furniture. PhD Thesis, Karadeniz Technical University Graduate School of Natural and Applied Sciences, Trabzon, Turkey.
- Pelit, H.; Korkmaz, M., 2019: Effect of water-based varnishes added nanographene on the surface properties of beech (*Fagus orientalis* Lipsky) wood, Journal of Politechnic, 22 (1): 203-212. https://doi.org/10.2339/politeknik.385436
- Saygin, E., 2016: The effects of resin cleaning process on the layer performance of water-based varnishes. MSc Thesis, Duzce University, Institute of Science and Technology, Duzce, Turkey.
- Sönmez, A., 1989: Durability of varnishes used on surfaces of wooden furniture against important physical mechanical and chemical effects. PhD Thesis, Gazi University, Institute of Science and Technology, Ankara, Turkey.
- Sonmez, A.; Kesik, H. I., 1999: The aging tests for varnish coating with cold-check. Journal of Polytechnic, 2 (4): 27-32.

#### **Corresponding address:**

- Sonmez, A., 2000: Ağaçişlerinde üstyüzey işlemleri: Hazırlık ve renklendirme, Gazi Üniversitesi Publication, Technical Education Faculty, Ankara (in Turkish).
- Sonmez, A.; Budakci, M.; Yakın, M., 2004: Effect of application methods on the hardness gloss and adhesion strength of waterborne varnish coating on the wooden surface. Journal of Polytechnic, 7 (3): 229-235.
- Ulay, G.; Budakci, M., 2015: Studies carried out in Turkey regarding water-based varnishes used on wood surfaces. Duzce University Journal of Science and Technology, 3 (2): 470-480.
- 33. Yakin, M., 2001: Effects of the resistance to adhesion, glossy and hardness at the waterborne varnishes. Master's thesis, Gazi University Graduate School of Natural and Applied Sciences, Ankara, Turkey.
- Vardanyan, V.; Galstian, T.; Riedl, B., 2015: Effect of addition of cellulose nanocrystals to wood coatings on color changes and surface roughness due to accelerated weathering, Journal of Coatings Technology and Research, 12 (2): 247-258. https://doi.org/10.1007/s11998-014-9634-3
- 35. Yolanda, L. T., 1998: Evaluation of finishing properties and coating systems for gubas, Forest Products Research and Development Institute, 24 (1): 33-41.
- 36. \*\*\*ASTM-D 1211:1997 Standard test method for temperature-change resistance of clear nitrocellulose lacquer films applied to wood. ASTM International, West Conshihocken, PA, USA.
- \*\*\*ASTM D2244-21:1998 Standard practice for calculation of color tolerances and color differences from instrumentally measured color coordinates. ASTM Standards, ASTM International, W. Conshohocken, PA.
- \*\*\*ASTM D 3023:1998 Standard practice for determination of resistance of factory applied coatings on wood products to stains and reagents, pp. 1-3. ASTM Standards, ASTM International, W. Conshohocken, PA.
- \*\*\*ASTM D3023-98R17:1998 Standard practice for determination of resistance of factory-applied coatings on wood products to stains and reagents, ASTM Standards, ASTM International, W. Conshohocken, PA.
- 40. \*\*\*DYO, 1996: Spray gun motion study. Technical Bulletin, Dewilux Factories Co., Izmir.
- \*\*\*ISO 3129:2019 Wood Sampling methods and general requirements for physical and mechanical testing of small clear wood specimen. Switzerland: International Organization for Standardization.
- 42. \*\*\*TS 2495 EN ISO 3274:2005 Geometrical Product Specifications (GPS) – Surface texture: Profile method – Nominal characteristics of contact (stylus) instruments. Turkish Standard Institution, Ankara, Turkey.
- \*\*\*TS EN ISO 2813:2014 Paints and varnishes Determination of gloss value at 20 degrees, 60 degrees and 85 degrees, Turkish Standard Institution, Ankara, Turkey.
- \*\*\*TS EN ISO 21920-3:2022 Geometrical Product Specifications (GPS) Surface texture: Profile Part 3: Specification operators, Turkish Standard Institution, Ankara, Turkey.
- 45. \*\*\*TS EN ISO 13061-1:2021 Physical and mechanical properties of wood – Test methods for small clear wood specimens. Part 1: Determination of moisture content for physical and mechanical tests, Turkish Standard Institution, Ankara, Turkey.

# Assoc. Prof. Dr. SAIT DUNDAR SOFUOGLU

Kutahya Dumlupinar University, Faculty of Simav Technology, Department of Wood Works Industrial Engineering, 43500 Simav/Kutahya, TURKEY, e-mail: sdundar.sofuoglu@dpu.edu.tr

Kenan Kilic<sup>1,2</sup>, Kursat Kilic<sup>3</sup>, Brian Bino Sinaice<sup>3</sup>, Ugur Ozcan<sup>4</sup>

# Wood Species Image Classification Using Two-Dimensional Convolutional Neural Network

# Klasifikacija vrsta drva prema slikama uz pomoć dvodimenzionalne konvolucijske neuronske mreže

#### **ORIGINAL SCIENTIFIC PAPER**

Izvorni znanstveni rad Received – prispjelo: 18. 2. 2023. Accepted – prihvaćeno: 20. 9. 2023. UDK: 630\*85 https://doi.org/10.5552/drvind.2023.0093 © 2023 by the author(s). Licensee University of Zagreb Faculty of Forestry and Wood Technology. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license.

**ABSTRACT** • The woodworking industry's recognition and classification of timber is essential for trade, production and timber science. Traditional methods of identifying wood types are complex, time-consuming, costly and require expertise in wood science. Traditional techniques have been replaced by convolutional neural networks (CNNs), a deep learning tool to better identify wood species. In contrast to earlier studies that used pretrained models, a novel architecture designed explicitly for the WOOD-AUTH dataset was proposed in this study to develop a new 2D CNN model. The data collection encompasses high-level visual representations of 12 distinct types of timber. It is aimed to create a simpler and faster model as an alternative to time-consuming and heavy wood classification models. Compared to previous studies, this research worked with a newly structured 2D CNN network based on 12 wood species. High accuracy and fast computation time were achieved using fewer numbers (three layers) of the convolutional neural network. The proposed model achieved 94 % accuracy, 87 % precision, 81 % recall, 80 % F1 score and 112 minutes 27 seconds computation time. The 2D CNN model performed better than the transfer learning models regarding training epochs. The primary benefit of the model is its ability to achieve high accuracy with lower computation time, even at high epochs compared to other models. The introduced 2D CNN model produced satisfactory outcomes for wood species classification.

**KEYWORDS:** 2D convolutional neural network; image classification; deep neural network; wood species

**SAŽETAK** • Identifikacija i klasifikacija drva u drvnoj industriji ključna je za trgovinu, proizvodnju i znanost o drvu. Tradicionalne metode identifikacije vrste drva složene su, dugotrajne i skupe te zahtijevaju stručnost s područja znanosti o drvu. Za bolju identifikaciju vrste drva tradicionalne su metode zamijenjene konvolucijskim neuronskim mrežama (CNN), odnosno alatom za duboko učenje. Za razliku od ranijih studija koje su se koristile unaprijed obučenim modelima, u ovoj je studiji predložena nova arhitektura dizajnirana upravo za skup podataka WOOD-AUTH kako bi se razvio novi 2D CNN model. Zbirka podataka obuhvaća vizualne prikaze visoke razluči-

<sup>&</sup>lt;sup>1</sup> Author is researcher at Gazi University, Graduate School of Natural and Applied Sciences, Department of Wood Products Industrial Engineering, Ankara, Turkey.

<sup>&</sup>lt;sup>2</sup> Author is researcher at Yozgat Bozok University, Yozgat Vocational School, Design Department, Yozgat, Turkey.

<sup>&</sup>lt;sup>3</sup> Authors are researchers at Akita University, Graduate School of International Resource Sciences, Geotechnology and Materials Engineering for Resources, Department of Geosciences, Akita, Japan.

<sup>&</sup>lt;sup>4</sup> Author is researcher at Gazi University, Faculty of Technology, Woodworking Industrial Engineering Department, 06500 Technical Schools, Ankara, Turkey.

vosti 12 različitih vrsta drva. Cilj je bio stvoriti jednostavniji i brži model kao alternativu dugotrajnim i složenim modelima klasifikacije drva. Za razliku od prethodnih istraživanja, u ovom je istraživanju primijenjena nova 2D CNN mreža koja se temelji na 12 vrsta drva. Visoka točnost i brzo vrijeme izračuna postignuti su korištenjem manjeg broja slojeva (tri sloja) konvolucijske neuronske mreže. Predloženim je modelom postignuta točnost od 94 %, preciznost od 87 %, opoziv od 81 %, F1 rezultat od 80 % i vrijeme izračuna od 112 minuta i 27 sekundi. Model 2D CNN pokazao se boljim od modela transfernog učenja u smislu epohe poduke. Primarna prednost modela jest njegova sposobnost postizanja visoke točnosti uz kraće vrijeme izračuna, čak i pri visokim epohama u usporedbi s drugim modelima. Prezentirani 2D CNN model dao je zadovoljavajuće rezultate za klasifikaciju vrste drva.

KLJUČNE RIJEČI: 2D konvolucijska neuronska mreža; klasifikacija slike; duboka neuronska mreža; vrste drva

#### **1 INTRODUCTION**

#### 1. UVOD

Wood has played an important role in human existence over time, fundamentally essential in many respects. With technological advancement, the industrial utilization and application of wood have been further enhanced. In its natural state, wood is an attractive traditional material with numerous advantages that have made it popular compared to other materials (Jones and Brischke, 2017; Nguyen et al., 2017; Popescu et al., 2011). It is important to possess a comprehensive understanding of its remarkable physical attributes, anatomical structure, mechanical properties, and chemical composition to ensure efficient use of wood. Furthermore, it is equally important to recognize its significant role as an essential prerequisite. (Brauns and Rocens, 2007; Kasal, 2004; Winandy and Rowell, 1984; Zabel and Morrell, 2020). Moreover, it should be noted that some wood species are protected nationally and globally. Therefore, identifying the wood species was important (Kırbaş and Çifci, 2022). The exact identification of wood species forms a pressing issue in different domains, including ecology, construction, furniture manufacturing and restoration, and determining the mechanical and economic properties of wood and wood-based materials. Wood species such as teak, ebony, and mahogany possess unique anatomical, physical, aesthetic, chemical, and mechanical properties (Tou et al., 2007; Vacha and Haindl, 2013). Differences in composition and structure vary between wood species and are used in their differentiation (Huang et al., 2020).

Indeed, the traditional method of wood species identification based on macroscopic and microscopic characteristics has limitations, such as being time-consuming, impractical, and expensive. Moreover, the accuracy of this method may depend on the expertise of the professionals involved, which may vary. Hence, using machine learning and computer vision methodologies to automate processes holds immense potential for delivering expedited, efficient, and cost-effective solutions of wood species identification (Mohan *et al.*, 2014; Rajagopal, 2019). The wood industry faces a significant hurdle in swiftly identifying a substantial tim-

ber volume. To address this issue, leveraging machine learning techniques and approaches can substantially enhance the speed and precision of wood species identification methods (Kırbaş and Çifci, 2022). Researchers have been striving to improve efficiency in recent years by shifting towards computer algorithms for wood image recognition methods instead of relying solely on human labour. One commonly utilized technique for recognizing wood species and their classification is to employ surface recognition models based on texture features. Most computer-based identification methods consist of two significant stages: feature extraction and classification. Boundary detection algorithms play an essential role in feature extraction methods. These methods include the Gray Level Co-occurrence Matrix (GLCM) (Manik et al., 2020; Santosa, 2019) for extracting grey colour matrices, the colour history statistical method (Zhao, 2013), and classification methods such as Support Vector Machine (SVM) (Sun, 2015; Souza et al., 2020), k-nearest neighbour algorithm (KNN) (Gani and Limam, 2013; Kobayashi et al., 2015; Fuentealba et al., 2014), and neural networks (Ravindran et al., 2018; Yinglai et al., 2020; Yang et al., 2019). Huayu et al. (2017) employed the structural covariance networks and mean squared error (SCN-MSE) technique to enhance the characteristics of digital images depicting Mongolian timber. Zhao (2013) devised a reliable approach for classifying wood species based on coloured wood surface images that effectively differentiate between intra-species and inter-species colour variations. Wang et al. (2013) successfully detected eight wood species in their experiments, achieving an accuracy rate of 86 %. Using the Fisher-Tree method, they extracted features from Mask Matching Images (MMI) and employed support vector machines (SVM) for species classification. Hafemann et al. (2014) conducted a comparative analysis, evaluating the performance of traditional classifiers against convolutional neural networks (CNNs). The first dataset comprised macroscopic images of 41 species, while the second dataset contained microscopic images of 112 species. The study found that the accuracy of CNN for the dataset of microscopic images was 97.32 %, while for the dataset of macroscopic images, it was 95.77 %. Kwon et al. (2017) introduced a CNN-based automatic identification system for wood species, while Huang et al. (2020) evaluated the performance of various CNN models for wood species recognition. The study reported that the LeNet3 model achieved the highest accuracy of 99.3 %. In addition, combining deep learning and machine learning techniques resulted in a recognition accuracy of 93.07 %. Tang et al. (2018) utilized the SqueezeNet architecture to develop a wood identification system for rapidly and reliably identifying wood species. Their research was focused on 100 frequently traded wood species in Malaysia, and their CNN-based model achieved top 1 and top 2 accuracies of 77.52 % and 87.29 %, respectively. Geus et al. (2020) conducted a study to evaluate the performance of four newly proposed CNN architectures for the classification of microscope images of wood. They compared the results with a pre-existing attribute method. Among the newly proposed architectures, the DenseNet model was found to achieve the highest accuracy of 98.75 %, indicating its effectiveness in wood species recognition. Lopes et al. (2020) assessed the performance of InceptionV4 and ResNetV2 architectures on North American hardwoods and obtained an accuracy of 92.60 %. Sun et al. (2021) used transfer learning with ResNet50 for wood species identification and achieved a high accuracy of 99.60 %. Fabijańska et al. (2021) employed a residual CNN architecture to recognize 14 European coniferous and flowering wood types in Europe. Their method achieved 93 % and 98.7 % accuracy rates for wood image patches and core images, respectively.

The literature review summarizes the drawbacks of previous research as listed below in bullet points:

- Conventional wood recognition is laborious, expensive, unfeasible, intricate, and arduous.
- The conventional techniques necessitate significant data processing, labour, and proficient expertise in selecting and extracting features.
- While deep learning methods have shown promise in many areas of image classification, the disadvantage is that the architectures employed are resource-intensive and time-consuming.
- It can be seen that the transfer learning models developed by (Haefmann *et al.*, 2014), (Kwon *et al.*, 2017), (Lopes *et al.*, 2020) and (Su *et al.*, 2021) provide accuracy higher than the proposed 2D CNN model. However, these applications require higher hardware systems for the model training, such as Tesla C2050 GPU with high numbers of epochs and XEON CPU with 64 GB RAM capacity computer specifications. In contrast to the high transfer learning models, the proposed 2D CNN model provided 94 % accuracy with NVIDIA GeForce RTX 2070 with 16 GB RAM capacity with shorter computation

time. The previous transfer learning applications are not only expensive for the hardware system but also expensive for the computation time.

As an alternative version, this paper aims to address the limitations of existing studies and develop an efficient 2-dimensional convolutional neural network (2D CNN) model that can quickly and accurately identify wood species. The proposed model is designed specifically for the WOOD-AUTH (Barmpoutis, 2017; Barmpoutis *et al.*, 2018) dataset and aims to improve classification accuracy compared to previous work that used popular pre-trained models. Therefore, it is desired to develop a more straightforward and faster model as an alternative to time-consuming and heavy wood classification models.

This paper is organized as follows: Section 2 details the dataset employed in the study. Section 3 outlines the methodology adopted in the research. Section 4 introduces the proposed 2D CNN model and discusses its implementation. Section 5 presents the findings and corresponding discussions. Lastly, Section 6 concludes the paper with a summary of the conclusions.

#### 2 DATA SET

#### 2. SKUP PODATAKA

The proposed model was developed using the WOOD-AUTH Laboratory of Wood Technology of Forestry and Natural Environment School of the Aristotle University of Thessaloniki, Greece (Barmpoutis, 2017; Barmpoutis *et al.*, 2018) dataset. The dataset consists of 12 wood species; the total data contain 8160 images. Table 1 illustrates each wood species class with a number of image datasets. The different number of datasets for Walnut wood is less than for other wood types because of pre-defined data. The dataset can be called imbalanced due to the wood sample size.

The images were cropped to  $200 \times 200$  pixels with 96 dots per inch (dpi). The  $200 \times 200$  is low for some deep-learning applications but still efficient for wood classification. The WOOD-AUTH was prepared with 400 × 400 pixels (Barmpoutis, 2017; Barmpoutis *et al.*, 2018) in the original data. However, the image resolution is decided based on the specific task, hardware, software and artificial intelligence model computation time. The 200 × 200 images were arranged for the proposed deep learning model to avoid the high computational time, memory requirement and processing power.

Figure 1 demonstrates the image samples from each class in the WOOD-AUTH. These sections indicate distinct features of the wood to provide detailed information for identifying the wood species. Considering various sections, the WOOD-AUTH dataset has different characteristics and variations.

Table 1 WOOD-AUTH dataset details the Latin name, type of species, category and number of images in each class	
Tablica 1. Skup podataka WOOD-AUTH detaljno opisuje latinski naziv, vrstu drva, kategoriju i broj podataka u svakoj kla	si

Class index Indeks klase	Latin name Latinsko ime	Species / Vrste drva	<b>Category</b> Kategorija	Number of images Broj slika
Wood-1.	Fagus sylvatica	European beech drvo bukve	Diffuse-porous hardwood difuzno porozna listača	1223
Wood-2.	Juglans regia	Juglans regia Walnut drvo oraha		289
Wood-3.	Castanea sativa	Castanea sativa Sweet chestnut drvo pitomog kestena		1532
Wood-4.	Quercus cerris	Turkey oak drvo turskog oraha	Ring-porous hardwood prstenasto porozna listača	600
Wood-5.	Alnusglutinosa	Alder drvo johe	Diffuse-porous hardwood difuzno porozna listača	696
Wood-6.	Fraxinusornus	Fraxinusornus Manna ash drvo crnog jasena		648
Wood-7.	Picea abies	Picea abies Norway spruce drvo obične smreke		460
Wood-8.	Pinus sylvestris	Scots pine drvo bijelog bora	Softwood četinjača	332
Wood-9.	Ailanthus altissima	Tree-of-heaven Ring-porous hardw nebesko drvo prstenasto porozna l		332
Wood-10.	. Robiniapseudoacacia Black locust drvo bagrema		Ring-porous hardwood prstenasto porozna listača	440
Wood-11.	Cupressus sempervirens	Mediterranean cypress drvo sredozemnog čempresa	Softwood četinjača	552
Wood-12.	Platanus orientalis	Old world sycamore drvo azijske platane	Diffuse-porous hardwood difuzno porozna listača	1056
		Total / Ukupno		8160



**Figure 1** Images display samples of twelve wood species. (a) European Beech, (b) Common Walnut, (c) Sweet Chestnut, (d) Turkey Oak, (e) Common Alder, (f) Manna Ash, (g) Norway Spruce, (h) Scots Pine, (i) Tree of Heaven, (j) Black Locust, (k) Mediterranean Cypress, (l) Oriental Plane

Slika 1. Slike prikazuju uzorke dvanaest vrsta drva: (a) drvo bukve, (b) drvo običnog oraha, (c) drvo pitomog kestena, (d) drvo turskog hrasta, (e) drvo obične johe, (f) drvo crnog jasena, (g) drvo obične smreke, (h) drvo bijelog bora, (i) nebesko drvo, (j) drvo bagrema, (k) drvo sredozemnog čempresa, (l) drvo azijske platane

The 8160-image data was split into the train (80 %) and test (20 %). 6528 images were used for the model training, and 1632 were used for the model test evaluation. The two-dimensional (2D) convolution neural network (CNN), model evaluation metrics and the proposed model are explained in detail in Section 3 and Section 4, respectively.

#### 3 METHODOLOGY

#### 3. METODOLOGIJA

Employing the MATLAB software neural network toolbox, the proposed 2D CNN was configurated from scratch based on the different number of epochs. First, minor epochs were implemented to check model progress with low accuracy. The number of epochs has been increased to 30 and 50; it should be noted that those numbers were decided by trial and error. The best results were obtained with 50 epochs. The 2D CNN classifier consists of the ReLU activation and softmax output function with Adam optimizer and cross-entropy loss. The hyperparameters are elaborated on in Section 4. The model was interpreted by performing accuracy, precision, recall and F1 score indicators and compared to some transfer learning models in Tensor-Flow Hub.

# **3.1 Explanation of CNN** 3.1. Objašnjenje CNN-a

O'Shea and R. Nash (2015) explained the introduction of the CNN algorithm. (CNNs) resemble traditional Artificial Neural Networks (ANNs) because they consist of neurons that undergo self-optimization through learning. Each neuron receives input and performs operations, such as a scalar product followed by a non-linear function, similar to many other ANNs. The entire network, from the input raw image vectors to the final output of the class score, can still be expressed as a single perceptive score function (the weight). The network last layer contains loss functions associated with the classes, and the usual tips and tricks developed for traditional ANNs are still applicable. CNNs differ from traditional ANNs in their primary use for image pattern recognition. This distinction allows the encoding of image-specific features into the architecture, which is essential when establishing CNN models. The convolutional layer is the first layer of CNN and works for feature extraction from the input image. The pooling layer reduces the number of parameters but does not have any function. The fully connected layer (last layer) is flattening the previous output layer in a vector. Kilic et al. (2022) explained the 1D CNN application for a regression analysis to predict machine-based datasets. Dhillon and Verma (2020) indicated that CNN consists of neurons with a learnable weight and bias. Sinaice *et al.* (2022) presented deep neural network image mapping and automatic recognition. Jannat *et al.* (2021) showed that CNN architecture has convolutional, pooling, and fully connected layers. In summary, CNN-based architectures are widely used in the literature for image classification; thus, 2D CNN was preferred to classify the WOOD-AUTH dataset. Section 3 demonstrates the model evaluation metrics, and Section 4 elaborates on the proposed model with hyperparameters.

#### 3.2 Evaluation metrics

3.2. Mjerila evaluacije

#### 3.2.1 Accuracy

#### 3.2.1. Točnost

According to Powers (2020), the accuracy measure refers to the ratio of correctly predicted data points to all data points in the dataset. Eq. 1 expresses the formula of accuracy metrics.

$$accuracy = \frac{TP - TN}{n^+ + n^-} = \frac{TP + TN}{TP + TN + FP + FN}$$
(1)

*TP* as a correctly identified positive instance, *TN* as a correctly identified negative instance, *FP* as a mistakenly identified positive instance, and *FN* as a mistakenly identified negative instance.

#### **3.2.2 Precision** 3.2.2. Preciznost

The precision score is a measure of the accuracy between the total number of true positive samples and the total number of predicted samples defined as positive (Power, 2020). Eq. 2 illustrates the calculation for the precision score.

$$precision = \frac{TP}{TP + FN}$$
(2)

#### 3.2.3 Recall

#### 3.2.3. Opoziv

The recall score was defined as a metric that compares true positive samples to the total number of true positive samples (Power, 2020).

$$precision = \frac{TP}{TP + FN}$$
(3)

# **3.2.4 F1 Score** 3.2.4. F1 rezultat

The F1 score is a harmonic mean of the precision and recall shown in Eq. 4

$$F1score = \frac{2 \cdot TP}{2 \cdot TP + FP + FN} = 2 \cdot \frac{precision \cdot recall}{precision + recall}$$
(4)

The worst value is 0 for all evaluation metrics, whereas the best is 1.

# 4 PROPOSED 2D CNN CLASSIFICATION MODEL FOR WOOD SPECIES 4. PREDLOŽENI 2D CNN MODEL ZA KLASIFIKACIJU VRSTA DRVA

The 2D CNN models are preferred for image recognition and classification for different tasks, such as the wood industry, medical science, and interdisciplinary engineering fields. Therefore, 2D CNN was selected to classify the wood species based on their texture. The image dataset was normalized using zero-centre normalization. Afterwards, the proposed model was designed based on the explanation of (Simonyan and Zisserman, 2015) using 3 and 5 kernels with ReLU activation function and 400 fully connected layers to reduce training time. The filter size was selected. The filter size range between 32 and 1024 as the power of two, and a large filter size can provide a powerful model. However, the large filter size was not selected to create the proposed model in case of overfitting. Therefore, the proposed model worked with 20, 40 and 50 filter sizes. Due to the multiple classes, the output function was decided to work with the softmax function with cross-entropy (Zhang and Sabuncu, 2018). The Adam optimizer was implemented in the convolutional network for gradient descent optimization. Compared to the previous studies, the proposed model included three convolutional layers. Figure 2 illustrates the proposed model structure with the values of each parameter.

#### 4.1 ReLU activation function

#### 4.1. ReLU aktivacijska funkcija

According to Albawi *et al.* (2017) and Ajit *et al.* (2020), the ReLU function is implemented as an activation function. It converts negative values to zero and does not influence volume and hyperparameters. Compared to the sigmoid and tanh activation functions, the

ReLU function is unaffected by vanishing gradient descent related to the deeper neural networks. Therefore, the ReLU has been preferred over the sigmoid and tanh activation functions. Eq. 5 shows the ReLU function equation.

$$ReLU(x) = \max(0, x)$$

$$\frac{d}{dx}(x) = \{1 \text{ if } x > 0; 0 \text{ otherwise}\}$$
(5)

# 4.2 Softmax output layer

4.2. Softmax izlazni sloj

According to Maharjan *et al.* (2020) and Bridle, (1989), the softmax output activation function effectively classifies multiple classes apart from the binary classification. The WOOD-AUTH dataset has 12 classes and is called various classes. Therefore, the softmax activation function has been selected as the output function. Abd-Ellah *et al.* (2018) presented the softmax function formula in Eq. 6.

$$softmax = \frac{\exp(a_{\rm r})}{\sum_{j=1}^{k} \exp(a_j)}$$
(6)

Where a is the input vector with k dimensions, and y is the output vector with k dimensions.

#### 4.3 Cross-entropy

#### 4.3. Unakrsna entropija

The cross-entropy was used as a loss function of the model for the multi-classification. (Zhang and Sabuncu, 2018) The cross-entropy loss function was defined as a loss training process in the final layer. The multi-classification of the loss (error) function is presented in Eq. 7.

$$cross - entropy = -\sum_{i=1}^{n} \sum_{j=1}^{k} t_{ij} lny_j(x_i, \theta)$$
(7)

Where  $\theta$  is the parameter vector,  $t_{ij}$  is the index that *i* sample related to *j* class, and  $y_i(x_i, \theta)$  is the output.



**Figure 2** Proposed 2D CNN model structure. Red colour rectangular shape refers to input dataset, pink layered rectangular shape corresponds to convolutional layers, green rectangular shape refers to batch normalization layers, yellow rectangular shape presents max-pooling layers, and the last blue rectangular shape is fully connected layer. The last layer is the output softmax layer. 2D CNN model hyperparameters are detailed in Sections 4.1 - 4.4.

Slika 2. Predložena struktura 2D CNN modela. Crveni pravokutni oblik odnosi se na ulazni skup podataka, ružičasti slojeviti pravokutni oblik odgovara konvolucijskim slojevima, zeleni pravokutnik odnosi se na seriju normalizacijskih slojeva, žuti pravokutni oblik predočuje slojeve maksimalnog udruživanja, a posljednji plavi pravokutni oblik odnosi se na potpuno povezani sloj. Zadnji sloj je izlazni softmax sloj. Hiperparametri 2D CNN modela detaljno su opisani u poglavljima 4.1. – 4.4.

#### 4.4 Adam optimizer

#### 4.4. Adamov optimizator

According to Bock *et al.* (2018), the Adam (adaptive moment estimation) optimizer is a well-known algorithm in gradient descent optimization. The Adam optimizer is distinguished in terms of faster optimization in the deep neural network. The optimizer can converge much faster for deeper networks and convolutional neural networks.

#### 5 RESULTS AND DISCUSSION 5. REZULTATI I RASPRAVA

The model was created based on the WOOD-AUTH to classify the 12 wood species. This model accomplished the task successfully. Apart from the proposed model, some popular transfer learning models have been applied to the dataset to show the strength of the proposed model. The 2D CNN model was outperformed in terms of accuracy and computation time compared to the Efficient B3 and Mobile Net transfer learning models. On the other hand, the outcomes of the proposed model can be compared to the previous applications in the literature. The proposed model performance was evaluated using accuracy metric, precision, recall, F1 score and computation time. Figure 3 demonstrates the confusion matrix of the 12 wood species distribution regarding the predicted and real classes. The 2D CNN model had an accuracy of 94 %, precision of 87 %, recall of 81 % and F1 score of 80 %. The metrics have been explained in Section 3.2. Table 2 demonstrates that the model outcomes are based on the metrics. Figure 3 illustrates the model training and loss relationship. It can be seen that the model worked properly with 50 epochs.

The proposed model training and cross-entropy loss function can be seen in Figure 4. Figure 4 illustrates the model behaviour with the loss function.

It can be noticed that the model was not exposed to overfitting, and the number of datasets was enough to train the model. In addition to Figure 4, Table 2 presents and compares the proposed model outcomes to other popular transfer learning models.

### 5.1 Model comparison

#### 5.1. Usporedba modela

The comparison indicates that the proposed model provided fast computation (112 min 27 sec) while obtaining high accuracy (94 %). The transfer learning model Xception accuracy is slightly higher than that of the proposed model; however, its computation time is higher than that of the offered model (the proposed model is 41.07 % faster than the Xception). The 2D CNN model is stronger than the VGG19 transfer learning in evaluation metrics and computation. The proposed model provided a reasonable result for the classification of wood species. This research created a new 2D CNN to classify the WOOD-AUTH dataset. In addition to the proposed model, some popular transfer

	wood1	<b>180</b> 11.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	1 0.1%	<b>0</b> 0.0%	7 0.4%	<b>0</b> 0.0%	<b>0</b> 0.0%	95.7% 4.3%
	wood10	<b>0</b> 0.0%	<b>88</b> 5.4%	<b>1</b> 0.1%	<b>1</b> 0.1%	<b>0</b> 0.0%	<b>2</b> 0.1%	<b>1</b> 0.1%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>2</b> 0.1%	<b>0</b> 0.0%	92.6% 7.4%
	wood11	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>69</b> 4.2%	<b>9</b> 0.6%	1 0.1%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	87.3% 12.7%
	wood12	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>23</b> 1.4%	<b>181</b> 11.1%	<b>1</b> 0.1%	<b>0</b> 0.0%	<b>0</b> 0.0%	6 0.4%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>1</b> 0.1%	85.4% 14.6%
isa	wood2	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>56</b> 3.4%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>D</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	100% 0.0%
na kla	wood3	<b>27</b> 1.7%	<b>0</b> 0.0%	<b>15</b> 0.9%	<b>20</b> 1.2%	<b>0</b> 0.0%	<b>290</b> 17.8%	<b>2</b> 0.1%	<b>18</b> 1.1%	<b>9</b> 0.6%	<b>10</b> 0.6%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>74.2%</b> 25.8%
' izlazı	wood4	<b>34</b> 2.1%	<b>0</b> 0.0%	<b>1</b> 0.1%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>12</b> 0.7%	<b>117</b> 7.2%	<b>0</b> 0.0%	<b>1</b> 0.1%	<b>43</b> 2.6%	<b>1</b> 0.1%	<b>1</b> 0.1%	55.7% 44.3%
class	wood5	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>74</b> 4.5%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	100% 0.0%
utput	wood6	<b>4</b> 0.2%	<b>0</b> 0.0%	0 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>2</b> 0.1%	<b>0</b> 0.0%	<b>34</b> 2.1%	<b>120</b> 7.4%	<b>9</b> 0.6%	<b>0</b> 0.0%	<b>1</b> 0.1%	<b>70.6%</b> 29.4%
õ	wood7	<b>0</b> 0.0%	<b>0</b> 0.0%	0 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>D</b> 0.0%	<b>0</b> 0.0%	<b>23</b> 1.4%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>100%</b> 0.0%
	wood8	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>1</b> 0.1%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>3</b> 0.2%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>63</b> 3.9%	<b>0</b> 0.0%	94.0% 6.0%
	wood9	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>3</b> 0.2%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>0</b> 0.0%	<b>63</b> 3.9%	95.5% 4.5%
		73.5% 26.5%	<b>100%</b> 0.0%	62.7% 37.3%	85.8% 14.2%	96.6% 3.4%	94.8% 5.2%	97.5% 2.5%	53.2% 46.8%	92.3% 7.7%	25.0% 75.0%	95.5% 4.5%	95.5% 4.5%	81.2% 18.8%
		Moodi	wood10	MOODIN	M00012	MOODY	WOOds	NOODA	wood5	w0000	MOODI	WOOD8	N00009	
						Farge	t class	/ cilj	ana kl	asa				

Confusion Matrix / matrica zabune

**Figure 3** Confusion matrix of 12 wood species with accuracy metrics **Slika 3.** Matrica zabune za 12 vrsta drva s metrikom točnosti



Figure 4 Model training and loss relationship in 50 epochs. Blue line refers to training accuracy, and black line corresponds to test accuracy. Orange bottom line refers to training loss, while black line refers to test loss
Slika 4. Model treninga i odnosa gubitka u 50 epoha. Plava se linija odnosi na točnost treninga, a crna odgovara točnosti testa. Narančasta donja crta odnosi se na gubitak treninga, a donja crna crta predočuje gubitak testa

learning models (Kumar, 2023) have been implemented to show the performance of the newly configurated 2D CNN model. In that case, Efficient B3 and Mobile Net transfer learning models were applied using Tensorflow Hub information.

On the one hand, Xception, Inception V3, ResNet50 and VGG19 (Kırbaş and Çiftçi, 2022) published in the literature were compared to the proposed model. The model's strength is its faster computation time, higher evaluation metrics and fewer convolutional layers. In contrast to the proposed model, the major shortcomings of the transfer learning models is the need for high memory size and slow training; their network architecture weights are exceptionally high.

# 6 LIMITATIONS

#### 6. OGRANIČENJA

Even though the proposed model is superior to the previous studies, it has some limitations; trial and error hyperparameter tuning and demanding training required to obtain the proper results, compared to the pre-trained models. The data augmentation can be applied to see the model performance on the balanced dataset. In the future, the model will be saved and applied to different wood species datasets. The model can be converted to an application using deep learning deployment techniques used by non-technical, model accomplished, users.

1										
Model name Naziv modela	Accuracy, % Točnost, %	<b>Precision, %</b> Preciznost, %	<b>Recall, %</b> <i>Opoziv,</i> %	<b>F1 Score, %</b> <i>F1 rezultat,</i> %	<b>Computation time</b> Vrijeme izračuna					
<b>The proposed 2D CNN model</b> predloženi 2D CNN model	94	87	81	80	112 min 27 sec					
Efficient B3	82	87	77	81	3 h 24 min					
Mobile Net	72	79	66	72	5 h 35 min					
Transfer learning models in literature (Kırbaş and Çiftçi, 2022)										
2	Transferni modeli i	učenja u literaturi	(Kırbaş i Çiftçi,	2022.)						
Xception	95	95	95	95	2 h 38 min 6 s					
Inception V3	88	87	87	86	2 h 23 min 34s					
ResNet50	37	58	56	57	2 h 6 m 13s					
VGG19	61	20	31	23	7 h 30 m 45s					

Table 2 Model comparisonTablica 2. Usporedba modela

# 7 CONCLUSIONS

# 7. ZAKLJUČAK

This research used the new 2D CNN model to employ the WOOD-AUTH dataset with multiple classes (12 wood species classes). The new 2D CNN model required less computation time with 8160 image datasets than the heavy pre-trained transfer learning models. Therefore, the study aims to release a new alternative to deep learning for future classification tasks.

The main conclusions of this research are:

Compared to previous studies, this research worked with the newly configurated 2D CNN network based on 12 wood species. High accuracy and fast computation time were acquired using fewer numbers (three layers) of the convolutional neural network.

The proposed model achieved 94 % accuracy, 87 % precision, 81 % recall, 80 % F1 score and 112 min 27 sec computation time.

Compared to the transfer learning models, the 2D CNN model worked with fewer epochs in training.

The model does not require a high memory size due to the lightweight network structure. Otherwise, transfer learning models need a high memory size, e.g. VGG19 requires 549 megabytes (M.B.) (Keras, 2022).

#### **5 REFERENCES**

5. LITERATURA

- Abd-Ellah, M. K.; Awad, A. I.; Khalaf, A. A.; Hamed, H. F., 2018: Two-phase multi-model automatic brain tumour diagnosis system from magnetic resonance images using convolutional neural networks. EURASIP Journal on Image and Video Processing, 2018 (1): 1-10. https://doi. org/10.1186/s13640-018-0332-4
- Ajit, A.; Acharya, K.; Samanta, A., 2020: A review of convolutional neural networks. In Proceedings of 2020 international conference on emerging trends in information technology and engineering (ic-ETITE), pp. 1-5. https://doi.org/10.1109/ic-ETITE47903.2020.049
- Albawi, S.; Mohammed, T. A., Al-Zawi, S., 2017: Understanding of a convolutional neural network. In: Proceedings of 2017 International Conference on Engineering and Technology (ICET), Antalya, Turkey, pp. 1-6. https:// doi.org/10.1109/ICEngTechnol.2017.8308186
- Barmpoutis, P., 2017: Contribution and combination of different wood sections in species recognition using image texture analysis methods. Pro Ligno, 13 (4).
- Barmpoutis, P.; Dimitropoulos, K.; Barboutis, I.; Grammalidis, N.; Lefakis, P., 2018: Wood species recognition through multidimensional texture analysis. Computers and Electronics in Agriculture, 144: 241-248. https://doi.org/10.1016/j.compag.2017.12.011
- Bock, S.; Goppold, J.; Weiß, M., 2018: An improvement of the convergence proof of the ADAM-Optimizer. arXiv preprint arXiv:1804.10587. https://doi.org/10.48550/ arXiv.1804.10587
- Bock, S.; Goppold, J.; Weiß, M., 2018: An improvement of the convergence proof of the ADAM-Optimizer. arXiv, Apr. 27, 2018. https://doi.org/10.48550/arXiv.1804.10587

- Brauns, J.; Rocens, K., 2007: Modification of wood: mechanical properties and application. In: Buschow, K. H. J.; Cahn, R. W.; Flemings, M. C.; Ilschner, B.; Kramer, E. J.; Mahajan, S.; Veyssi'ere, P. (eds.), 2007: Encyclopedia of Materials: Science and Technology. Elsevier, pp. 1–9. https://doi.org/10.1016/B978-008043152-9.02174-6
- Bridle, J., 1983: Training stochastic model recognition algorithms as networks can lead to maximum mutual information estimation of parameters. In: Advances in Neural Information Processing Systems, Morgan-Kaufmann (online). https://proceedings.neurips.cc/paper/1989/hash /0336dcbab05b9d5ad24f4333c7658a0e-Abstract.html (Accessed Jul. 03, 2023)
- de Geus, A.; da Silva, S. F.; Gontijo, A. B.; Silva, F. O.; Batista, M. A.; Souza, J. R., 2020: An analysis of timber sections and deep learning for wood species classification. Multimedia Tools and Applications, 79 (45): 34513-34529. https://doi.org/10.1007/s11042-020-09212-x
- Dhillon, A.; Verma, G. K., 2020: Convolutional neural network: a review of models, methodologies and applications to object detection. Progress in Artificial Intelligence, 9 (2): 85-112. https://doi.org/10.1007/s13748-019-00203-0
- Fabija'nska, A.; Danek, M.; Barniak, J., 2021: Wood species automatic identification from wood core images with a residual convolutional neural network. Computers and Electronics in Agriculture, 181: 105941. https://doi. org/10.1016/j.compag.2020.105941
- Fuentealba, C.; Simon, C.; Choffel, D.; Charpentier, P.; Masson, D., 2004: Wood products identification by internal characteristics readings. In: Proceedings of 2004 IEEE International Conference on Industrial Technology, IEEE ICIT'04, 2: 763-768. https://doi.org/10.1109/ ICIT.2004.1490171
- Gani, W.; Limam, M., 2013: Performance evaluation of one-class classification-based control charts through an industrial application. Quality and Reliability Engineering International, 29 (6): 841-854. https://doi. org/10.1002/qre.1440
- Hafemann, L. G.; Oliveira, L. S.; Cavalin, P., 2014: Forest species recognition using deep convolutional neural networks. In: Proceedings of 22<sup>nd</sup> international conference on pattern recognition, pp. 1103-1107. https://doi. org/10.1109/ICPR.2014.199
- Huang, P.; Zhao, F.; Li, X.; Wu, Z.; Zhu, Z.; Zhang, Y., 2020: Variant transfer learning for wood recognition. In: Proceedings of 2020 International Conferences on Internet of Things (iThings) and IEEE Green Computing and Communications (GreenCom) and IEEE Cyber, Physical and Social Computing (CPSCom) and IEEE Smart Data (SmartData) and IEEE Congress on Cybermatics (Cybermatics), 2 – 6 November 2020, Rhodes, Greece, pp. 743-748. https://doi.org/10.1109/iThings-GreenCom-CP-SCom-SmartData-Cybermatics50389.2020.00128
- Huang, P.; Zhao, F.; Li, X.; Wu, Z.; Zhu, Z.; Zhang, Y., 2020: Variant transfer learning for wood recognition. In: Proceedings of 2020 International Conferences on Internet of Things (iThings) and IEEE Green Computing and Communications (GreenCom) and IEEE Cyber, Physical and Social Computing (CPSCom) and IEEE Smart Data (SmartData) and IEEE Congress on Cybermatics (Cybermatics), pp. 743-748. https://doi.org/10.1109/iThings-GreenCom-CPSCom-SmartData Cybermatics50389. 2020.00128
- Huang, Y.; Meng, S.; Hwang, S. W.; Kobayashi, K.; Sugiyama, J., 2020: Neural network for classification of

Chinese zither panel wood via near-infrared spectroscopy. BioResources, 15 (1): 130-141.

- Jannat, T.; Sayeed, A.; Afrin, S., 2021: Supervised linear discriminant analysis for dimension reduction and hyperspectral image classification method based on 2d-3d cnn. In: Proceedings of 2021 International Conference on Automation, Control and Mechatronics for Industry 4.0 (ACMI), pp. 1-6. https://doi.org/10.1109/ ACMI53878.2021.9528191
- Jones, D.; Brischke, C., 2017: Performance of bio-based building materials. Woodhead Publishing, Elsevier. https://doi.org/10.1016/C2015-0-04364-7
- Kasal, B., 2004: Mechanical properties of wood. In: Burley, J.; Evans, J.; Youngquist, J. A. (eds.), Encyclopedia of Forest Sciences. Elsevier, Oxford. https://doi.org/10.1016/B0-12-145160-7/00041-7
- Kilic, K.; Toriya, H.; Kosugi, Y.; Adachi, T.; Kawamura, Y., 2022: One-dimensional convolutional neural network for pipe jacking EPB TBM cutter wear prediction. Applied Sciences, 12 (5): 2410. https://doi.org/10.3390/ app12052410
- Kırbaş, İ.; Çifci, A., 2022: An effective and fast solution for classification of wood species: A deep transfer learning approach. Ecological Informatics, 69: 101633. https://doi.org/10.1016/j.ecoinf.2022.101633
- 24. Kobayashi, K.; Akada, M.; Torigoe, T.; Imazu, S.; Sugiyama, J., 2015: Automated recognition of wood used in traditional Japanese sculptures by texture analysis of their low-resolution computed tomography data. Journal of Wood Science, 61: 630-640. https://doi.org/10.1007/ s10086-015-1507-6
- Kumar, S., 2023: Top 10 Pre-trained models for image embedding every data scientist should know. Medium, Apr. 21 (online). https://towardsdatascience.com/top-10-pre-trained-models-for-image-embedding-every-data-scientist-should-know-88da0ef541cd (Accessed Jul. 02, 2023).
- 26. Kwon, O.; Lee, H. G.; Lee, M. R.; Jang, S.; Yang, S. Y.; Park, S. Y.; Yeo, H., 2017: Automatic wood species identification of Korean softwood based on convolutional neural networks. Journal of the Korean Wood Science and Technology, 45 (6): 797-808. https://doi.org/10.5658/ WOOD.2017.45.6.797
- Maharjan, S.; Alsadoon, A.; Prasad, P. W. C.; Al-Dalain, T.; Alsadoon, O. H., 2020: A novel enhanced softmax loss function for brain tumour detection using deep learning. Journal of Neuroscience Methods, 330: 108520. https://doi.org/10.1016/j.jneumeth.2019.108520
- Manik, F. Y.; Saputra, S.; Ginting, D. S. B., 2020: Plant classification based on extraction feature gray level cooccurrence matrix using k-nearest neighbour. Journal of Physics: Conference Series, 1566: 012107. https://doi. org/10.1088/1742-6596/1566/1/012107
- 29. Mohan, S.; Venkatachalapathy, K.; Sudhakar, P., 2014: An intelligent recognition system for identification of wood species. Journal of Computational Science, 10 (7): 1231-1237. https://doi.org/10.3844/jcssp.2014.1231.1237
- Nguyen, T. T.; Ji, X.; Nguyen, T. H. V.; Guo, M., 2017: Wettability modification of heat treated wood (HTW) via cold atmospheric-pressure nitrogen plasma jet (APPJ). Holzforschung, 72 (1): 37-43. https://doi.org/10.1515/hf-2017-0004
- O'Shea, K.; Nash, R., 2015: An introduction to convolutional neural networks. arXiv: 1511.08458. https://doi. org/10.48550/arXiv.1511.08458

- 32. Popescu, M. C.; Popescu, C. M.; Lisa, G.; Sakata, Y., 2011: Evaluation of morphological and chemical aspects of different wood species by spectroscopy and thermal methods. Journal of Molecular Structure, 988 (1-3): 65-72. https://doi.org/10.1016/j.molstruc.2010.12.004
- Powers, D. M., 2020: Evaluation: from precision, recall and F-measure to ROC, informedness, markedness and correlation. arXiv preprint arXiv:2010.16061. https:// doi.org/10.48550/arXiv.2010.16061
- 34. Rajagopal, H.; Khairuddin, A. S. M.; Mokhtar, N.; Ahmad, A.; Yusof, R., 2019: Application of image quality assessment module to motion-blurred wood images for wood species identification system. Wood Science and Technology, 53 (4): 967-981. https://doi.org/10.1007/ s00226-019-01110-2
- Ravindran, P.; Costa, A.; Soares, R.; Wiedenhoeft, A. C., 2018: Classification of CITES-listed and other neotropical Meliaceae wood images using convolutional neural networks. Plant Methods, 14: 1-10. https://doi. org/10.1186/s13007-018-0292-9
- 36. Santosa, S.; Pramunendar, R. A.; Prabowo, D. P.; Santosa, Y. P., 2019: Wood types classification using back-propagation neural network based on genetic algorithm with gray level co-occurrence matrix for features extraction. IAENG International Journal of Computer Science, 46 (2).
- Simonyan, K.; Zisserman, A., 2015: Very deep convolutional networks for large-scale image recognition. arXiv:1409.1556v6. https://doi.org/10.48550/arXiv.1409.1556
- 38. Sinaice, B. B.; Owada, N.; Ikeda, H.; Toriya, H.; Bagai, Z.; Shemang, E.; Kawamura, Y., 2022: Spectral angle mapping and A. I. methods applied in automatic identification of Placer deposit magnetite using multispectral camera mounted on UAV. Minerals, 12 (2): 268. https:// doi.org/10.3390/min12020268
- 39. Souza, D. V.; Santos, J. X.; Vieira, H. C.; Naide, T. L.; Nisgoski, S.; Oliveira, L. E. S., 2020: An automatic recognition system of Brazilian flora species based on textural features of macroscopic images of wood. Wood Science and Technology, 54 (4): 1065-1090. https://doi. org/10.1007/s00226-020-01196-z
- 40. Sun, Y.; Cao, Y.; Xiong, F., Yue, X.; Qiu, J.; He, X.; Zhao, F., 2015: The wood slice cell image identification algorithm based on singular value decomposition. Journal of Computational and Theoretical Nanoscience, 12 (12): 5372-5378. https://doi.org/10.1166/jctn.2015.4529
- 41. Sun, Y.; Lin, Q.; He, X.; Zhao, Y.; Dai, F.; Qiu, J.; Cao, Y., 2021: Wood species recognition with small data: a deep learning approach. International Journal of Computational Intelligence Systems, 14 (1): 1451-1460. https:// doi.org/10.2991/ijcis.d.210423.001
- 42. Tang, X. J.; Tay, Y. H.; Siam, N. A.; Lim, S. C., 2018: MyWood-ID: Automated macroscopic wood identification system using smartphone and macro-lens. In: Proceedings of the 2018 International Conference on Computational Intelligence and Intelligent Systems, pp. 37-43. https://doi.org/10.1145/3293475.3293493
- Tou, J. Y.; Lau, P. Y.; Tay, Y. H., 2007: Computer visionbased wood recognition system. In: Proceedings of International Workshop on Advanced Image Technology, pp. 197-202.
- 44. Vacha, P.; Haindl, M., 2013: Wood Variety Recognition on Mobile Devices, 93. ERCIM News.
- 45. Verly Lopes, D. J.; Burgreen, G. W.; Entsminger, E. D., 2020: North American hardwoods identification using

machine-learning. Forests, 11 (3): 298. https://doi. org/10.3390/f11030298

- Wang, H. J.; Zhang, G. Q.; Qi, H. N., 2013: Wood recognition using image texture features. Plos One, 8 (10): e76101. https://doi.org/10.1371/journal.pone.0076101
- Winandy, J. E.; Rowell, R. M., 1984: The chemistry of wood strength. In: Rowell, R. M. (ed.), The Chemistry of Solid Wood. American Chemical Society, Washington, DC, pp. 211-256.
- Yang, J.; Huang, P.; Dai, F.; Sun, Y.; Wang, L.; Bi, H., 2019: Application of deep learning in wood classification. In: Proceedings of 2019 IEEE International Conference on Computer Science and Educational Informatization (CSEI), pp. 124-129. https://doi.org/10.1109/ CSEI47661.2019.8938960
- 49. Zabel, R. A.; Morrell, J. J., 2020: Wood Microbiology: Decay and its Prevention, 2<sup>nd</sup> ed. Academic Press, San

Diego, CA, USA, pp. 149-183. https://doi.org/10.1016/ B978-0-12-819465-2.00006-1

- Zhang, Z.; Sabuncu, M. R., 2018: Generalized Cross Entropy Loss for Training Deep Neural Networks with Noisy Labels. arXiv:1805.07836v4. https://doi. org/10.48550/arXiv.1805.07836
- Zhao, P., 2013: Robust wood species recognition using variable color information. Optik – International Journal for Light and Electron Optics, 124: 2833-2836. https:// doi.org/10.1016/j.ijleo.2012.08.058
- \*\*\*Keras documentation: Keras Application: 2022 (online). https://keras.io/api/applications/ (Accessed Dec. 13, 2022)
- 53. \*\*\*TensorFlow Hub, TensorFlow: 2022 (online). https://www.tensorflow.org/hub (Accessed Dec. 13, 2022).

#### **Corresponding address:**

#### KENAN KILIÇ

Gazi University, Graduate School of Natural and Applied Sciences, Department of Wood Products Industrial Engineering, 06500, Ankara, TURKEY, e-mail: kenan.kilic@bozok.edu.tr

# SVEUČILIŠTE U ZAGREBU • FAKULTET ŠUMARSTVA I DRVNE TEHNOLOGIJE

UNIVERSITY OF ZAGREB • FACULTY OF FORESTRY AND WOOD TECHNOLOGY

Svetošimunska 23, 10000 Zagreb, Hrvatska / *Croatia* tel: 00385 1 2352 454 e-mail: dcvetan@sumfak.hr IBAN: HR0923600001101340148 OIB / VAT ID: HR07699719217

# PRETPLATNI LIST / SUBSCRIPTION SHEET

Uskoro izlazi **volumen 75 (2024. godina)** časopisa Drvna industrija. Pozivamo Vas da obnovite svoju pretplatu ili se pretplatite na časopis, te na taj način pomognete njegovo izdavanje. Cijena sva četiri broja jednog godišta (volumena) je 55,00 EUR bez PDV-a. Ukoliko ste suglasni s uvjetima pretplate za jedno godište časopisa, molimo Vas da popunite obrazac za pretplatu i pošaljete ga na našu poštansku ili elektroničku adresu.

*Volume 75 (2024)* of the journal Drvna industrija will be published soon. We invite you to renew your subscription or subscribe to a journal to support it. The price of all four issues of one year (volume) is 55 EUR without VAT. If you agree to the subscription terms for one volume, please complete the subscription form and send it to our postal or e-mail address.

Predsjednik Izdavačkog savjeta časopisa Drvna industrija President of Publishing Council

izv. prof. dr. sc. Vjekoslav Živković

Glavna i odgovorna urednica časopisa Drvna industrija *Editor-In-Chief* 

prof. dr. sc. Ružica Beljo Lučić

or Copy/Scan



Pretplaćujemo se na časopis *Drvna industrija* u količini od \_\_\_\_ godišnje pretplate (četiri broja). Cijena jednog godišta (volumena) iznosi 55 EUR, bez PDV-a. Pretplata obuhvaća sve brojeve jednog godišta. *We subscribe to the journal* Drvna industrija *in amount of \_\_\_\_ annual subscription(s) (four issues). Price of one volume (year) is 55 EUR, without VAT. The subscription covers all issues of one volume.* 

Hrvatska: HR0923600001101340148 s naznakom "Za časopis Drvna industrija" poziv na broj: 3-02-03 EU / World: Bank: Zagrebačka banka IBAN: HR0923600001101340148 Swift: ZABA HR 2X

Osoba / Name:	e-mail:
Tvrtka, ustanova / Company, institution:	
OIB / VAT ID:	Telefon / Phone:
Adresa / Address:(ulica / street)	Pošta. broj: Postal code:
Grad / <i>City</i> :	Regija / <i>Region</i> :
Država / Country:	

# Cebrail Açık<sup>1</sup> Modeling of Color Design on Furniture Surfaces with CNC Laser Modification

# Modeliranje dizajna boje površine namještaja CNC laserskom modifikacijom

# **ORIGINAL SCIENTIFIC PAPER**

**Izvorni znanstveni rad** Received – prispjelo: 14. 3. 2023. Accepted – prihvaćeno: 18. 10. 2023. UDK: 674.07 https://doi.org/10.5552/drvind.2023.0096 © 2023 by the author(s). Licensee University of Zagreb Faculty of Forestry and Wood Technology. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license.

**ABSTRACT** • Today, handcrafted furniture surface treatment techniques are less used due to application difficulties and high costs. Recently, the use of CNC router processing machines has resulted in a revival of these techniques. However, since this method is insufficient in micro-processes and color modifications that require precision, these processes have started to be performed with lasers. In this research, the beech wood surface was processed using a CNC laser processing machine with a carbon dioxide gas tube by applying different engraving power and engraving speeds. Product design and manufacturing parameters were determined in the CIE L\*a\*b\* color system using the regression modeling method. A sample application was made by applying laser engraving to a furniture surface designed according to the obtained parameters. As a result of the study, it has been explained that using L\* color group regression modeling method and CAD/CAM supported laser technology in furniture top surface color design techniques can be applied with laser to furniture designed for CNC laser production.

KEYWORDS: laser pyrography; furniture surfaces; color design

**SAŽETAK** • Primjena ručnih tehnika površinske obrade namještaja smanjila se zbog otežane izvedbe i visokih troškova. Ručne su se tehnike nedavno pokušale oživjeti obradom CNC glodalima. Međutim, CNC glodanje nije odgovarajuća tehnika za mikroprocese i modifikacije boja koje zahtijevaju preciznost, pa su se ručne tehnike počele provoditi laserima. U ovom je istraživanju površina bukovine obrađena CNC laserom s plinskom cijevi od ugljikova dioksida, uz primjenu različite snage i brzine graviranja. Dizajn proizvoda i proizvodni parametri određeni su u sustavu boja CIE L\*a\*b\* primjenom metode regresijskog modeliranja. Na površini namještaja laserskim je graviranjem izrađen uzorak dizajniran prema dobivenim parametrima. Rezultatima istraživanja utvrđeno je da je primjena L\* komponente boje u regresijskom modeliranju, potpomognuta CAD/CAM laserskom tehnologijom, prikladna u procesima dizajniranja boje površine namještaja unutar inženjerskog pristupa u industriji. Utvrđeno je da se na namještaju dizajniranom za CNC lasersku proizvodnju laserom mogu provoditi mnoge tehnike dizajniranja boje površine.

KLJUČNE RIJEČI: laserska pirografija; površine namještaja; dizajn boje

<sup>&</sup>lt;sup>1</sup> Author is researcher at Ministry of National Education Onikisubat District Directorate of National Education, Kahramanmaraş, Turkey. https://orcid.org/0000-0002-1094-6946

### **1 INTRODUCTION**

#### 1. UVOD

One of the most important issues in laser wood surface treatment is to be able to explain the reaction methodology of wood surfaces against laser beams. Laser engraving on wooden surfaces is generally done to increase the commercial and aesthetic value of furniture or wooden products. Furniture surface decoration is the evaluation of furniture with color, shape, painting and motifs by using various techniques in order to add beauty without disturbing its function. Furniture, which has been in continuous development since ancient times, continued to change in terms of technology, concept and function at the beginning of the 20th century. With the 20th century, industrial society was formed and manual labor lost its previous validity. In the 20th century, it is seen that furniture design movements and architectural movements overlap and affect each other (Çifçi and Demirarslan, 2021). Technology offers the opportunity to reach the development of art techniques, which have become a part of culture and innovations, in a practical, fast and easy way. While art is affected by the development of technology, industry and industrial products also benefit from art and design (Aytepe, 2013). In order to produce the parts of the furniture designed with CAD systems in the computer environment, the designs of these parts can be directed to the CAM systems. In the laser cutting and engraving processes of wood and wood-based materials, the probability of faulty production is less than in the machining made with traditional techniques and using cutters in CNC (router) machines. In addition, in laser processing of wood materials, very precise and microprocesses that cannot be performed on CNC machines with cutters can be easily performed (Karabıyık, 2016).

Studies on color design by laser modification of wood surfaces are not new. Jurek and Wagnerova (2021) investigated the transfer of non-vector photographic images in different color tones to the wooden surface by laser engraving. Kubovsky and Babiak (2009) investigated the changes in the color system of maple (Acer pseudoplatanus, L.) and beech (Fagus sylvatica, L.) wood surface according to different irradiation doses in the CIE  $L^*a^*b^*$  color system. The changes of scanning speed, scanning power and nozzle height parameters in the CIE  $L^*a^*b^*$  color system in laser engraving of poplar wood were investigated using the mathematical modeling method according to the response surface method (RSM) (Li et al., 2018). Reinprecht and Vidholdov (2021) analyzed the effects of CO<sub>2</sub> laser etched surfaces on adhesion resistance after coating with a polymer layer of polyvinyl acetate (PVAc) and polyurethane (PUR) at different irradiation doses for the modification of European beech (Fagus sylvatica L.) and Norway spruce

(*Picea abies L.*) wood surfaces. CIE  $L^*a^*b^*$  color measurements were performed on samples obtained from beech, ash, linden and spruce tree species to evaluate aesthetic changes due to different laser beam intensity and the number of laser points (points/inch) on the surface (Petutschnigg et al., 2013). In order to examine the photodegradation of wood, two different lasers (KrFand XeCl-laser) were etched and the changes in wood color resulting from heat treatment after laser irradiation were investigated by DRIFT (Diffuse Reflectance Infrared Fourier Transform) spectroscopy and color measurement of the changes on the wood surface (Mitsui et al., 2005). Color changes on aspen, birch, spruce, pine, red beech, oak, alder, birch, plywood and MDF surfaces using different power and speed laser combinations were visually examined (Teivonen, 2016). The color changes of the effect of CO<sub>2</sub> laser engraving speed at different rates on sycamore maple (Acer pseudoplatanus L.) were investigated by making CIE  $L^*a^*b^*$  color measurements (Petru and Lunguleasa, 2017). Gurau et al. (2017) investigated the effects of CO, laser beam power output and scanning speed on the surface roughness and color changes in samples obtained from beech (Fagus sylvatica) wood. The effects of CO<sub>2</sub> laser modification of beech wood surfaces modified with Aspergillus niger and Penicillium brevicompactum molds using different (J/8) radiation dose and power were investigated in the CIE  $L^*a^*b^*$  system (Vidholdova *et al.*, 2017). The color changes of the effect of CO<sub>2</sub> laser engraving speed and irradiation dose (Joule) at different rates on beech wood were investigated by making CIE  $L^*a^*b^*$  color measurements (Kacik and Kubovsky, 2011). Kudela et al. (2020) investigated the variation of color, roughness and waviness parameters of the beech wood surface depending on the CO<sub>2</sub> laser engraving power and raster density. Kubovsky and Kacik (2014) investigated the changes in color and wood chemical components caused by irradiation dose of lime (Tilia vulgaris L.) tree with CO, laser beam at different rates.

Until today, many studies on color design on wood surface treatment with laser have been made. Studies on the effect of laser surface modification of wood on color changes have generally focused on color change rate parameters. These studies have generally remained at the experimental level and their application in furniture design and production process has not been explained. In this study, experimental studies were carried out on a medium-sized CNC laser machine, which is widely used, by using beech wood, which is frequently used in industrial production. In line with the findings, furniture surface decoration color design and application were made with the regression modeling method, which is important in industrial production. An original study was carried out by presenting the material, machine, production method, design and manufacturing processes required by the industrial production sector as a whole.

# 2 MATERIALS AND METHODS

# 2. MATERIJALI I METODE

# 2.1 Materials

#### 2.1. Materijali

In the study, beech wood with a density of 721 kg/m<sup>3</sup> in air-dried condition at 12 % humidity was used. The samples were prepared from randomly selected 1st class wood material, with smooth fibers, no knots, no cracks, no difference in color and density, and sapwood parts. Experimental specimens measuring 10 mm × 100 mm × 100 mm were cut and sanded with 120 grit sandpaper. In material processing, engraving processes were carried out on the experimental samples in a CNC laser machine with a maximum power output of 130 watts, a carbon dioxide gas tube, 2 inch (50.8 mm) focal length lens, water-cooled, 1.5 mm nozzle diameter and 10.6  $\mu$ m wavelength.

#### 2.2 Methods

#### 2.2. Metode

The factorial experimental design method was preferred in the study. With this method, it was possible to evaluate each of the variable factors together with each other and to determine the extent of the effect of each variable on the event. In addition, as a result of the interaction of the variables with each other, different behaviors that the factors may show can be determined. There are many factors that affect laser processing performance. According to the information obtained from the experimental studies carried out so far, it is aimed to determine the effect of the factors that the user is allowed to access, by keeping the factors belonging to the system constant. In this study, 10 %, 40 % and 70 % laser power and engraving speeds of 200, 350 and 500 mm/s were applied to wood samples in a CNC laser machine, taking into account the limitations of machine processing functions. There are 2 variable parameters in the experimental design and each of them takes 3 different values. According to the specified levels, Taguchi orthogonal design is the full factorial design, with 27 data for each type of laser engraving and can be called L27(23). This notation indicates that the number of experiments is 27, with 2 factors having 3 different levels.

In the tests performed, the experimental data were evaluated using SPSS statistics version 22. The interaction of the laser engraving speed and laser engraving power dependent variables and the color independent variable were analyzed. For each analysis performed, multiple linear regression analysis was preferred to determine whether there was a significant difference. Preliminary estimates made to achieve optimum yield quality in production are calculated according to the regression Eq. 1.

$$\hat{\mathbf{Y}} = \boldsymbol{\beta}_0 + \boldsymbol{\beta}_1 \cdot \boldsymbol{X}_1 + \boldsymbol{\beta}_2 \cdot \boldsymbol{X}_2 \tag{1}$$

Where;

- $\hat{Y}$  dependent variable (color value),
- $\beta_0$  donstant beta value,
- $\beta_1$  1st independent variable (engraving power) beta value,
- $X_1$  value of 1st independent variable (Engraving Power Watt%),
- $\beta_2$  2nd independent variable (engraving speed) beta value,
- $X_2$  value of 2nd independent variable (engraving speed mm/s) defined.

#### 2.3 Color measurement

#### 2.3. Mjerenje boje

For color measurements, CIELAB color range was defined by the International Commission on Illumination (CIE) in 1976 according to the principles specified in ASTM D 2244. In this system, color is represented as a point in 3 dimensions on the x, y and z



**Figure 1** (a) Example of laser engraving experiment, (b) structure of axes  $L^*$ ,  $a^*$ ,  $b^*$  in the coordinate system CIELAB (Kubovsky and Babiak, 2009)

**Slika 1.** (a) Primjer eksperimenta laserskoga graviranja, (b) struktura osi  $L^*$ ,  $a^*$ ,  $b^*$  u koordinatnom sustavu CIELAB (Kubovsky i Babiak, 2009.)

axis. In the CIE  $L^*a^*b^*$  color system, the differences in colors and their locations are determined according to the  $L^*$ ,  $a^*$ ,  $b^*$  color coordinates. Here,  $L^*$  is on the black-white ( $L^{*=0}$  for black,  $L^{*=100}$  for white) axis,  $a^*$  is on the red-green (red is positive, green is negative) axis,  $b^*$  is on yellow-blue (yellow is positive, blue is negative) axis. In addition, in the  $bi^*$ ,  $br^*$ ,  $ai^*$ , ar,  $Li^*$ ,  $Lr^*$  notations, *i* shows the treated experimental measurement value and *r* shows the untreated reference value. Figure 1.a shows the sample of the experiment which was measured. In the color area shown in Figure 1.b, the  $L^*$  coordinate constitutes the vertical axis (y), the  $a^*$  coordinate constitutes the horizontal (x) axis, and the  $b^*$  coordinate constitutes the (z) axis.

#### 3 RESULTS AND DISCUSSION 3. REZULTATI I RASPRAVA

The laser engraving colors obtained according to the speed (S) and laser processing power (P) test factors in laser engraving applied to the parallel directions of the wood laser engraving samples are shown in Table 1.

In the untreated natural state of the beech massif, the white value  $(L^*)$  on the black and white axis was measured as 76.21, the yellow value  $(a^*)$  on the blueyellow axis was 7.27 and the red value  $(b^*)$  on the green-red axis was measured as 19.94. In laser engraving of wood surface, power increments in all parameters decreased the white value and increased the black value. This situation changed inversely with the engraving speed increments. In other words, the engraving speed increments increased the whiteness value in all parameters. These results have been confirmed in many studies. For example, it has been stated that the scanning speed negatively affects the  $L^*$  change in laser engraving of poplar wood (Li *et al.*, 2018). In the laser engraving of wooden surfaces, the engraving speed increments did not have any effect on the  $a^*$  and  $b^*$  color values in engravings with a power of 10 %. This situation has also been explained by the changes in the wood surface in engraving with XeCl-lacelet to examine the photodegradation of wood (Mitsui et al., 2005). On the other hand, engraving speed increments increased both the  $a^*$  value and the  $b^*$  value in engravings with 40 % and 70 % power. In other words, while the engraving speed at high powers increased the yellow and red color values, it decreased the green and blue color values. This was confirmed in a study made to evaluate the aesthetic changes due to different laser beam intensity and the number of laser points (points/ inch) on the surface of samples obtained from beech trees (Petutschnigg et al., 2013). In wood surface engraving, the color changes in the green-red axis showed the same behavior as the changes in the blue-yellow axis in all parameters. In another study, it was stated that the trends in the blue-yellow and green-red axis of beech wood surfaces treated with laser engraving were the same (Jurek and Wagnerova, 2021). However, in another study, it was stated that the yellow and red tendency of the beech wood surface increased with laser engraving power increments (Kudela et al., 2020). This may be due to operating under 10 % power. As the wood cell structure has not yet been degraded by full laser irradiation, it has been explained that complete carbonization cannot occur (Kubovsky and Kacik, 2014). In laser engraving of the wood surface,  $b^*$  yellow color values were lower than an untreated reference value, below the engraving speed of about 300 mm/s when processing at 40 % engraving power. The same situation occurred in the color changes under the effect of laser engraving speed at different rates on Sycamore maple (Acer pseudoplatanus L.) (Petru and Lunguleasa, 2017). In this study, the engraving power increased both the blue and green color ratio of the

N	<i>P</i> , W	<i>S</i> , mm/s	<i>L</i> *	<i>a</i> *	<i>b</i> *	N	<i>P</i> , W	S, mm/s	<i>L</i> *	<i>a</i> *	<i>b</i> *
1	Coi	ntrol	76.67	7.31	19.72	16	40	40 350		13.71	24.71
2	10	500	68.54	9.92	29.42	17	40	200	22.66	9.16	14.07
3	10	350	65.87	10.65	30.42	18	70	500	34.97	14.75	25.61
4	10	200	59.36	11.66	30.09	19	70	350	29.02	12.44	20.4
5	40	500	43.77	15.38	31.71	20	70	200	20.91	7.87	11.79
6	40	350	34	13.85	25.14	21	Cor	Control		7.32	19.41
7	40	200	24.64	9.29	14.11	22	10	500	66.7	10.38	26.82
8	70	500	35.38	14.43	26.07	23	10	350	62.18	11.00	26.26
9	70	350	28.63	12.41	19.94	24	10	200	52.98	11.97	26.18
10	70	200	21.36	7.61	11.35	25	40	500	36.08	14.53	26.06
11	Cor	ntrol	76.77	7.18	20.71	26	40	350	25.85	10.94	17.31
12	10	500	68.97	9.81	27.93	27	40	200	15.38	5.36	7.71
13	10	350	66.14	10.38	28.54	28	70	500	30.65	12.63	20.55
14	10	200	58.2	12.13	29.95	29	70	350	24.26	9.71	14.79
15	40	500	41.99	15.47	30.99	30	70	200	15.79	4.15	6.34

Table 1 Wood surface laser engraving colorsTablica 1. Boje za lasersko graviranje površine drva



**Figure 2** Surface engraving color changes at (a) 10 % power, (b) 40 % power, (c) 70 % power **Slika 2.** Promjene boje gravirane površine pri (a) 10 % snage, (b) 40 % snage, (c) 70 % snage

wooden surface. The same results were obtained in the modification of beech wood surfaces with laser using different (J/8) radiation doses and power (Vidholdova *et al.*, 2017). In this study, the color changes of the laser-treated surfaces of the beech massif at different engraving speed parameters at 10 % engraving power are shown in Figure 2a, color changes at different engraving speed parameters at 40 % engraving power are shown in Figure 2b, and at 70 % in Figure 2c. The color changes are shown at different engraving speed parameters at 40 % engraving speed parameters are shown in Figure 2b, and at 70 % in Figure 2c. The color changes are shown at different engraving speed parameters and at a specific engraving power.

Multivariate linear regression analysis was performed to determine the effect of engraving power and engraving speed variables on color changes in wood surface laser engraving. The analysis results of the laser engraving process are shown in Table 2 below.

According to the analysis results in Table 2 above, laser engraving power negatively and significantly affected the color of whiteness  $(L^*)$  in the blackwhite axis on the laser engraving surface with an effect size of 80 % ( $pr^2 = 0.802$ ). In other studies, it has been reported that the color changes on the wood surface of beech (Fagus sylvatica, L.) are the mostly  $L^*$  color changes according to different (J·cm<sup>-2</sup>) irradiation doses (Kubovsky and Babiak, 2009). The laser engraving speed affected the white color on the laser engraving surface positively and significantly with an effect size of 41 % ( $pr^2 = 0.412$ ). The regression analysis model was 95 % reliable (p < 0.05), and 68 % of the whiteness color change ( $R^2$ =0.723) was performed by the scratch power and engraving speed variables. Based on the data in Table 2 above, the mathematical modeling re-

Dependent	<i>L</i> *			a*			b*		
Zavisne varijable	β	pr	p	β	pr	p	β	pr	p
Constant / konstanta	46.993	-	0.000	6.372	-	0.000	17.567	-	0.000
Engraving power, W % snaga graviranja, W %	-0.607	-0.896	0.000	-0.004	-0.039	0.850	-0.183	-0.731	0.000
Engraving speed, mm/s brzina graviranja, mm/s	0.050	0.642	0.000	0.014	0.616	0.001	0.035	0.712	0.000

# Table 2 Regression analysis results Tablica 2. Rezultati regresijske analize

*B* – Beta, pr – Partial correlations,  $p > 0.005 / \beta$  – *beta*, pr – *djelomične korelacije*, p > 0.005.

sult of the beech massif was used to theoretically provide optimum laser engraving quality and predict the whiteness color, generating the following regression equation:

#### *L*\*=46,993+(*Engraving power*\*-0.607)+ +(*Engraving speed*\*0.050)

In wood surface engraving, laser engraving power did not affect the redness ( $a^*$ ) color on the green-red axis on the laser engraving surface at the 95 % confidence level. Laser engraving speed affected the red color on the laser engraving surface positively and significantly with an effect size of 38 % ( $pr^2 = 0.379$ ). The regression analysis model was 95 % reliable (p<0.05), and 11 % of the whiteness color change ( $R^2=0.107$ ) was achieved by the engraving power and engraving speed variables. Since the effect of engraving power is meaningless and the impact value of engraving speed is low, the regression equation was not produced to predict the red color of the beech massif as a result of mathematical modeling.

In wood surface engraving, laser engraving power affected the yellowness ( $b^*$ ) color of the blue-yellow axis on the laser engraving surface negatively and significantly with an effect size of 53 % ( $pr^2 = 0.534$ ). The laser engraving speed affected the yellowing color on the laser engraving surface positively and significantly with an effect size of 51 % ( $pr^2 = 0.506$ ). The regression analysis model was 95 % reliable (p<0.05), and 47 % of the yellowness color change ( $R^2$ =0.469) was performed by the engraving power and engraving speed variables. Based on the data in Table 2 above, the mathematical modeling result of the beech massif was used to theoretically ensure optimum laser engraving quality and predict the turban color, generating the following regression equation:

*b*\*=17.657+(*Engraving power*\*-0.183)+ +(*Engraving speed*\*0.035)

# **3.1. Design and implementation** 3.1. Dizajn i implementacija

CNC laser machine works with a computer hardware loaded with CAD (Computer Aided Design) program and CAM (Computer Aided Manufacturing) programs that convert CAD drawings into machine codes. In this study, a case study was conducted to determine the applicability of CNC laser wood surface decoration with regression modeling method. The steps of design and production process of CNC laser operations on the wooden box made of beech massif used in the experiment are explained below.

The engraving paths to be made with the ornament motif forming the wooden box surfaces were drawn in a CAD program. The motif has been transferred to the CAM program in the software of the CNC laser machine as vector. The production design was completed by determining the motif design, and production parameter values were measured according to the box. Since the effect level of the white color  $(L^*)$ value in the above regression analysis in the motif design is much higher than the yellow  $(a^*)$  and red  $(b^*)$ color values, calculations were made over the white color analysis values in the power parameter calculations. As the  $L^*$  color is more effective in color designs in other studies, a color design study was carried out based on the  $L^*$  value (Jurek and Wagnerova, 2021). For the motif on the box surface, it is aimed to obtain three-stage colors that will create contrast to each other. The white value  $(L^*)$  on the black and white axis of the untreated natural state of the beech massif was measured as 76.21. The change of this value in the black direction was obtained as 19.35 at the maximum 200 mm/s engraving speed and 70 % engraving power parameter, as seen in Figure 2.a. It was determined between the maximum (76.21) and minimum (19.35) values according to the mean values in the three targeted white color test findings. These values were determined as 20, 40 and 60 according to the homogeneous distribution. In order to complete the engraving power process with maximum efficiency and in the shortest time, the engraving speed was kept constant at the highest value of 500 mm/s. The speed-power parameter to be applied to the box surface was calculated from the Scratch color regression formula obtained from the above findings.

 $L_1^*=46.993+(Engraving power*-0.607)+(Engraving speed*0.050)$ 

20 =46.993+(*Engraving power*\*-0.607)+(*Max speed*\*0.050)



**Figure 3** (a) CNC laser product manufacturing design, (b) CAM manufacturing initial stage **Slika 3.** (a) Dizajn proizvodnje CNC laserskog proizvoda, (b) početna faza CAM proizvodnje

*Engraving power*\*-0.607= 46.993+(500\*0.050)-20 (Projected white color)

Engraving power= 46.993+5/0.607=85.65

 $L_2^*=46.993+(Engraving power*-0.607)+(Engraving speed*0.050)$ 

40 =46.993+(*Engraving power\*-*0.607)+(*Max speed\**0.050)

*Engraving power*\*-0.607= 46.993+(500\*0.050)-40 (Projected white color)

Engraving power= 46.993-15/0.607=52.70

 $L_{3}^{*}=46.993+(Engraving power*-0.607)+(Engraving speed*0.050)$ 

60 =46.993+(*Engraving power\*-*0.607)+(*Max speed\**0.050)

*Engraving power*\*-0.607= 46.993+(500\*0.050)-60 (Projected white color)

#### *Engraving power*= 46.993-35/0.607=19.75

The required engraving power ratios were determined as 85 %, 53 % and 20 % in order to perform the prescribed laser engraving processes with a whiteness value of 20.40 and 60 at 500 mm/s engraving speed. In Figure 3a, the design is ready for production in the CAM program and in Figure 3b, the initial stage of surface processing is shown. The CAM design, developed with the obtained parameters, was completed and the production phase was started. The completed surface treatment is shown in Figure 4a, the local view of the motif processed at 20 %, 55 % and 85 % speed rates is shown in Figure 4b, and the perspective of the finished wooden box in Figure 4c.

# 4 CONCLUSIONS

#### 4. ZAKLJUČAK

In this study, the applicability of CNC laser assisted regression modeling method in furniture surface treatments in CIE  $L^*a^*b^*$  color system was investigated. The highest values of the effect of laser wood surface treatment parameters on  $L^*$ ,  $a^*$  and  $b^*$  color changes were obtained in the  $L^*$  black and white axis. While the laser engraving power increased the black color level, the engraving speed decreased the black color. It has been determined that the regression modeling method can be successfully applied over  $L^*$  values in furniture surface decoration with CNC laser, provided that material, machine and processing parameters are taken into consideration. The same research can be done for color types with a higher effect level in studies to be carried out on other tree species. However, in this study, both the effect level of the  $a^*$  and  $b^*$ 



**Figure 4** (a) CNC laser engraved surface, (b) Local view of the engraving motif, (c) Assembled wooden box **Slika 4.** (a) CNC laserski gravirana površina, (b) lokalni prikaz motiva za graviranje, (c) sastavljena drvena kutija

color groups was low and the regression relationships were low. When a good design is made, the color tones from natural wood color to black on the engraving surfaces add value to the decoration, depending on the processing power of the laser.

If the motif design, color design and production parameter process in laser wood decoration are not well managed, poor images may be encountered in color contrasts on the decoration surface. While it is desired to obtain darker colors, layer differences on the surfaces may be a disadvantage due to the engraving depth. As a result, this study has shown that the regression modeling method of the CNC laser engraving technique can be applied in furniture top surface color design in mass production in accordance with the industrial engineering approach, although it has some limitations.

## **5 REFERENCES**

#### 5. LITERATURA

- Aytepe, B., 2013: Evaluation of the curriculum design and effectiveness of the application of laser etching in graphics and ceramics. PhD Thesis, Gazi University Institute of Educational Sciences Department of Applied Arts Education Graphic Education, Department Ankara.
- Çifçi, S. K.; Demirarslan, Z., 2021: An overview of furniture design styles in the 20<sup>th</sup> century. Electronic Journal of Social Sciences, 20 (79): 1607-1627. https://doi. org/10.17755/esosder.842189
- Gurau, L.; Petru, A.; Varodi, A.; Timar, M. C., 2017. The influence of CO<sub>2</sub> laser beam power output and scanning speed on surface roughness and colour changes of beech (*Fagus sylvatica*). BioResources, 12 (4): 7395-7412. https://doi.org/10.15376/biores.12.4.7395-7412
- Jurek, M.; Wagnerova, R., 2021: Laser beam calibration for wood surface colour treatment. European Journal of Wood and Wood Products, 79: 1097-1107. https://doi. org/10.1007/s00107-021-01704-3
- Karabıyık, K., 2016: Effects of current digital technologies on furniture design and production. MSc Thesis, Beykent University Graduate School of Natural and Applied Sciences, Department of Interior Architecture, pp. 87.

- Katik, F.; Kubovsky, I., 2011: Chemical changes of beech wood due to CO<sub>2</sub> laser irradiation. Journal of Photochemistry and Photobiology A: Chemistry, 222: 105-110. https://doi.org/10.1016/J.JPHOTOCHEM.2011.05.008
- Kubovsky, I.; Babiak, M., 2009: Color changes induced by CO<sub>2</sub> laser irradiation of wood surface. Wood Research, 54 (3): 61-66.
- Kubovsky, I.; Kacik, F., 2014: Colour and chemical changes of the lime wood surface due to CO<sub>2</sub> laser thermal modification. Applied Surface Science, 321: 261-267. https://doi.org/10.1016/j.apsusc.2014.09.124
- Kudela, J.; Kubovsky, I.; Andrejko, M., 2020: Surface properties of beech wood after CO<sub>2</sub> laser engraving. Coating, 10 (1): 77. https://doi.org/10.3390/coatings10010077
- Li, R.; Xu, W.; Wang, X.; Wang, C., 2018: Modeling and predicting of the color changes of wood surface during CO<sub>2</sub> laser modification. Journal of Cleaner Production, 183: 818-823. https://doi.org/10.1016/j.jclepro.2018.02.194
- Mitsui, K.; Tolvaj, L.; Papp, G.; Bohus, J.; Szatmari, S.; Berkesi, O., 2005: Changes in the properties of light-irradiated wood with heat treatment – Application of laser. Wood Research, 50: 1-8.
- Petru, A.; Lunguleasa, A., 2017: Colour infuence of the laser raster speed on wood pyrography. Pro L1gno, 13 (4): 508-515.
- Petutschnigg, A.; Stöckler, M.; Steinwendner, F.; Schnepps, J.; Gütler, H.; Blinzer, J.; Holzer, H.; Schnabel, T., 2013: Laser treatment of wood surfaces for ski cores: An experimental parameter study. Advances in Materials Science and Engineering, 2013: 123085. http://dx.doi.org/10.1155/2013/123085
- Reinprecht, L.; Vidholdová, Z., 2021: The impact of a CO<sub>2</sub> laser on the adhesion and mold resistance of a synthetic polymer layer on a wood surface. Forests, 12 (2): 242. https://doi.org/10.3390/f12020242
- Teivonen, A., 2016: Laser Surgery System. Lahti University of Applied Sciences Degree Program in Materials Engineering, pp. 23-35.
- Vidholdova, Z.; Reinprecht, L.; Igaz, R., 2017: The impact of laser surface modification of beech wood on its color and occurrence of molds. BioResources, 12 (2): 4177-4186. https://doi.org/10.15376/biores.12.2.4177-4186
- \*\*\*ASTM D 2244:2007 Standard Practice for Calculation or Color Tolerances and Color Differences from Instrumentally Measured Color Coordinates, ASTM Standards, A. B. D.

#### **Corresponding address:**

#### Dr. CEBRAİL AÇIK

Ministry of National Education Onikisubat District Directorate of National Education, 46050, Kahramanmaraş, TURKEY, e-mail: cebrail46@hotmail.com
Tatjana Kočetov Mišulić<sup>1</sup>, Aleksandra Radujković<sup>1</sup>, Zdravko Popović<sup>2</sup>, Ksenija Hiel<sup>3</sup>

# Timber Strength Grading as Necessary Basis for Structural Design in Ex-YU Region: Part 1

Ocjenjivanje drva prema čvrstoći kao nužna osnova za projektiranje konstrukcija na području bivše Jugoslavije: dio 1.

# **ORIGINAL SCIENTIFIC PAPER**

**Izvorni znanstveni rad** Received – prispjelo: 19. 4. 2023. Accepted – prihvaćeno: 20. 9. 2023. UDK: 674.032; 630\*85 https://doi.org/10.5552/drvind.2023.0105 © 2023 by the author(s). Licensee University of Zagreb Faculty of Forestry and Wood Technology. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license.

**ABSTRACT** • Classification of timber for various commercial purposes is essential for its proper application in order to ensure the reliability and economic use. Visual grading of structural timber is commonly used in a number of EU countries, with different grading national standards optimized for locally available wood. Countries in the ex-YU region are traditional partners in wood trade and had the same standards for visual grading, but in most of the regions these standards are not completely compliant with EN requirements. Consequently, that leads to the fact that the most of regionally available structural timber is not assigned into strength classes, which is the starting point for the limit-state concept in design of timber structures. The aim of this paper is to emphasize the lack of strength classification of structural timber in the ex-YU region, which is a prerequisite for the design of timber structures made by civil engineers. Based on an overview of visual classification types with regional experience in grading, relevant EN standards, and differences in design concepts with possible consequences of grading approach, it can be concluded that "quality" grades and "strength" classes are not easily comparable.

**KEYWORDS:** structural coniferous timber; visual grading; strength classes; safety factors; European standards

**SAŽETAK** • Klasifikacija drvne građe za različite komercijalne namjene ključna je za njezinu pravilnu upotrebu kako bi se zajamčila sigurna i ekonomična uporaba. Vizualno ocjenjivanje konstrukcijskog drva obično se provodi u mnogim zemljama EU-a uz pomoć različitih nacionalnih standarda ocjenjivanja optimiziranih za lokalno dostupno drvo. Zemlje bivše Jugoslavije tradicionalni su partneri u međusobnoj trgovini drvom i imale su zajedničke standarde za vizualno ocjenjivanje drvne građe, ali u većini njih ti standardi nisu u potpunosti usklađeni s EN zahtjevima. To posljedično rezultira činjenicom da većina regionalno dostupnoga konstrukcijskog drva nije

<sup>&</sup>lt;sup>1</sup> Authors are assistant professors at University of Novi Sad, Faculty of Technical Sciences, Department of Civil Engineering and Geodesy, Novi Sad, Serbia. https:// orcid.org/0000-0003-3002-5885; https://orcid.org/0000-0003-4863-4486

<sup>&</sup>lt;sup>2</sup> Author is full professor at University of Belgrade, Faculty of Forestry, Department of Wood Science and Technology Belgrade, Serbia. https://orcid.org/0000-0003-1557-4982

<sup>&</sup>lt;sup>3</sup> Author is assistant professor at University of Novi Sad, Faculty of Agriculture, Department of Fruit Growing, Viticulture, Horticulture and Landscape Architecture, Novi Sad, Serbia. https://orcid.org/0009-0008-9793-7292

razvrstana u klase čvrstoće, što je polazište za koncept graničnog stanja u projektiranju drvnih konstrukcija. Cilj rada jest upozoriti na nepostojanje klasifikacije čvrstoće konstrukcijskog drva na području bivše Jugoslavije, a to je preduvjet za projektiranje drvnih konstrukcija, što je posao građevinskih inženjera. Na temelju tipova vizualne klasifikacije i regionalnih iskustava u ocjenjivanju, relevantnih EN normi te razlika u konceptima projektiranja s mogućim posljedicama pristupa ocjenjivanju, zaključeno je da ocjene razreda kvalitete i klase čvrstoće u promatranim zemljama jednostavno nisu međusobno usporedive.

KLJUČNE RIJEČI: konstrukcijsko crnogorično drvo; vizualno ocjenjivanje; klase čvrstoće; faktori sigurnosti; europske norme

# **1 INTRODUCTION**

# 1. UVOD

Wood as a building material has been used for centuries, but during the past few decades, it has been recognized as one of the favourite materials in the construction sector. Renewable sources, availability in the immediate surroundings, relatively low energy consumption to process in addition to continuous construction technology innovation have all contributed to the wider affirmation of wood-based products in modern architecture, providing extraordinary possibilities in the shaping and design of structures, Figure 1.

The built environment generates 47 % of annual global CO, emissions: building operations are responsible for 27 %, while building materials and construction (typically referred to as embodied carbon) are responsible for an additional 20 % (Agopian, 2022). Growing awareness leads to wood becoming the trending material of the 21st century, in the form of solid and engineered wood products (EWPs) for structural and indoor/outdoor use. Modern architectural trends in prefabricated residential buildings and large-scale timber construction lead to eco-friendly urbanization, where timber grows in popularity compared to steel, concrete and other mineral-based building materials. According to Hetemaki et al., 2017, the indirect share estimation of timber construction was 15 % in turnover and 19 % in employment of the total EU building construction sector. On the basis of recent decade production of wood fibre insulation boards, glulam, cross-laminated

timber and laminated veneer products, the estimation of growth rates is in a range from 2.5 % to 15 % (Hildebrandt et al., 2017), which could create an added value by a lower environmental impact throughout the entire life cycle. Previous low economic competitiveness of multi-storey timber buildings has been overcome by technological progress in EWPs production, so the share of timber construction in total building market increases in the EU and world. The share of "wooden houses" depends on the region, forest resources and on tradition of building with timber, as well as on technological opportunities (North and Western Europe N/ WE) and/or political and economic opportunities (South Europe SE). It was found that "social and political barriers most typically limit the development of wood construction" (Leszczyszyn et al., 2022), which means that it is necessary to provide the basic knowledge and social conditions for increasing the share of wood in countries where the use of timber is still under the average EU level.

South-Eastern Europe (SEE region) has a long tradition in building with timber, which was first reflected in the craftsmanship and later in industrial approach to construction. Nowadays, structural engineers and architects in the ex-YU region are faced with two challenges: the imperative to increase the use of timber in the construction sector and the harmonization / adjustment of regulations with the recently introduced safety concepts in design. The transition from the concept of global safety factors, which refers to common ex-YU design codes (JUS), to the concept of partial safety factors, which re-



Figure 1 a) Pancho Aréna Hungary (2014), b) Mjøstårnet tower Norway (2019), c) Mistissini Bridge Canada (2014) Slika 1. a) stadion Pancho Aréna, Mađarska (2014.), b) neboder Mjøstårnet, Norveška (2019.), c) most Mistisini, Kanada (2014.)

fers to the set of Eurocodes (ENs) in design of timber structures, requires more precise classification of structural timber and products, where cooperation between wood processing industry and civil engineers is necessary, as well as between countries that traditionally trade in timber in the region.

SEE region is considered as the smallest forest area in total ( $30446 \cdot 10^3$  ha) and in forest share (23.5%) compared to the rest of Europe, e.g. the highest forest area share is 53.2 % in NE (Alexandrov and Iliev, 2019). Data on the forestry potentials in the ex-YU countries (Bosnia and Herzegovina, Montenegro, North Macedonia, Serbia as well as Croatia and Slovenia as EU members), as part of the SEE region, are given in Table 1, where the data on coniferous sawn wood as the basic raw material for construction is particularly highlighted. Data is given according to FAO database (FAO, 2022).

Table 2 presents the data about wood product trade i.e. coniferous sawn wood export/import in six ex-YU countries, together with leading trade partners. The data is given according FAOSTAT and INDEXBOX platforms (FAOSTAT, 2021; INDEXBOX, 2022). Strong trade connections between the analyzed countries are evident, besides traditional partner countries (Austria, Germany, Italy) and some new ones (Albania, Turkey, Bulgaria, Slovakia and Czech Republic).

Based on natural recourses, Bosnia and Herzegovina and Slovenia are recognized as the main suppliers of sawn coniferous wood in the ex-YU region, which dates back to the past. The existing trade and interconnection of the countries based on tradition and common regulations for the visual classification of sawn wood in the ex-YU area indicates the necessity to harmonize the quality assessment of structural coniferous timber in the sense of EU requirements and construction industry needs.

This paper is limited to the SEE ex-YU countries because of previous common legislation in construction and forestry domain standards and a number of existing timber structures from earlier period (XX century). Besides, sawn-wood mutual export/import is still

Table 1 Forest are	ea, growing stock and	sawn wood proc	duction in ex-YU	countries (2020/21)
Tablica 1. Površi	na šuma, drvne zalihe	i piljeno drvo u	zemljama bivše J	ugoslavije (2020./21.)

1 5		5	<b>e</b>	<i>,</i>		
Country	Forest area		Growing stock		Sawn wood	
Zemlja	Površina	a pod šumom	Drvna zaliha		Piljeno drvo	
	Total,	Of land	Total,	Coniferous,	Total,	Coniferous
E- VII	10 <sup>3</sup> ha	area, %	10 <sup>6</sup> m <sup>3</sup>	10 <sup>6</sup> m <sup>3</sup> / %	10 <sup>3</sup> m <sup>3</sup>	10 <sup>3</sup> m <sup>3</sup>
EX-YU	Ukupno,	Zemljišne	Ukupno,	Četinjače,	Ukupno,	Četinjače,
	10 <sup>3</sup> ha	površine, %	$10^{6}  \text{m}^{3}$	$10^{6}{ m m^{3}}/\%$	$10^{3}  m^{3}$	$10^{3}  m^{3}$
Bosnia & Herzegovina / Bosna i Hercegovina	2188	42.7	404.7	n/a	1650	743
Croatia / Hrvatska	1939	34.7	427.2	50.6 / 11.8	1298	257
Montenegro / Crna Gora	827	61.5	121.4	42.8 / 35.3	109	108
North Macedonia	1001	20.7	76 4	m/a	7	2
Sjeverna Makedonija	1001	39.7	/0.4	11/a	/	2
Serbia / Srbija	2723	31.1	420.9	27.0 / 6.4	452	99
Slovenia / Slovenija	1238	61.5	414.1	172.0 / 41.5	1029	904

Table 2 Wood product trade in ex-YU: coniferous sawn wood (10<sup>3</sup> m<sup>3</sup>) (2021)

Tablica 2. Promet proizvoda od drva na području bivše Jugoslavije: piljena građa četinjača (10<sup>3</sup> m<sup>3</sup>) (2021.)

Country	Consumption	Production	Export	Import	Leading partners Vodeći partneri		
Zemlja	Potrošnja	Proizvodnja	Izvoz	Uvoz	Export Izvoz	Import Uvoz	
Bosnia & Herzegovina Bosna i Hercegovina	32.3	743.0	717.9	7.2	Serbia, Croatia, Albania, Italy, Austria	Austria, Croatia, Serbia	
Croatia Hrvatska	343.3	256.5	265.0	351.8	Slovenia, Italy, Austria	Austria, B&H, Slovenia	
Montenegro Crna Gora	17.4	107.8	93.6	3.2	n/a	B&H	
North Macedonia Sjeverna Makedonija	45.1	2.0	0.5	43.6	Turkey, Italy, Serbia	B&H, Montenegro, Bulgaria	
Serbia <i>Srbija</i>	379.0	99.0	15.0	295.0	Italy, B&H, Germany	B&H, Montene- gro, Austria	
Slovenia Slovenija	693.1	904.0	851.6	640.7	Italy, Austria, Croatia	Austria, Czech Rep., B&H, Slovakia	



very frequent in the region although two countries are the members of the EU. In order to understand the importance of the transition problem from quality (stress) grades to the system of strength classes of structural timber, and to obtain the relevant conclusions, an overview of visual classification types with regional experience in grading, differences in design concepts, relevant standards, as well as possible consequences, is given in the next chapter. The final goal of the present paper (Part 1) is to emphasize the necessity to assign visual grades and species to strength classes of structural timber in the region. In Part 2 of the paper, the integral procedure for the conversion of the II grade coniferous timber into strength classes is presented and its application is illustrated through the analysis of archive data sample obtained from regional sources.

## 2 VISUAL GRADING AND TIMBER STRUCTURAL DESIGN

- 2. VIZUALNO OCJENJIVANJE I PROJEKTIRANJE DRVNIH KONSTRUKCIJA
- 2.1 Importance and types of timber visual grading
- 2.1. Značenje i načini vizualnog ocjenjivanja drva

The quality assessment of wood as a material is crucial for engineering design, and therefore for the structure reliability. Without knowledge of the theory and practice of classification, there may be misunderstandings that threaten the structural safety, while the grading system is expected to be as uniform and unified as possible within the European market. For the correct use of timber in constructions, it is necessary to carry out adequate classification, because today civil engineers face two problems: assessing the quality of timber built in existing buildings due to the need for reconstruction, as well as assessing the quality of locally available wood for use in newly designed buildings.

Wood as a material could be classified at different stages of harvesting and/or production using different procedures, which are initially based on visual inspection and appropriate measurement. Thus, after a rough dimensional classification (EN 1315) of timber, when the final use is still unknown, the roundwood classification is made according to quality (classes A, B, C, D) in accordance with EN standard (e.g. EN 1927 or EN 1316) depending on the type/species of wood and the presence, size and distribution of factors that affect quality (EN 1309). These EN standards are officially in force in ex-YU countries, although not-binding, because they are intended for the log trade where different conditions can be agreed in trade contracts (e.g. traditional quality according to previous JUS). Generally, from the forestry aspect, the main difference in the classification of roundwood lies in the fact that EN standards classify according to quality without prejudicing its future purpose, while JUS classified according to its final use. According to ex-YU regulations, in addition to being classified into dimensional grades, logs and round timber had to be classified into quality classes for intended use, where, for structural purposes, the classification of sawmill logs into three categories (I, II, III) is of importance for designers.

Sawn timber could be used for non-structural and structural purposes, so different grading procedures could be applied due to the final use of the material. Generally, the grading process is always based on visual sorting, but opposite to grading of non-structural elements that is exclusively based on (surface) appearance, grading of structural elements also considers determination of relevant strength & stiffness properties, i.e. visual grading is followed by strength classification and testing. Appearance grading is a process of assessing the prescribed number and size of parameters by visual sorting and it is not designed to take into account the final use of timber (e.g.: linings, joinery, packaging or construction). For example, acceptable timber for a structural engineer in terms of bearing strength could be completely unacceptable for an architect in appearance (Swedish wood, 2023).

In the ex-YU region, the structural grading meant assessing the timber grades for construction purposes by adequate appearance requirements (particularly related to size and position of knots and their perceived effects on bending strength and stiffness) as useful indication, performed by experienced operators. Such assessment was often followed by additional testing of load-bearing properties on relevant samples by destructive, non-destructive or semi-destructive methods (Nowak *et al.*, 2021). Machine strength grading is not present in the region, while evaluation by non (semi) destructive methods are not widely used, except for research purposes or for *in-situ* evaluation of existing buildings with high importance (Stepinac *et al.*, 2017).

In order to preserve traditional national grading systems, and to provide the necessary unification level as well, EU introduced the standard EN 14081-1 for structural timber of rectangular cross-section with prescribed general requirements. The standard gives general limitations in 3 aspects of strength reducing characteristics that have to be taken into account during visual grading process as basic principles: Limitations for strength and stiffness reducing characteristics (knots, slope to the grain, density and rate of growth, fissures); Limitations for geometrical characteristics (wane, warp); Limitations for biological characteristics (insect and fungal damage).

The existing regional standards for visual grading of timber (ex JUS U.D0.001/1983) are based on

DIN 4074:1958. The changes introduced in DIN during the harmonization with EN 14081-1 were incorporated in national legislation of Slovenia and Croatia (2009), while in the rest of the region the grading standard remained unchanged. Due to new demands in timber industry (use of smaller/slender timber pieces in construction and laminated products), the crucial novelties introduced in DIN 4074-1:2012 are the four kinds of cross-section with different specifications and possibilities of flatwise and edgewise orientation of planks and boards. That imposes that different orientation of the same slender timber element could lead to different "quality" grade because of different criteria due to final position of timber element in the structure. Although this looks very similar compared to previous versions of the standard, the amendment about possibility of edgewise orientation of boards and planks may have significant implications for the timber classification and its use in construction, particularly when designed according to EC5 (EN 1995-1-1, 2004).

# 2.2 Design concepts of timber structures: global vs. partial safety factors

# 2.2. Koncepti projektiranja drvnih konstrukcija: globalni nasuprot parcijalnim činiteljima sigurnosti

The safe use of timber in construction is a question of design and quality of chosen timber (products). All engineering design methods can be reduced to the basic concept of safety, according to which the design resistance of structural elements should be greater or equal to the effects of design loads (actions), with adequate safety factors. The way of providing a safety factor determines the concept of design: deterministic ("working" - allowable stress design (ASD) with global safety factor) or reliability-based (semi-probabilistic) design concept (limit states design (LSD), with partial safety factors). In the basic form of ASD, the factor that provides safety is applied <u>only</u> to the resistance of structural material, while loading is generally taken as nominal, with some modifications due to loadduration effects and load combinations. LSD takes into account the uncertainty of inaccurate models and unfavourable deviations associated with strength properties, but also with uncertainties in the assessment of the effects of actions.

ASD has served for a long time to provide a simple approach to design procedure, with uncertainty in reliability of structures designed with such a procedure. The analysis performed in the ex-YU region showed that: "the previous (JUS) regulations have large deviations in reliability depending on the location of the structure. The reliability of constructions in areas with higher snow load is insufficient..." (Čizmar *et al.*, 2018). That reflects disadvantages in aspects of selection of different factors of safety by "intuition" regarding the variety of materials and by using the same factor of safety for all types of load. Global safety factors for different stress states in the ASD concept are in the range of n = 2 - 4, where permissible stress is based on extensive research of properties by testing small clear wood samples. The absence of strength-reducing characteristics and favourable orientation on small clear specimens provide an indication of the upper limit of the expected performance of realistic timber pieces (Crews and Ritter, 1996), so "the critical design criterion is not how strong a piece of timber is, but rather how weak it could be". ASD codes have been focused on providing adequate strength and achieving a proposed level of safety, while LSD ones, besides strength limit state, also recognize other limit states, such as serviceability, stability, fire and fatigue. In LSD, the partial safety factors for strength limit state and timber materials are in the range of  $\gamma_m = 1.25 - 1.30$  that indicate lower safety factor and require higher reliability in assessment of relevant timber properties.

The conclusion of rough comparison of ASD and LSD is that global and partial safety factors are different in "composition" and in values. LSD, as advanced and more reliable concept, requires different experimental procedures and statistical distribution models of testing data, followed by improved visual classification of timber.

# 3 STRENGTH CLASS SYSTEM IN EUROPEAN STRUCTURAL CODES

# 3. SUSTAV KLASA CVRSTOCE U EUROPSKIM KODOVIMA ZA PROJEKTIRANJE

# 3.1 Visual ("quality" or "stress") grades vs. strength class system

# 3.1. Vizualne ocjene ("kvaliteta" ili "naprezanje") u odnosu prema sustavu klase čvrstoće

In a new era of timber structures design, the first challenge for the engineers is handling the classification of timber as a structural material in order to obtain the input data for calculations. Namely, according to ASD concept, the structural coniferous timber in the region was visually graded into quality "stress" grades (I, II, III) in accordance with DIN 4074 standard (S13, S10, S7). In contrast to that, LSD concept introduced in Eurocode 5, considers the so called "strength class system" (EN 338) of softwood (12 classes - coniferous and poplar), where structural timber is graded based on a set of rules given in EN 14081-1 and supporting ENs. It is obvious and important to emphasise that quality grades are wider terms than strength classes (SC), Figure 2, as well as that "strength classes" are not based only on strength but also on stiffness and density of timber from a particular source and region. "Strength



Figure 2 a) Grades vs. strength classes, b) Visual grades according to national standards, c) Strength classes EN338 - class determining properties (Ridley-Ellis *et al.*, 2016)

**Slika 2.** a) Ocjene u odnosu prema klasi čvrstoće, b) vizualne ocjene prema nacionalnim standardima, c) klase čvrstoće EN338 – svojstva koja određuju klasu (Ridley-Ellis *et al.*, 2016.)

class system" groups together grades and species with similar strength properties making them interchangeable. Additional species/grades can be incorporated into the system at any time, and engineers could "specify a chosen strength class in their calculations without awareness of costs or availability of alternative species or grades" (EN 338). Due to the number of classes and grading precision, "SC system" is primarily optimized for machine grading (EN 14081:1-5) and large sawmills with huge industrial capacity.

Despite machine grading, visual grading is still in common use in a number of European countries (including ex-YU region), with different national grading standards optimized for locally available wood. Due to diversity of locally spread wood, it is impossible to lay down a single standard for all EU Member States, so the prescribed general requirements for structural timber (within EN 14081-1) have to be met by national standards. It is necessary to respect EN 14081 recommendations in order to avoid large inconsistency in classification criteria and further application of timber (Prka *et al.*, 2001; Ištvanić *et al.*, 2008). Due to evident overlapping of SCs, Figure 2c, and economic reasons, it is useful to adopt a small number of SCs in design practice (Bather, 2021).

#### 3.2 Relevant EN standards for testing and quality control of structural timber

# 3.2. Relevantni EN standardi za ispitivanje i kontrolu kvalitete konstrukcijskog drva

In order to provide an adequate and consistent basis for the introduction of "SC system" in design of timber structures and on the EU market, a set of standards for testing and statistical procedures are established, accompanied with a standard that allows the assignment of visual grades and species of locally spread wood into strength classes. Harmonized normative standards that follow general requirements in EN 14081-1 are:

- EN 338: Structural timber Strength classes (provides characteristic values of strength, stiffness and density for softwood and hardwood, where the dominant classification is made by designation due to characteristic value of bending strength).
- EN 408: Timber structures Determination of some physical and mechanical properties (specifies test methods for determining the structural properties).
- EN 384: Structural timber Determination of characteristic values of mechanical properties and density (provides a procedure to derive characteristic values that are comparable in terms of population. The standard permits the use of as much existing test data as possible from various sampling and testing techniques).
- EN 14358: Timber structures Calculation and verification of characteristic values (gives statistical methods for the determination of characteristic values from test results on a sample drawn from a clearly defined reference population. In case of solid timber, it combines with specific adjustment factors given in EN 384).
- EN 1912: Structural timber Strength classes Assignment of visual grades and species (national document that lists visual strength grades, species and sources of timber specifying the strength classes to

which they are assigned according to tradition and adequate quality control).

Evaluation of structural timber of the ex-YU region from visually established 3 quality grades (ASD) into numerous strength classes (LSD) is a demanding but necessary task for structural designers in the region that implies knowledge about standardized testing methods and statistical procedures for converting the archive data and proper analysis of newly obtained data. In ASD, the leading parameter of classification was the mean value of bending strength, tested by 3-point test on small clear specimens, while stiffness parameter (bending modulus of elasticity II to the grains) together with density were given as general mean values that slightly differ for solid and glulam timber. In LSD concept, a detailed procedure of quality assessment is based on 3 reference material properties (5 % characteristic bending strength, mean stiffness and 5% characteristic density), where the minimum value is relevant for strength class estimation (EN 338). The reliability basis for strength class system is given in JCSS Probabilistic model code: preferred theoretical distributions and desirable coefficients of variations (CoV) of referent properties, while the established relationships with other relevant material properties are given in Table 2 (EN 384). Test methods (EN 408) for "key" strength and stiffness properties in bending (overall span 18 times the specimen depth, including full size specimens) are based on 4-point test. Modulus of elasticity (MoE), considered as "the best single indicator of timber quality" (Bostrom, 1999), could be measured as local and global, which gives the opportunity to use the archive data from previous tests where the *MoE* is measured as global. Determination of wood density is provided on small defect free specimens (EN 408), which is consistent with the previous JUS standards.

Once properties are determined through EN 408, standards EN 384 and EN 14358 give the necessary statistical basis and methods for determining the characteristic values of mechanical properties and density for defined populations of visual grades and/or strength classes of machine graded structural timber. These standards provide the possibility of assessment by "calculation" of archive data results obtained from previously conducted test under different load arrangements from defect-free specimens or from products of structural size.

Finally, by applying to CEN/TS 124 with a documented report, EN 1912 will publish the list of assignments of local wood to strength classes according to EN 14081-1 and EN 338. This list is not exhaustive, but in every moment gives good guidelines for the quality of trade market and input data for architectural and civil engineers' design projects. With an insight into prEN 1912:2022, it can be concluded that only Slovenia, of all the ex-YU countries, has implemented this procedure and has classified its national timber resources into strength classes guided by national SIST standards (mixed spruce/fir of grade S10 is assigned to C24, of grade S7 to C18, while only fir of grade S7 is assigned to C16).The official assignment is very useful for structural engineers because some provisional explanations could be found in the regional documents (e.g. JUS grade I is sometimes assigned to series of SCs from C30 to C50, II grade as C24-C27, while III as C22). That kind of "assignment" overestimates the structural timber from regional recourses and could mislead the designers.

Regardless of the fact that the classification of timber into the strength class is not a prerequisite for trade, it is a condition for the CE marking and for placing timber on the EU market (Negro *et al.*, 2013). In the ex-YU region, apart from large glulam factories (e.g. "Voćin", Croatia) that closely cooperated with EU companies, strength classified timber is not offered in the regional trade. That requires a mutual effort in which structural engineers and engineers from forest-based industries should participate in order to ensure proven quality for the design and additional trade value.

# 4 CONCLUSIONS 4. ZAKLJUČAK

The consideration of EN standards, JUS visual grading rules and practice, limit state design approach with sensitivity on consistent grading, leads to the following conclusions:

The substantial improvement of the regional visual grading rules must take into account the future position of the element in the structure i.e. visual assessment has to be conducted for edgewise and/or flatwise orientations of the boards. In addition, in a lack of machine grading, the regional visual grading of structural timber must be consistent and stricter in application of EN requirements because of smaller safety material factors in structural design compared with the previous global ones.

The proclaimed SC system with classes from C14 to C50 (although the highest recognised timber class in EU is C35 with limited application) is established by statistical tools with overlaps in relevant parameters and it is optimized for machine grading. In visual grading it is practical and effective to have only a few SC with consistent grading rules defined by final purpose in structure (e.g. Sweden established classes with structural description of use: C14, C18, C24 for normal structural use and C30, C35 for extra load-bearing purposes but not for elements of large dimensions).

It is important to notice that SC classification of coniferous solid timber boards directly influences the

production and classification of glued laminated products (glulam and cross-laminated timber), the prefabricated products that are the essence of modern building with timber.

Although the strength classification of timber is not necessary for regional trade, for structural engineers and designers it is of high importance to have a framework of available construction timber in the ex-YU region (Part 2). Otherwise, the overestimation or underestimation by random selection of SC could lead to inadequate or low quality projects, as well as to high prices and conflicts with investors and suppliers.

The joint efforts between civil and wood processing engineers are necessary in order to classify regional structural timber according EN requirements. One of the first tasks is to assign visual grades and regional species of structural timber to strength classes according to EN 1912.

### Acknowledgements – Zahvala

This research has been supported by the Ministry of Science, Technological Development and Innovation through project No. 451-03-47/2023-01/200156 "Innovative scientific and artistic research from the FTS domain".

# **5 REFERENCES**

# 5. LITERATURA

- Agopian, N., 2022: Countering climate change: Building decarbonization + Building resiliency = Improved human condition. RenewAire White Paper, 6 p.
- Alexandrov, A. H.; Iliev, I., 2019: Forests in South-Eastern Europe. Poplar, 203: 79-85.
- 3. Bather, M., 2021: Technical note on the use of visual grading codes for the appraisal of individual in situ structural timber elements. University of Liverpool, 43 p.
- Bostrom, L., 1999: Determination of modulus of elasticity in bending of structural timber – comparison of two methods. HolzalsRoh und Werkstoff, 57: 145-149. https://doi.org/10.1007/s001070050030
- Crews, K.; Ritter, M., 1996: Development of Limit State Design Procedures for Timber Bridges. In: National conference on wood transportation structures, Madison, USA, pp. 292-300.
- Čizmar, D.; Serdarević, Š.; Volarić, I.; Gelo, D., 2018: Reliability assessment of timber purlins. Polytechnic & Design, 6 (2): 74-84. https://doi.org/10.19279/TVZ. PD.2018-6-2-02
- Hetemäki, L.; Hanewinkel, M.; Muys, B.; Ollikainen, M.; Palahí, M.; Trasobares, A., 2017: Leading the way to a European circular bioeconomy strategy. From Science to Policy 5. European Forest Institute. https://doi. org/10.36333/fs05
- Hildebrandt, J.; Hagemann, N.; Thrän, D., 2017: The contribution of wood-based construction materials for leveraging a low carbon building sector in Europe. Sustainable Cities and Society, 34: 405-418. https://doi. org/10.1016/j.scs.2017.06.013
- Ištvanić, J.; Antonović, A.; Pervan, S.; Jambreković, V.; Benković, Z.; Kavran, M., 2008: Sawmilling in Croatia.

Part 2. Croatian Sawmilling in New Millennium (in Croatian). Drvna industrija, 59 (4): 169-180.

- Leszczyszyn, E.; Heräjärvi, H.; Verkasalo, E.; Garcia-Jaca, J.; Araya-Letelier, G.; Lanvin, J.-D.; Bidzińska, G.; Augustyniak-Wysocka, D.; Kies, U.; Calvillo, A.; Wahlströmh, M.; Kouyoumji, J.-L., 2022: The future of wood construction: opportunities and barriers based on surveys in Europe and Chile. Sustainability, 14 (7): 4358: 1-29. https://doi.org/10.3390/su14074358
- Negro, F.; Cremonini, C.; Zanuttini, R., 2013: CE Marking of structural timber: the European standardization framework and its effects on Italian manufacturers. Drvna industrija, 64 (1): 55-62. https://doi.org/10.5552/ drind.2013.1214
- Nowak, T.; Patalas, F.; Karolak, A., 2021: Estimating mechanical properties of wood in existing structures – Selected aspects. Materials, 14 (8): 1941:1-26. https:// doi.org/10.3390/ma14081941
- 13. Prka, T.; Ištvanić, J.; Mekić, S., 2001: Application of standards in unedged oak board classification (in Croatian). Drvna industrija, 52 (1): 7-22.
- Ridley-Ellis, D.; Stapel, P.; Bano, V., 2016: Strength grading of sawn timber in Europe: an explanation for engineers and researchers. European Journal of Wood and Wood Products, 74: 291-306. https://doi.org/10.1007/ s00107-016-1034-1
- Stepinac, M.; Rajčić, V.; Barbalić, J., 2017: Inspection and condition assessment of existing timber structures. Građevinar, 69 (9): 861-873. https://doi.org/10.14256/ JCE.1994.2017
- 16. \*\*\*DIN 4074-1:2012 Strength grading of wood. Part 1: Coniferous sawn timber. German National Standard.
- 17. \*\*\*Drvene konstrukcije: https://www.drvene-konstrukcije.hr/ (Accessed Jan 27, 2023).
- \*\*\*EN 338:2016 Structural timber Strength classes. European Committee for Standardization, Brussels.
- \*\*\*EN 384:2016 Structural timber Determination of characteristic values of mechanical properties and density. European Committee for Standardization, Brussels.
- \*\*\*EN 408:2012 Timber structures Structural timber and glued laminated timber – Determination of some physical and mechanical properties. European Committee for Standardization, Brussels.
- \*\*\*EN 1309-1:2000 Round and sawn timber Method of measurement of dimensions. Part 1: Sawn timber. European Committee for Standardization, Brussels.
- 22. \*\*\*EN 1315:2010 Dimensional classification of round timber. European Committee for Standardization, Brussels.
- \*\*\*EN 1316-1/2:2012 Hardwood round timber Qualitative classification. Part 1: Oak & beech / Part 2: Poplar. European Committee for Standardization, Brussels.
- 24. \*\*\*EN 1927-1:2008 Qualitative classification of softwood round timber. Part 1: Spruces and firs. European Committee for Standardization, Brussels.
- \*\*\*EN 14081-1:2016 Timber structures Strength graded structural timber with rectangular cross section. Part 1: General requirements. European Committee for Standardization, Brussels.
- 26. \*\*\*EN 14358:2016 Timber structures Calculation and verification of characteristic values. European Committee for Standardization, Brussels.
- 27. \*\*\*EN 1995-1-1:2004 Eurocode 5: Design of timber structures. Part 1-1: General – Common rules and rules for buildings. European Committee for Standardization, Brussels.

- \*\*\*FAO, 2022: Global Forest Resources Assessments. Country Reports. https://www.fao.org/forest-resourcesassessement/fra-2020/country-reports/en/ (Accessed Sep 2, 2022).
- 29. \*\*\*FAOSTAT 2021: Forestry Production and Trade. https://www.fao.org/faostat/en/#data/FO (Accessed Sep 5, 2022).
- \*\*\*Imre Makovecz & Tamas Dobrosi: https://www10. aeccafe.com/blogs/arch-showcase/2014/08/29/panchoarena-in-felcsut-hungary-by-dobrosi-tamas/ (Accessed Sep 21, 2022).
- 31. \*\*\*INDEXBOX: https://app.indexbox.io/ (Accessed Sep 10, 2022).
- 32. \*\*\*JCSS, 2006: Probabilistic model code. Part 3.5: Timber. Joint Committee on Structural Safety, 16 p.

- 33. \*\*\*prEN 1912:2022 Structural timber Strength classes
   Assignment of visual grades and species. European Committee for Standardization, Brussels.
- 34. \*\*\*Stantec company: https://www.nordic.ca/en/projects/ structures/mistissini-bridge (Accessed Sep 21, 2022).
- \*\*\*Swedish wood: https://www.swedishwood.com/ wood-facts/about-wood/wood-grades (Accessed Jan 27, 2023).
- \*\*\*Voll Arkitekter: https://www.inexhibit.com/casestudies/clt-goes-tall-high-rise-buildings-in-cross-laminated-timber/ (Accessed Sep 21, 2022).
- \*\*\*Zbirka jugoslovenskih pravilnika i standarda za gradjevinske konstrukcije, 1995. Knjiga 4: Drvene konstrukcije. Beograd, 687 p.

**Corresponding address:** 

# Assist. Prof. ALEKSANDRA RADUJKOVIĆ

University of Novi Sad, Faculty of Technical Sciences, Department of Civil Engineering and Geodesy, Dr Sime Miloševića 12, 2100 Novi Sad, SERBIA, e-mail: leksa@uns.ac.rs

# Laboratorij za ispitivanje namještaja i opreme dječjih igrališta



laboratorij za ispitivanje namještaja i opreme dječjih igrališta osposobljen je prema normi HRN EN ISO/IEC 17025

više od 40 akreditiranih metoda u području ispitivanja namještaja, premaza i dijelova za namještaj, dječjih igrališta i opreme

usluge izvan područja akreditacije:

istraživanje konstrukcija i ergonomije namještaja

ispitivanje materijala i postupaka površinske obrade

ispitivanje zapaljivosti madraca i ojastučenog namještaja

vještačenja namještaja i stručna mišljenja



Laboratorij je član udruge Hrvatski laboratoriji CROLAB – čiji je cilj razvoj hrvatskih laboratorija kao infrastrukture razvoja proizvodnje i gospodarstva u okviru zahtjevnog otvorenog tržišta, korištenjem zajedničkih potencijala i sinergijskih učinaka udruživanja, dok je...

...Fakultet šumarstva i drvne tehnologije punopravni član udruženja INNOVAWOOD – kojemu je cilj doprinijeti poslovnim uspjesima u šumarstvu, drvnoj industriji i industriji namještaja s naglaskom na povećanje konkurentnosti europske industrije.

Istraživanje kreveta i spavanja, istraživanja dječjih kreveta, optimalne konstrukcije stolova, stolica i korpusnog namještaja, zdravog i udobnog sjedenja u školi, uredu i kod kuće neka su od brojnih istraživanja provedena u *Zavodu za namještaj i drvo u* graditeljstvu, kojima je obogaćena riznica znanja o kvaliteti namještaja.



Dobra suradnja s proizvođačima, uvoznicima i distributerima namještaja čini nas prepoznatljivima

# Znanje je naš kapital





Sveučilište u Zagrebu • Fakultet šumarstva i drvne tehnologije Laboratorij za ispitivanje namještaja i opreme dječjih igrališta Zavod za namještaj i drvo u graditeljstvu Svetošimunska cesta 23 10000 Zagreb, Hrvatska Sevinç Kaz<sup>1</sup>, Saim Ateş<sup>2</sup>, Tuba Külçe<sup>3</sup>

# Comparison of Reaction Wood and Normal Wood of Some Commercial Tree Species

# Usporedba reakcijskoga i normalnog drva nekih komercijalnih vrsta drva

# **ORIGINAL SCIENTIFIC PAPER**

Izvorni znanstveni rad Received – prispjelo: 6. 6. 2023. Accepted – prihvaćeno: 24. 10. 2023. UDK: 674.038.15 https://doi.org/10.5552/drvind.2023.0123 © 2023 by the author(s). Licensee University of Zagreb Faculty of Forestry and Wood Technology. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license.

**ABSTRACT** • This study aims to analyze the reaction wood samples for some industrial tree species naturally grown in Kastamonu province in Turkey and compare them with the relevant standards. Some anatomical, chemical, fiber morphological, optical properties, and color changes before and after drying were analyzed for the reaction wood (RW) samples. While the holocellulose content of fir and pine compression wood (CW) was found to be lesser (~3-4 %), the lignin content was higher than those of the opposite wood (OW) (~34 % for pine and 12 % for fir). On the contrary, the amount of holocellulose was found to be higher ( $\sim 1-4$  %), and the lignin was lower (at about 6-15 %) in the tension wood (TW) samples. It was observed that average lengths are more extended in TW (~50-54 %) and shorter in CW (~13-17 %) than those of OW. Significant differences were observed between the anatomical structures of the coniferous and deciduous species studied. Although, the greatest color differences in wet and oven-dried samples of coniferous trees were measured in CW (~15-17%), it has been found as about 0.7-3 % in TW for deciduous species. Some differences were observed in the anatomical, optical, fiber morphological, and chemical properties of the RW for the studied wood species. Due to its higher lignin content and better physical properties, CW can be used for producing small households and hand tools, ornaments, toys, etc. It will also be appropriate for use in milling and turning work. It is recommended that, because of the lower lignin content and higher polysaccharide ratio, TW should be primarily used for the cellulose, pulp, and paper industries, where high mechanical resistance values are required. Consequently, RW formation causes some physical, chemical, mechanical, anatomical, and optical differences compared to OW in deciduous and coniferous species.

**KEYWORDS:** *pine; fir; oak; beech; anatomical and chemical structure* 

**SAŽETAK** • Cilj ovog istraživanja bio je analizirati uzorke reakcijskog drva nekih industrijskih vrsta drva koje prirodno rastu u pokrajini Kastamonu, u Turskoj, i usporediti ih s relevantnim standardima. Na uzorcima reakcijskog drva (RW) analizirana su njihova anatomska i kemijska svojstva, morfološka svojstva vlakana te optička svojstva i promjena boje prije i nakon sušenja. Sadržaj holoceluloze u kompresijskom drvu (CW) jelovine i borovine bio je manji nego u opozitnom drvu (OW) ( $\sim 3 - 4$  %), dok je sadržaj lignina u kompresijskom drvu bio veći nego u opozitnome ( $\sim 34$  % u borovini i 12 % u jelovini). Naprotiv, utvrđeno je da je u uzorcima tenzijskog drva (TW) količina holoceluloze veća ( $\sim 1 - 4$  %), a lignina manja ( $\sim 6 - 15$  %). Uočeno je također da su prosječne duljine

<sup>&</sup>lt;sup>1</sup> Author is graduate student at Kastamonu University, Kastamonu, Turkey.

<sup>&</sup>lt;sup>2</sup> Author is researcher at Kastamonu University, Faculty of Forestry, Department of Forest Industrial Engineering, Kastamonu, Turkey.

<sup>&</sup>lt;sup>3</sup> Author is researcher at Kastamonu University, Forestry and Nature Tourism Specialized Coordinatorship, Kastamonu, Turkey.

vlakanaca u tenzijskom drvu veće (~50 – 54 %), a u kompresijskom drvu manje (~13 – 17 %) nego u opozitnome. Nadalje, uočene su značajne razlike u anatomskoj strukturi između proučavanih crnogoričnih i bjelogoričnih vrsta drva. Iako su u crnogoričnih vrsta drva razlike u boji između mokrih i suhih uzoraka bile najveće u kompresijskom drvu (~15 – 17 %), u bjelogoričnih su vrsta drva razlike u boji bile veće u tenzijskom drvu, i to za oko 0,7 – 3 %. U proučavanih vrsta drva uočene su neke razlike u anatomskim, optičkim, morfološkim i kemijskim svojstvima reakcijskog drva. Zbog većeg sadržaja lignina i boljih fizičkih svojstava kompresijsko se drvo može upotrebljavati za proizvodnju malih kućanskih i ručnih alata, ukrasa, igračaka itd., a prikladno je i za glodanje odnosno tokarenje. Preporučuje se da se tenzijsko drvo zbog nižeg sadržaja lignina i većeg omjera polisaharida primarno rabi u industriji celuloze i papira, gdje je potrebna velika mehanička otpornost. Stvaranje reakcijskog drva posljedično uzrokuje neke fizičke, kemijske, mehaničke, anatomske i optičke razlike u usporedbi s opozitnim drvom bjelogoričnih i crnogoričnih vrsta.

KLJUČNE RIJEČI: borovina; jelovina; hrastovina; bukovina; anatomska i kemijska struktura

# **1 INTRODUCTION**

# 1. UVOD

The wood material is important globally because it is both natural and sustainable. Turkey has a total forest area of 22,740,297 hectares, while the productive forest area is 13,830,510 hectares. Oak, pine, beech, juniper, and fir are the first five most important commercial tree species, with the highest distribution in Turkey (Kuzka, 2013; OGM, 2021).

Wood materials are used for various purposes, such as fiber-firewood in the paper industry, woodbased material production, etc. Although Turkey's total annual wood consumption is 32 million m<sup>3</sup>, the annual wood production is 26.3 million m<sup>3</sup>. The difference between production and consumption is supplied from private plantation forests (5 million m<sup>3</sup>) and covered by imports (1.5-2 million m<sup>3</sup>) (OGM, 2021).

Reaction wood, formed under the effect of forces that bend wood due to external factors (wind, etc.), affects the end-product properties of the wood (Rowell, 2005). These defects are CW (compression wood) in coniferous trees and TW (tension wood) in hardwoods. In CW, force is exerted on the lower side of the bent stem to make the trunk upright, while in TW, pressure is applied on the upper side of the bent stem. These abnormal formations in wood sourced from RW (reaction wood) are seen as defects in living trees and sawn timber (Donaldson and Singh, 2016). Almost all forests are located on sloping lands in Turkey. Therefore, the reaction wood formation rate is high in industrial woods (OGM, 2022).

There are many studies on reaction wood in the literature. Köksal and Kılıç Pekgözlü (2016) investigated the microscopic structures of *Pinus sylvestris L.*, *Pinus nigra Arnold*, and *Pinus brutia Ten*. compression woods in their study. Ruelle *et al.* (2010) examined the physical and mechanical properties of CW and OW (opposite wood) samples from three different tropical tree species. It has been determined that the compressive strength of TW is generally lower than that of OW. Geffertova *et al.* (2019), in their study with beech TW samples, determined that the samples fiber length and width were higher than those of OW.

In this study, RW samples taken from fir (*Abies nordmanniana* (Stev.) Spach. subsp. *bornmuelleriana*), pine (*Pinus nigra Arn.* subsp. *pallassiana* (Lamb.) Holmboe), oak (*Quercus robur* L.), and beech (*Fagus orientalis* Lipsky) were studied in terms of anatomical, chemical, fiber morphological, and optical properties. Besides, some evaluations and comparisons between RW and OW, coniferous and deciduous woods, and inter-species properties have been reported.

# 2 MATERIALS AND METHODS

2. MATERIJALI I METODE

## 2.1 Materijal

2.1. Materijal

# 2.1.1 Study area

### 2.1.1. Istraživani lokalitet

The wood samples were taken from Bostan Forestry Management Chiefdom, located within the borders of Kastamonu/Turkey. The area of the management directorate is 8,287.7 hectares, the forest area is 6,378.4 hectares, and the open area is 1,910.3 hectares. It is located between 41°02'58"- 41°09'57" north latitude and 33°40'33"- 33°51'50" east longitude. The highest point is Büyük Hacet Hill, with a height of 2,587 m, and the lowest point is 1,155 m, where Karasu Stream and Açlık Stream meet. Samples containing RW were selected from log storage areas of the Kastamonu Forest Regional Directorate.

# 2.1.2 Sample collection and preparation2.1.2. Prikupljanje i priprema uzoraka

For each tree species, three wood disks were taken from 1.30 m of breast height with a 10 cm thickness. The RW and OW zones of the disk samples were determined according to Figure 1. All RW and OW zones of the wood disks were cut carefully and chipped to the size of a matchstick (0.5-1.0 mm). Air-dried and grounded in a Willey-type mill, wood samples were



**Figure 1** Preparation of RW (reaction wood) and OW (opposite wood) samples for analyses (mm) **Slika 1.** Priprema uzoraka RW-a (reakcijskog drva) i OW-a (opozitnog drva) za analizu (dimenzije su u milimetrima)

screened through 40 to 60 mesh sieves. The coarse and thick parts were eliminated by staying above 40 mesh, and the thin pieces were eliminated by going below 60 mesh. The remaining material over 60 mesh was used for experiments. The powdered wood was stored in a dry and clean glass jar with its mouth closed.

# 2.2 Methods

#### 2.2. Metode

To determine the chemical analyses, lignin (TAP-PI T 222 om-88), holocellulose (Wise's Chlorite method), alpha-cellulose (TAPPI T 203 os-71), ethanol solubility, 1 % NaOH solubility, hot water and cold water solubility (TAPPI T 207 om-88) experiments were done according to the above applicable standards.

For microscopic examinations,  $1.5 \text{ cm} \times 1.5 \text{ cm} \times$ 

from the area near the next annual ring boundary of latewood zones. For the morphological properties of fibers, the chlorite method was used for maceration (Wise and Karl, 1952), and measurements were taken using Digimiser<sup>®</sup> for at least 150 fiber samples.

For the determination of the optical properties of the samples, the samples were optically measured after both fresh cutting and oven-dried conditions at  $(37\pm3)$  °C for four days. The color coordinates (*L*\*, *a*\*, *b*\*) were measured according to the D65 standard with the Konica Minolta CM2500d device (Geffertova *et al.*, 2019). The percentage rates can be found by dividing the value by the total value and then multiplying the result by 100. The formula used to calculate the percentage is (value/total value) × 100 %.

### 3 RESULTS AND DISCUSSION 3. REZULTATI I RASPRAVA

# 3.1 Chemical analysis

# 3.1. Kemijska analiza

The holocellulose content in the pine CW was 3.87 % less than in OW and 4.97 % less in fir CW than in OW. Studies conducted on pine show that the holocellulose content is around 60-70 %, and in some studies, its amount increases up to 75 % (Ateş, Kırcı and Tutuş, 2008). In studies on fir species, 70-80 % holocellulose content was determined (Topaloğlu and Erisir, 2018). The details are given in Figure 2.

It has been determined that there may be differences in both the amount of CW and holocellulose content of the coniferous wood samples according to harvested regions (Arslan *et al.*, 2021; As *et al.*, 2001; Ataç and Eroğlu, 2013; Gülsoy and Öztürk, 2015; Popescu *et al.*, 2011).

Lignin content was 12.6 % and 34.8 % higher in coniferous trees compared to OW. Wood properties vary from species to species and within the same species depending on many factors (age, genetic charac-





Slika 2. Kemijski sastav reakcijskoga i opozitnog drva različitih vrsta (TW – tenzijsko drvo, CW – kompresijsko drvo, OW – opozitno drvo)



Figure 3 Solubility of reaction and opposite wood of tree species (TW – tensile wood, CW – compression wood, OW – opposite wood)

Slika 3. Topljivost reakcijskoga i opozitnog drva različitih vrsta (TW – tenzijsko drvo, CW – kompresijsko drvo, OW – opozitno drvo)

teristics, environmental characteristics, etc.) (Doğu *et al.*, 2001). The reason why lignin contents are different between pine and fir species is thought to be due to genetic factors. It was observed that the lignin content of TW in the deciduous trees was 15.74 % and 6.68 % in the beech and oak OW, respectively.

Figure 3 shows that ethanol solubility in RW is  $\sim$ 18.25 % greater than that of OW for coniferous wood. The difference is assumed to be caused by changes in the quantity of CW in the tree species (Kilic et al., 2010). While the ethanol solubility of fir species is commonly considered to be around 3 %, pine species have a solubility of 3-4 % (Rowell, 2005). Similar to previous investigations, ethanol solubility was shown to be 11.39 % greater in pine CW compared to OW (Kiliç et al., 2010). It demonstrates that fir OW has a lower extractive content than pine samples of about 1.37 %. The extractive content of wood samples for the same species is also affected by variables such as altitude, climate, and so on (Akyıldız and Ateş, 2008). Also, it was determined that the amount of ash in CW of coniferous species was higher (13.15 % in pine and 26 % in fir) than in OW. Compared to the literature findings, alcohol benzene solubility is rising up to ~5.9 % in oak wood (Miranda et al., 2017), and this rate is coming to around 3 % in beech (Malakani et al., 2014). As a result, ethanol solubility was determined to be 5.38 % in coniferous trees (fir, pine) and 3 % in deciduous trees (oak, beech) on average.

The amount of  $\alpha$ -cellulose in the samples was 18.48 % and 37.08 % less than OW and CW of pine and fir, respectively. Similarly,  $\alpha$ -cellulose content in pine wood was found to be between 35 % and 50 % (Arslan *et al.*, 2021; Kılıç *et al.*, 2010). It has been determined that the cellulose content of CW is reduced by up to 30 % compared to normal wood (Bozkurt and Erdin, 2011). The  $\alpha$ -cellulose content in fir CW was found to be 58.94 % and 75.82 % lower than beech and oak TW, respectively. It has been observed that the amount of  $\alpha$ -cellulose in deciduous trees is generally

around 40-50 % (Fišerová *et al.*, 2013; Rowell, 2005). The average amount of  $\alpha$ -cellulose was found to be 29.82 % lower in CW than in TW.

The hot water solubility of pine CW was found to be 3.69 % and 3.0 % for fir CW. The same solubility was found at 3.16 % and 1.56 % for pine and fir OW, respectively. Compared to deciduous species, hot water solubility is 37.08 % lower in coniferous species. Similar studies were found in the literature (Kılıç *et al.*, 2010; Hafizoğlu and Usta, 2005). Cold water solubility was found to be 32.11 % higher in TW of deciduous tree species compared to CW of coniferous species. Also, cold water solubility was found to be 24.67 % higher for deciduous trees than for coniferous species. It was observed that the cold water solubility of TW was 4.54 % higher than that of CW. Tutuş *et al.* (2010) determined the cold water solubility as 3.42 % in their research on pine wood.

1 % NaOH solubility of fir CW and pine CW was found to be 13.15 % and 15.73 %, respectively; it was determined as 9.37 % for fir OW and 12.40 % for pine. 1 % NaOH solubility of RW of deciduous species was found to be greater at about 78.27 % than RW of coniferous species, and TW was also higher at a rate of 61.14 % compared to CW. Similar studies were found in the literature (Benouadah *et al.*, 2018).

# 3.2 Anatomical properties

# 3.2. Anatomska svojstva

Earlywood vessel and tracheid diameters on transverse sections were greater than latewood cross sections in all studied species. In Figure 4, the tangential diameter of the earlywood tracheid in OW of coniferous species was found to be larger (~1.4  $\mu$ m) than that of CW, while the latewood tracheid was found to be larger (~2.94  $\mu$ m) in CW.

The tangential diameter of the latewood vessel cells in beech TW was found to be 12.44 % and was narrower than that of OW. It has been stated in the literature that the beech has a circular vessel structure,



Figure 4 Average tangential diameters of vessel or tracheid (TW – tensile wood, CW – compression wood, OW – opposite wood)

Slika 4. Prosječni tangentni promjeri traheja ili traheida (TW – tenzijsko drvo, CW – kompresijsko drvo, OW – opozitno drvo)

the latewood vessels are smaller in diameter and few in number, and the rays are in the form of longitudinal lines in the transverse section (Safdari *et al.*, 2008).

While the tangential direction of tracheid diameters was found to be 1.69 % wider in the average latewood dimeters of pine CW in cross-section, it was found to be 54.24 % wider in the earlywood than that of OW (Figure 4). The earlywood and latewood widths of fir and pine CW were found to be wider (Fir: ~63.5 % earlywood, 55 % latewood. Pine: 28.3 % earlywood, 34.8 % latewood) than those of OW. The earlywood widths of oak (~60.8 %) and beech (~15.8 %) TW were found to be less than those of OW, and the latewood (~2.7 % beech, ~22.2 % oak) widths were higher than those of TW. Earlywood tracheids had a larger lumen and a thin wall structure (Esteban and Palacios, 2009). For pine wood, tracheid diameters were averagely narrower in CW (about 0.22 %) than in OW, and they were also 8.59 % wider in latewood. CW transition from earlywood to latewood is gradually compared to OW.

The average number of rays per mm<sup>2</sup> of pine CW was found to be 4.58, and the average number of rays per mm<sup>2</sup> of OW was found to be 4.00. In Figure 5, it was observed that the average number of rays per mm<sup>2</sup> of pine CW in the tangential section was 14.50 % larger than that of OW. The average number of rays 1 mm<sup>2</sup> in

fir CW was 6.24 and 5.74 in OW. This rate was found to be 8.70 % for fir in the same direction. On average, about two times more ray numbers were found on 1mm<sup>2</sup> in fir wood. The average number of rays in 1 mm<sup>2</sup> was 9.60 and 7.40 in the beech TW and OW, respectively.

The average number of rays in 1 mm<sup>2</sup> of oak TW was 10.20, and the average number of rays in 1 mm<sup>2</sup> of oak OW was 9.63. For oak and beech TW, 5.91 % and 29.72 % more ray numbers were measured, respectively, in 1mm<sup>2</sup> than for OW. The average ray number was determined to be 16.0 % higher in oak wood than in beech. The average number of rays per 1mm<sup>2</sup> in deciduous species was 78.98 % higher than in coniferous species. Similarly, the average number of rays in 1mm<sup>2</sup> of TW was 82.99 % higher than that of CW. Compared to pine and fir woods, the average number of rays in 1mm<sup>2</sup> was higher in beech and oak. Similar results were obtained for *Quercus aucheri* Jaub. and Spach (11.17), *Quercus cocifera* L. (16.23), and *Quercus ilex* L. (7.03) (Kadem and Fakir, 2017).

# **3.3 Annual ring widths 3.3.** Širina godova

Figure 6 shows that annual ring widths are almost 100 % higher in pine OW than fir OW. The earlywood zone for pine CW was 1.20 % greater than that of fir



Figure 5 Number of ray cells in 1 mm<sup>2</sup> of tangential section of wood species (TW – tensile wood, CW – compression wood, OW – opposite wood)

**Slika 5.** Broj stanica trakova na 1 mm<sup>2</sup> tangentnog presjeka različitih vrsta drva (TW – tenzijsko drvo, CW – kompresijsko drvo, OW – opozitno drvo)



Figure 6 Earlywood and latewood widths of studied wood species (TW – tensile wood, CW – compression wood, OW – opposite wood)

Slika 6. Širine ranoga i kasnog drva ispitivanih vrsta (TW – tenzijsko drvo, CW – kompresijsko drvo, OW – opozitno drvo)

CW. The latewood portion of pine OW was wider (66.67 %) than that of fir OW. Also, the late wood of pine CW was 15 % wider than that of fir TW.

The total annual ring width found for beech OW was 21.77 % wider than that of oak OW, although the annual ring widths of pine OW were on average 93.43 % greater than those of fir OW. However, the annual rings for the beech TW are on average 46.36 % broader than those of oak TW. This ratio for pine TW was 4.15 % higher than for fir TW. Bektaş et al., (2016) have found an annual ring width of 2.11mm in oak. The average width of oak annual rings was between 1.09 and 2.94 mm (Matisons and Brümelis, 2012). Another study stated that climate significantly impacts annual ring width (Pourtahmasi et al., 2011; Roibu et al., 2020; Yaman et al., 2020). In the literature, it has been reported that the anatomical characteristics of oak are affected by environmental conditions (Bozkurt and Erdin, 1995; Gricar et al., 2013).

## **3.4 Morphological properties of fibers** 3.4. Morfološka svojstva vlakana

Morphological measurements of the studied wood fibers are presented in Figures 7 and 8. Although the fiber length of pine OW was 10.61 % longer than that of fir OW, pine CW had 14.04 % longer fiber length than fir CW. Also, the average fiber length of oak OW is 1.79 % longer than that of beech OW, and this rate for oak TW is 9.35 % higher than for beech TW. Figure 8 shows that both fiber and lumen widths are decreasing in CW (~19.75 %) and increasing for TW samples (~7.13 %). However, the average fiber width of beech OW was 24.83 % greater than that of oak OW.

It was determined that the tracheid width of pine OW was 18.11 % wider than that of fir OW. Oak TW fibers were 16.71 % smaller than those of beech TW. Tracheid widths of fir CW were on average 27.79 % narrower than those of pine CW. It was also determined that the lumen width of fir OW cells was 43.15 % less than that of pine OW, and oak OW had 58.53 % less fiber cell lumen than beech OW. While this rate for oak TW was 50.83 % smaller than that of beech TW, fir CW cell lumen widths were 41.95 % less than those of pine CW.

Comparing the cell wall thickness, beech OW was 8.48 % thicker than oak. This rate was 2.95 % higher for pine OW than for fir OW. Also, 0.15 % thicker beech TW was observed than oak TW. This rate for pine CW was 15.70 % higher than for fir CW. Although the difference is not significant, all RW samples have a thicker cell wall than the OW zone. The growth location characteristics and the growth environment af-



**Figure 7** Fiber length of the studied wood samples (TW – tensile wood, CW – compression wood, OW – opposite wood) **Slika 7.** Duljina vlakana ispitivanih uzoraka drva (TW – tenzijsko drvo, CW – kompresijsko drvo, OW – opozitno drvo)



Figure 8 Fiber width, lumen width and cell wall thicknesses (TW – tensile wood, CW – compression wood, OW – opposite wood)

fect the characteristics of the same type of tree (Abuamoud *et al.*, 2018).

#### 3.5 Optical properties

# 3.5. Optička svojstva

Wood samples were oven-dried. As shown in Figure 9, the color changes ( $\Delta E^*$ ) of green wood and dried wood were 36.86 % higher in coniferous species compared to deciduous species. The color change in CW was higher (42.23 %) than that of TW. The total color change in fir CW was 15.09 % higher than in fir OW, and 17.01 % higher in pine CW than that of OW. It was determined that the total color change of oak and beech TW was higher than that of OW, 2.66 % and 0.72 %, respectively. As a result, it was assumed that the color differences between the different wood species may be due to the different chemical content, especially the extractive contents.

The average color change in fir and pine species was 8.87 % higher than in oak and beech species. The red color parameter ( $a^*$ ) was 1.92 % lower in fir and pine CW than in OW; it was observed that the TW of

oak and beech was 0.09 % higher than that of OW. While the yellow color parameter ( $b^*$ ) was 1.42 % lower in fir and pine CW than in OW, it was also 1.04 % lower in oak and beech TW than in OW. The brightness parameter ( $L^*$ ) was 2.65 % higher in fir and pine CW than in OW; it was observed that the TW of oak and beech was 0.35 % lower than OW.

In similar studies, it has been stated that the color of the wood is generally dark, and that the temperature is the main factor. Also, RW caused an increase in the color change in the wood. Due to the high content of extractive substances and lignin in coniferous species, the color difference in wood is found to be high (Aydemir and Gündüz, 2009; Tarmian *et al.*, 2011; Yazıcı and Özlüsoylu, 2020).

# 4 CONCLUSIONS

# 4. ZAKLJUČAK

This study investigated the anatomical and chemical structures, fiber, and optical properties of the RW and



**Figure 9** Optical properties of reaction and opposite woods of fir, pine, beech and oak species ( $\Delta E^*$  – total color change value,  $L^*$  – brightness value,  $a^*$  – red color value,  $b^*$  – yellow color value, TW – tensile wood, CW – compression wood, OW – opposite wood)

Slika 9. Optička svojstva reakcijskoga i opozitnog drva jelovine, borovine, bukovine i hrastovine ( $\Delta E^*$  – ukupna promjena boje,  $L^*$  – svjetlina,  $a^*$  – crvenozeleni ton,  $b^*$  – žutoplavi ton, TW – tenzijsko drvo, CW – kompresijsko drvo, OW – opozitno drvo)

Slika 8. Širina vlakana, širina lumena i debljina stanične stijenke (TW – tenzijsko drvo, CW – kompresijsko drvo, OW – opozitno drvo)

OW parts of the beech, oak, pine, and fir wood species naturally grown in a specific environment over time.

The amount of ash was found to be higher in deciduous species compared to other wood species. Besides, it was determined to be higher in TW and CW of coniferous species than in OW.

Although the amount of  $\alpha$ -cellulose was higher in TW and lower in CW part than in OW, the lignin amount was higher in CW and lower in TW.

Compared to OW, although fiber and lumen widths are decreasing in CW and increasing for TW samples, the color change ratio is higher for CW than for TW.

TW contains more cellulose and less lignin than normal wood in the production of paper and cellulose. However, the strength of the fibers was lower in samples containing high TW. Since CW and TW are more sensitive than normal wood, they cause problems in applications such as structural elements, carriers, surface treatments, cutting, and sawing.

This study determines the anatomical, chemical, fiber, and optical properties of the four wood species commonly used in industry and gives their comparison with each other. The data obtained will likely provide important benefits in practice. In addition, in scientific terms, such comparison studies fill an important gap in the literature in reaching clear results.

During the growing period of trees, necessary environmental, sylvicultural, and genetic interventions should be taken to prevent the formation of reaction wood and to provide more efficient use of wood in the forest product industry.

### Acknowledgements - Zahvala

This paper was an extension of a paper presented orally at the 2nd International Multidisciplinary Congress of Natural Sciences and Engineering held in Ankara, Turkey, on December 1-2, 2022.

# **5 REFERENCES**

#### 5. LITERATURA

- Abuamoud, M. M.; Ateş, S.; Durmaz, E., 2018: Comparison of some anatomical, chemical and fibrous characteristics of Turkish Red Pine (*Pinus brutia* Ten.) sampled from different regions. Kastamonu University Journal of Forestry Faculty, 18 (1): 75-82. https://doi. org/10.17475/kastorman.364592
- Akyıldız, M. H.; Ateş, S., 2008: Effect of heat treatment on equilibrium moisture content (EMC) of some wood species in Turkey. Research Journal of Agriculture and Biological Sciences, 4 (6): 660-665.
- Arslan, B. M.; Avcı, E.; Tozluoğlu, A.; Birtürk, T.; Saraçbaşı, A.; Örtel, E., 2021: Comparison of some wood properties of juvenile black pines of different origin planted in the same habitat. Maderas Ciencia y Tecnología, 66 (23): 1-12. http://dx.doi.org/10.4067/s0718-221x2021000100466

- As, N.; Koç, H.; Doğu, D.; Atik, C.; Aksu, B.; Erdinler, S., 2001: Anatomical, physical, mechanical and chemical properties of trees of industrial importance growing in Turkey. Journal of Istanbul University Faculty of Forestry, 55 (1): 72-87 (in Turkish).
- Ataç, Y.; Eroğlu, H., 2013: The effects of heartwood and sapwood on kraft pulp properties of *Pinus nigra* J. F. Arnold and *Abies bornmülleriana* Mattf. Turkish Journal of Agriculture and Forestry, 37 (2): 243-248. https://doi. org/10.3906/tar-1205-20
- Ateş, S.; Kırcı, H.; Tutuş, A., 2008: Influence of cooking variables of crimean pine (*Pinus nigra* Arnold sbsp. *pallassiana*) kraft pulp on the resulting paper sheets. Wood Research, 53 (1): 81-90.
- Aydemir, D.; Gündüz, G., 2009: The effect of heat treatment on the physical, chemical, mechanical and biological properties of wood. Journal of Bartin Faculty of Forestry, 11(15): 71-81 (in Turkish).
- Bektaş, İ.; Oruç, S.; Bal, B. C., 2016: Relationship between wood physical properties of english oak (*Quercus robur* L.) and soil variables. Journal of Forestry, 12 (1): 61-71.
- Benouadah, N.; Aliouche, D.; Pranovic, A.; Willför, S., 2018: Chemical characterization of *Pinus halepensis* sapwood and heartwood. Wood Material Science & Engineering, 14 (3): 157-164. https://doi.org/10.1080/1748 0272.2018.1448436
- Bozkurt, A. Y.; Erdin, N., 1995: Definition Characteristics of Coniferous and Deciduous Wood (Wood Anatomy II), Publication No: 3907, Institute of Science and Technology. Publication No: 6, ISBN 975-404-406-6, Istanbul (in Turkish).
- Bozkurt, Y.; Erdin, N., 2011: Odunun kimyasal ve ultramikroskopik yapısı: 2. Baskı, Ağaç teknolojisi.
- Doğu, D.; Koç, H.; As, N.; Atik, C.; Aksu, B.; Erdinler, S., 2001: Basic identity information of trees of industrial importance growing in Turkey and general evaluation of their use. Journal of Istanbul University Faculty of Forestry, 51 (82): 69-84 (in Turkish).
- Donaldson, A. L.; Singh, P. A., 2016: Reaction wood. In: Secondary Xylem Biology, Academic Press, Elsevier, pp. 93-110. https://doi.org/10.1016/B978-0-12-802185-9.00006-1
- Esteban, G. L.; Palacios, D. P., 2009: Comparative wood anatomy in Abietoideae (*Pinaceae*). Botanical Journal of the Linnean Society, 160 (2): 184-196. https://doi. org/10.1111/j.1095-8339.2009.00971.x
- Fišerová, M.; Opálená, E.; Illa, A., 2013: Comparative study of hemicelluloses extraction from beech and oak wood. Wood Research, 58 (4): 543-554.
- Geffertova, J.; Geffert, A.; Vilkovska, T.; Klement, I.; Vacek, O., 2019: Selected aspects of the reaction beech wood. Acta Facultatis Xylologiae Zvolen, 61 (2): 5-15.
- Gricar, J.; Luis, M.; Hafner, P.; Levanic, T., 2013: Anatomical characteristics and hydrologic signals in treerings of oaks (*Quercus robur* L.). Trees, 27: 1669-1680. https://doi.org/10.1007/s00468-013-0914-9
- Gülsoy, K. S.; Öztürk, F., 2015: Kraft pulping properties of European black pine cone. Maderas. Ciencia y Tecnología, 17 (4): 875-882. http://dx.doi.org/10.4067/ S0718-221X2015005000076.
- Hafizoğlu, H.; Usta, M., 2005: Chemical composition of coniferous wood species occurring Turkey. Holz als Rohund Werkstoff, 63: 83-85. https://doi.org/10.1007/ s00107-004-0539-1
- 20. Kadem, E.; Fakir, H., 2017: Anatomical characteristics of evergreen oak (*Quercus* L.) species naturally distributed

in Turkey. Turkish Journal of Forestry, 18 (4): 275-281 (in Turkish).

- Kılıç, A.; Sarıusta, E. S.; Hafizoğlu, H., 2010: Chemical structure of Scots pine, larch and red pine pressure wood. Journal of Bartın Faculty of Forestry, 12 (18): 33-39 (in Turkish).
- 22. Köksal, S. E.; Kılıç Pekgözlü, A., 2016: Microscopic structure of Scots pine (*Pinus sylvetris* L), Black pine (*Pinus nigra* Arnold.) and Red pine (*Pinus brutia* Ten.) pressure wood. Duzce University Faculty of Forestry Journal of Forestry, 12 (1): 72-82 (in Turkish).
- KUZKA, 2013: TR82 Level 2 Region Kastamonu Çankırı Sinop Provinces Regional Plan 2014-2023 (online). https://www.kuzka.gov.tr/Planlama.asp?P=8& Planlama=tr82-bolge-planlari (Accessed Nov. 20, 2022) (in Turkish).
- Malakani, M.; Khademieslam, H.; Hosseinihas, S. K.; Zeinaly, F., 2014: Influence of fungal decay on chemimechanical properties of beech wood (Fagus Orientalis). Cellulose Chemistry and Technology, 48 (1-2): 97-103.
- Matisons, R.; Brümelis, G., 2012: Influence of climate on tree-ring and earlywood vessel formation in Quercus robur in Latvia. Trees, 26: 1251-1266. https://doi. org/10.1007/s00468-012-0701-z
- 26. Miranda, I.; Sousa, V.; Ferreira, J.; Pereira, H., 2017: Chemical characterization and extractives composition of heartwood and sapwood from *Quercus faginea*. Plos-One, 12 (6): e0179268. https://doi.org/10.1371/journal. pone.0179268 J
- Popescu, C. M.; Popescus, M. C.; Lisa, G.; Sakata, Y., 2011: Evaluation of morphological and chemical aspects of different wood species by spectroscopy and thermal metods. Journal of Molecular Structure, 988 (1-3): 65-72. https://doi.org/10.1016/j.molstruc.2010.12.004
- Pourtahmasi, K.; Lotfiomran, N.; Bräuning, A.; Parsapajouh, D., 2011: Tree-Ring width and vessel characteristics of oriental beech (Fagus Orientalis) along an altitudinal gradient in the caspian forests, northern iran. IAWA Journal, 32 (4): 461-473. https://doi.org/10.1163/ 22941932-90000071
- Roibu, C. C.; Sfecla, V.; Mursa, A.; Ionita, M.; Nagavciuc, V.; Chiriloaei, F.; Leşan, I.; Popa, I., 2020: The climatic response of tree ring width components of ash (*Fraxinus excelsior* L.) and common oak (*Quercus robur* L.) from eastern Europe. Forests, 11 (5): 600. https://doi.org/10.3390/f11050600

- 30. Rowell, M. R., 2005: Handbook of wood chemistry and wood composites. New York, London.
- Ruelle, J.; Jacques, B.; Hiroyuki, Y.; Bernard, T., 2010: Variations in physical and mechanical properties between tension and opposite wood from three tropical rainforest species. Wood Science and Technology, 45: 339-357. https://doi.org/10.1007/s00226-010-0323-9
- Safdari, V.; Moinuddin, A.; Palmer, J.; Baig, M. B., 2008: Identification of iranian commercial wood with hand lens. Pakistan Journal of Botany, 40 (5): 1851-1864.
- Tarmian, A.; Azadfallah, M.; Gholamiyan, H.; Shahverdi, M., 2011: Inter-tracheid and cross-field pitting in compression wood and opposite wood of norway spruce (*Picea abies* L.). Notulae Scientia Biologicae, 3 (2): 145-151.
- Topaloğlu, E.; Erisir, E., 2018: Longitudinal variation in selected wood properties of Oriental Beech and Caucasian Fir. Maderas. Ciencia y Tecnología, 20 (3): 403-416. http://dx.doi.org/10.4067/S0718-221X2018005031101.
- Tutuş, A.; Kurt, R.; Alma, M. H.; Meriç, H., 2010: Chemical analysis and thermal properties of Scots pine wood. In Proceedings of III. National Black Sea Forestry Congress, pp. 1845-1851 (in Turkish).
- Wise, E. L.; Karl, H. L., 1952: Cellulose and hemicelluloses in pulp and paper science and technology. New York: McGraw Hill-Book Co.
- Yaman, B.; Köse, N.; Özel, H. B.; Şahan, E. A., 2020: The effect of climate on the radial growth of Oriental Beech in the Southern limit of its distribution area. Forestist, 70 (1): 53-59. https://doi.org/10.5152/forestist.2020.19024
- Yazıcı, H.; Özlüsoylu, İ., 2020: The effect of accelerated weathering on some surface properties of heat-treated sweet bay wood (*Laurus nobilis* L.). Turkish Journal of Forestry, 21 (4): 468-474. https://doi.org/10.18182/ tjf.809139
- 39. \*\*\*OGM., 2021: https://www.ogm.gov.tr/tr/duyurularsitesi/Documents/ORMAN%20GE NEL%20M%C3%9C D%C3%9CRL%C3%9C%C4%9E%C3%9C%20 2021%20YILI%20PERFORMANS%20PROGRA-MI%20%C4%B0.pdf (Accessed Oct. 20, 2021).
- 40. \*\*\*OGM., 2021: https://www.ogm.gov.tr/tr/e-kutu p h a n e s i t e s i / Y a y i n l a r / A s l i % 2 0 A % C 4 % 9Fa%C3%A7%20T%C3%BCrleri.pdf (Accessed Oct. 20, 2021).
- 41. \*\*\*OGM., 2022: Orman genel müdürlüğü faaliyet raporu: Ankara (Accessed Oct. 20, 2022).

#### **Corresponding address:**

### TUBA KÜLÇE

Kastamonu University, Forestry and Nature Tourism Specialization Coordinatorship, Kastamonu, TURKEY, e-mail: tubakulce@kastamonu.edu.tr

Yang Li<sup>1</sup>, Tao Yao<sup>1</sup>, Yong Zhu<sup>1</sup>, Shengquan Liu<sup>1</sup>, Zuju Shu<sup>2</sup>, Redžo Hasanagić<sup>3</sup>, Leila Fathi<sup>4</sup>, Demiao Chu<sup>1</sup>

# Thermal Modification Intensity of Heat-treated Poplar Wood Part 1: Characterization and Predication of Surface Layer

# Intenzitet toplinske modifikacije topolovine. Dio 1: Karakterizacija i predikcija površinskog sloja

# **ORIGINAL SCIENTIFIC PAPER**

**Izvorni znanstveni rad** Received – prispjelo: 8. 6. 2023. Accepted – prihvaćeno: 9. 10. 2023. UDK: 630\*84; 674.047.3 https://doi.org/10.5552/drvind.2023.0125 © 2023 by the author(s). Licensee University of Zagreb Faculty of Forestry and Wood Technology. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license.

**ABSTRACT** • Wood heat treatment is an environmentally friendly method, and the heat-treated wood properties are closely related to thermal modification intensity. This study focuses on the 0-3 mm surface layer (SL) of poplar wood heat treated at 160~220 °C. The modification intensity, including surface color, hardness, chemical component and morphological changes of the SL, was evaluated. The findings of this research showed that the color difference of the poplar wood before and after heat treatment ( $\Delta E^*_{,l}$ ) increased; the color difference between up-surface and down-surface of the SL ( $\Delta E^*_{,l}$ ) also increased with the treatment temperature. Consequently, the surface hardness ( $H_R$ ) decreased with the increase of treatment intensity. When the treatment temperature was higher than 160 °C, the up-surface and down-surface of the SL were statistically different in color. Chemical component analysis revealed that the heat treatment degrades wood components, especially the hemicellulose, and correlation analysis showed a significant correlation between the change rate of hemicellulose and the  $\Delta E^*_{,l}$ or  $H_R$  value; the prediction functions have been established at a high confidence level of 0.99. Overall, the thermal modification intensity of the heat-treated surface layer (SL) of poplar wood varies, and the  $H_R$  and  $\Delta E^*_{,l}$  value could be used to characterize and predict the modification intensity and degree of thermal degradation of the surface layer of heat-treated poplar wood.

**KEYWORDS:** *heat treatment; modification intensity; hardness; color; poplar wood* 

**SAŽETAK** • Toplinska obrada drva ekološki je prihvatljiva metoda, a svojstva toplinski modificiranog drva usko su povezana s intenzitetom toplinske modifikacije. Ovo je istraživanje usredotočeno na površinski sloj (SL) od

<sup>&</sup>lt;sup>1</sup> Authors are researchers at Anhui Agricultural University, School of Forestry & Landscape Architecture, Key Lab of State Forest and Grassland Administration on "Wood Quality Improvement & High Efficient Utilization", Hefei, China.

<sup>&</sup>lt;sup>2</sup> Author is researcher at Anhui Agricultural University, School of Textile Engineering and Academy of Art, Hefei, China.

<sup>&</sup>lt;sup>3</sup> Author is researcher at University of Bihać, Faculty of Technical Engineering, Department of Wood Science and Technology, Bihać, Bosnia and Herzegovina.

<sup>&</sup>lt;sup>4</sup> Author is researcher at Shahrekord University, Department of Natural Resources and Earth Science, Shahrekord, Iran.

0-3 mm toplinski modificirane topolovine pri 160 – 220 °C. Istražen je utjecaj intenziteta modifikacije na boju površine, tvrdoću, kemijski sastav i morfološke promjene površinskog sloja. Rezultati su pokazali da se s povećanjem temperature modifikacije povećala razlika u boji topolovine prije i nakon modifikacije ( $\Delta E^*_{1}$ ) te razlika u boji između gornje i donje površine površinskog sloja ( $\Delta E^*_{2}$ ). Posljedično se tvrdoća površine ( $H_R$ ) smanjila s povećanjem intenziteta modifikacije. Pri temperaturi modifikacije višoj od 160 °C razlika u boji između gornje i donje površine površinskog sloja statistički je značajna. Analiza kemijskog sastava otkrila je da toplinska modifikacija razgrađuje komponente drva, posebice hemicelulozu, a korelacijskom je analizom utvrđena značajna međusobna ovisnost između stupnja promjene hemiceluloze i vrijednosti  $\Delta E^*_{1}$  i  $H_R$ . Utvrđene su funkcije predviđanja s visokom razinom pouzdanosti od 0,99. Sve u svemu, intenzitet toplinske modifikacije topolovine u površinskom sloju varira, a vrijednosti  $H_R$  i  $\Delta E^*_{1}$  mogu se iskoristiti za karakterizaciju i predviđanje intenziteta modifikacije i stupnja toplinske razgradnje površinskog sloja toplinski modificirane topolovine.

KLJUČNE RIJEČI: toplinska obrada; intenzitet modifikacije; tvrdoća; boja; topolovina

# **1 INTRODUCTION**

# 1. UVOD

Lignocellulosic materials are thought to be one of the most promising alternative resources for fossil fuels. Worldwide attention has been focused on environment-friendly material, and biomass-based functional materials have been applied in various fields (Li *et al.*, 2018; Wu *et al.*, 2021). Wood is renewable and degradable with a high strength-to-weight ratio, and it could also provide a comfortable indoor environment because of its decorative color and humidity-control properties (Zhang *et al.*, 2016).

As one of the main artificial trees in China, poplar wood could provide a large amount of timber for uses such as paper making and wood-based panels (Yang et al., 2020). However, poplar wood also has many shortcomings, such as low density, dimensional stability, and durability (Korkut et al., 2008), which prevents its use as solid wood material. Many modification methods like acetylation (Mantanis, 2017), resin impregnation (Xu et al., 2021), densification (Srivaro et al., 2021) and natural, eco-friendly chemicals (Primaningty as et al., 2021), and heat treatment (Chu et al., 2020) were applied to provide a better performance and economic benefit to poplar wood and broaden its application field (Dong et al., 2020). Heat treatment deepens the color of wood (Liu et al., 2021) and reduces wood moisture absorption, enhances dimensional stability (Gu et al., 2019), and improves biological durability (Gérardin, 2016). Therefore, demands for heat-treated wood have increased rapidly worldwide (Hill et al., 2021), and its application has grown dramatically in various fields, including furniture, architectural decoration, etc. (Gurleyen et al., 2017).

Studies show that heat treatment influences the chemical properties of wood through changes in the chemical content and structure of chemical components such as extractives, hemicellulose, cellulose, and lignin (Miklečić *et al.*, 2016). The mechanical properties of treated wood are affected by the reaction medium, which mainly includes steam, inert gas (nitrogen), air, hot oil,

or water (Rahimi et al., 2019). High-temperature heat treatment of wood usually reduces its mechanical properties (Wu et al., 2019; Widmann et al., 2012), and increases overall brittleness (Hughes et al., 2015; Yildiz et al., 2006). Guo et al. (2016) used the stress-strain curve to divide the elastic and plastic regions and found that the brittleness of wood increased significantly with the increase in thermal modification intensity. Schneid et al. (2014) used the same method to calculate the brittleness of the heat-treated wooden chair; the result showed that the brittle was highly correlated with thermal mass loss. Fu et al. (2020) found that the change of chromatic value can accurately predict the mechanical properties of heattreated wood, and that the surface color was closely related to mechanical properties (Nasir et al., 2019). Lengowski et al. (2021) found that the hardness of the heat-treated teak wood decreased due to the degradation of amorphous polysaccharides during the heat treatment process, and that the hardness was related to thermal mass loss.

In practice, the processing performance of heattreated wood is closely associated with the properties of the surface layer of a certain thickness, and the chemical or physical characteristics of wood surfaces affect the gluing and coating processes (Gu et al., 2019; Hill et al., 2021). Thermal modification intensity of the surface layer is an influential factor in determining the processing performance and quality of heat-treated wood, according to the theory of mechanical weak boundary layer (MWB) (Stehr et al., 2020; Chu et al., 2020). Wang (2011) studied the thermal effect of vacuum heat treatment on eucalyptus grandis and found that the sample dimension had a significant impact on unit mass heat absorption, leading to uneven heat transfer in the direction of thickness. Ramdane et al. (2015) found a significant difference in temperature with the increase of sample thickness based on transient analysis of the heat and mass transfer process. The thermal degradation intensity of wood surface layer is higher than that of the inner part of wood, because of the low thermal conductivity of wood and the heat and mass diffusion, and the moisture or contacting medium difference.



**Figure 1** Flow chart of sample preparation **Slika 1.** Dijagram toka pripreme uzorka

In our previous studies, it has been reported that the heat treatment process declined the surface abrasion and Shore D hardness, as well as bonding capacity (Chu et al., 2020), and that an enhanced surface layer could be generated by an impregnated chemical agent and thermal compression (Chu et al., 2019). However, there are few studies on the degree of thermal degradation of the heat-treated wood at 0-3 mm surface layer to the best of the authors knowledge, although there are several studies involving the overall properties of the wood (Schneid et al., 2014; Costa et al., 2020). This study aims to study the influence of temperature and duration on thermal modification intensity of heattreated poplar wood at 0-3 mm surface layer and establish a potential fitting model for revealing the variation of degradation degree of the heat-treated wood regarding the surface layer. The results would provide the basis for evaluating the processability of the heat-treated poplar wood.

### 2 MATERIALS AND METHODS

2. MATERIJALI I METODE

#### 2.1 Sample preparation

2.1. Priprema uzoraka

The 12-year-old Zhonglin 46 poplars (*Populus euramericana 'zhonglin 46'*) were selected, and the straight part of the air-dried lumber was cut into 200 mm  $\times$  150 mm  $\times$  30 mm sapwood boards. A self-made high-temperature experiment device was used to conduct high-temperature heat treatment on the poplar wood samples, and a cistern was installed as a steam generating device.

The poplar wood samples were put evenly in the experiment box, the temperature of the inside of the box was raised from room temperature to 130 °C at a heating rate of 20 °C·min<sup>-1</sup>, kept for 30 minutes, and then the box was set at a heating rate of 10 °C·min<sup>-1</sup>. The internal temperature was raised to the target treatment temperature (160 °C, 190 °C, 220 °C), for 2h and 4h. The experimental group numbers were  $H_{160-2}$ ,  $H_{160-4}$ ,  $H_{190-2}$ ,  $H_{190-4}$ ,  $H_{220-2}$ ,  $H_{220-4}$ , the control group was the untreated wood. Each group had ten to fifteen duplicate specimens.

The surface layer of the heat-treated poplar wood with a thickness of 3 mm was named *SL*, and its upper-

surface and bottom-surface were marked as  $S_u$  and  $S_b$ , respectively. Then, the *SL* part of the heat-treated wood was smashed into wood powder for further use.

#### 2.2 Measurement of color and hardness of heat-treated poplar

### 2.2. Mjerenje boje i tvrdoće toplinski modificirane topolovine

According to the CIE1976LAB standard colorimetric system, a precision colorimeter (HP-200, Shenzhen Hampoo) was used to obtain the surface color parameters. There were 10 replicates for each experimental group, and 5 points were randomly selected on the upper-surface  $(S_u)$  and bottom-surface  $(S_b)$  of the heat-treated wood for color measurement. Color difference value  $\Delta E^*$  was calculated as follows:

$$\Delta L_{l}^{*} = L_{h} - L \tag{1}$$

$$\Delta a_{l}^{*} = a_{h} - a \tag{2}$$

$$\Delta b_l^* = b_h - b \tag{3}$$

$$\Delta E_{1}^{*} = [(\Delta L_{1}^{*})^{2} + (\Delta a_{1}^{*})^{2} + (\Delta b_{1}^{*})^{2}]^{1/2}$$
(4)

$$\Delta E_2^* = [(\Delta L_2^*)^2 + (\Delta a_2^*)^2 + (\Delta b_2^*)^2]^{1/2}$$
(5)

Where,  $L_h$ ,  $a_h$ ,  $b_h$  are the chromatic parameters of the poplar wood after HT, and L, a, b are the parameters of wood before the heat treatment;  $\Delta L_2^*$ ,  $\Delta a_2^*$ ,  $\Delta b_2^*$  are the chromatic parameters difference of  $S_u$  and  $S_b$  on the heat-treated wood,  $\Delta E_2^*$  is the color difference between  $S_u$  and  $S_b$  of the heat-treated wood on the surface layer.

Pressure ball hardness test uses a high-precision double-column universal testing machine (AG-X plus, Shimadzu, Japan), using a hemispherical steel indenter mold, in accordance with the national standard GB/T1941-2009 Wood hardness test method. The hemispherical steel indenter was pressed into the specimen surface of 2 mm at a uniform speed of 1 mm/min, and the pressing load at the pressed depth of 0.5 mm, 1.0 mm, 1.5 mm, and 2.0 mm was recorded and marked as *P*. Each experimental group was repeated 15 times. The surface hardness (H<sub>R</sub>) was calculated based on the indentation load of the untreated poplar wood (Yu *et al.*, 2020).

$$H_{R} = P_{l} / P_{0} \tag{6}$$

 $P_1$  – pressing load on 1 mm surface of heat-treated wood, unit: N;

 $P_{0}$  – pressing load on 1 mm surface of untreated wood, unit: N.

#### 2.3 Chemical component and morphological analysis

# 2.3. Kemijske komponente i morfološka analiza

Triplicates were used to perform quantitative chemical analysis of control and modified wood samples. According to the Chinese standards of wet chemical analysis of wood components shown in Table 1, the hemicellulose content was estimated by the difference between holocellulose and a-cellulose. Each group had three replicates, and the average value was calculated.

FT-IR analysis was performed on the samples (potassium bromide compressed tablets) by Fourier infrared spectrometer. Samples were scanned in the range of 390-4000 cm<sup>-1</sup> with a 4 cm<sup>-1</sup> resolution, and each sample was analyzed by 32 scans.

The experimental equipment used an XRD-6 Xray diffractometer. The X-ray tube was a copper target and the wavelength  $\lambda$ =1.5406 nm. The test temperature was room temperature, the tube voltage was 40 kV, and the tube current was 30 mA. Surface sample wood powder was dried and pressed into thin sheets at room temperature. Then a 20 diffraction intensity curve was drawn. The scanning angle 20 ranged from 5° to 35°, the scanning rate was 2 °/min, the step width was 0.01, and the diffraction intensity curve was recorded. It was output by the plotter immediately, and each sample was sampled twice, and the average value was taken.

The surface of untreated wood and heat-treated wood were selected, a *SL*ide *SL*icer (SM2010R, Leica, Germany) was used to cut the surface of heat treated poplar. Scanning electron microscopy (SEM) scans of the tangential section of the wood chips were obtained using a scanning electron microscope FEG-XL30 (FEI, USA). Samples were covered by galvanic gold deposition using an MC1000 ion sputter (FEI, USA) with a current of 5 mA for 45 s. The analyses were performed with an acceleration voltage of 20 kV.

Tablica 1.	Standardi z	a ispitivanje	e toplinski	modificirane
topolovine				

Measured items Mjerene sastavnice	<b>Testing standard reference</b> Ispitni standardi
Moisture content sadržaj vode	GB/T 36,055–2018
Benzene-alcohol extractives benzen-alkoholni ekstrakti	GB/T 35,816–2018
Holocellulose holoceluloza	GB/T 35,818–2018
Cellulose / celuloza	GB/T 744–2004
Lignin / lignin	GB/T 35,818–2018

## 2.4 Data analysis and modeling of degradation intensity at surface layer

# 2.4. Analiza podataka i modeliranje intenziteta razgradnje površinskog sloja

The correlations between the chemical component content,  $\Delta E_I^*$ , hardness, and crystallization values of untreated and heat-treated samples were calculated, and the correlation matrix was obtained using Origin software. Then, the prediction models of the wood degradation intensity at surface layer were established based on the hemicellulose change rate and the color changes before and after the heat treatment.

#### 3 RESULTS AND DISCUSSION 3. REZULTATI I RASPRAVA

# 3.1 Color and hardness of heat-treated wood on the surface layer

### 3.1. Boja i tvrdoća površinskog sloja toplinski modificiranog drva

Surface color and hardness are prominent properties of furniture, flooring, and other wooden materials. The heat treatment usually darkens the surface and decreases the hardness of wood, which could reflect the degradation intensity and brittleness degree of the heat-treated wood. Table 2 summarizes the chromatic values and hardness of the surface layers (*SL*) of heattreated poplar.

As presented in Table 2 and Figure 2, the lightness value  $L^*$  of the SL of heat-treated poplar is significantly lower than that of untreated poplar. The redgreen index  $a^*$  and yellow-blue index  $b^*$  are much higher. Among the chromatic parameters, the change in  $L^*$  value is the most obvious; it decreases with the increase of heat treatment temperature. The  $L^*$  of H<sub>160-2</sub> and H<sub>220-2</sub> are 84.69 and 34.58, respectively, which is 3.39 % and 60.55 % lower than that of untreated poplar. The  $a^*$  and  $b^*$  values first increased and then decreased with the increase of the heat treatment temperature; both  $a^*$  and  $b^*$  reached their maximum at the temperature of 190 °C. Heat treatment duration has a minor effect on the changing of wood surface color; the  $L^*$  value of  $H_{160-4}$  was 3.64 % lower than that of  $H_{160-2}$ . There was no notable difference in the chromatic parameters between  $\rm H_{220\mathchar`2}$  and  $\rm H_{220\mathchar`4}$  . Numerous studies have shown that the surface color of heat-treated wood becomes darker and tends to be red and yellow (Mantanis, 2017), and that the change of surface color is closely related to the heat treatment temperature (Chen *et al.*, 2018; Cao *et al.*, 2018). The  $\Delta E_1^*$  of  $H_{220-2}$  is 10.97 times higher than that of  $H_{160-2}$ .

The surface color of  $S_u$  and  $S_b$  of the *SL* in the same condition of heat-treated poplar is different, and the  $\Delta E_2^*$  value ranges from 0.89 to 3.42, according to Table 2. The increase of  $\Delta E_2^*$  value indicated that the

Groups		<b>Color change</b> Promjena boje			<b>Pressing load at different depth, N</b> Tlačno opterećenje pri različitoj dubini, N				
Skup	ine	$\Delta L^*$	$\Delta E_{I}^{*}$	$\Delta E_2^*$	0.5 mm	1.0 mm	1.5 mm	2.0 mm	
Control		-	-	-	77.52±5.76	166.80±4.85	254.68±11.60	368.81±13.88	
TT	S <sub>u</sub>	2.97	4.46	0.80	52 16 6 57	110 49 10 72	171 (0) 15 04	224.06117.27	
п <sub>160-2</sub>	S <sub>b</sub>	2.32	4.34	0.89	52.10±0.57	110.46±10.72	1/1.09±13.04	234.80±17.57	
II	Su	24.37	27.2	2 72	22 27+5 16	82 24 + 5 22	159 66+0 16	240 66 1 17 85	
П <sub>190-2</sub>	S <sub>b</sub>	21.86**	24.52**	2.73 32.27±5.16	83.24±3.32	138.00±9.10	240.00±17.83		
II	Su	53.08	53.46	2 4 2	19.07+2.06	59.50+12.57	129 00 1 19 20	202.03±17.88	
п <sub>220-2</sub>	S <sub>b</sub>	50.24**	50.87**	3.42	18.97±3.00	38.39±12.37	128.00±18.20		
II	Su	6.05	8.58	2.22	69 52 16 52	155 (2) 5 (7	245 02+0 68	220 92 19 54	
п <sub>160-4</sub>	S <sub>b</sub>	5.02*	6.55**	2.33	08.33±0.32	133.02±3.07	243.95±9.08	559.65±18.54	
11	S <sub>u</sub>	35.36	37.23	2.92	45.9(+0.01	110 27 22 42	220.20+20.01	210.01 + 29.40	
П <sub>190-4</sub>	S <sub>b</sub>	33.39*	35.76	2.83	45.80±9.01	118.2/±22.45	220.20±29.91	519.01±28.49	
П	Su	53.13	53.62	202	27 47 7 70	84 44 + 14 70	162.82±23.19	260.82 + 46.77	
п <sub>220-4</sub>	S <sub>b</sub>	50.52**	51.15**	2.83	37.47±7.70	84.44±14.79		200.85±40.77	

 Table 2 Chromatic values and pressing load of surface layer of heat-treated poplar wood

 Tablea 2. Kromatske vrijednosti i tlačno ispitivanje površinskog sloja toplinski modificirane topolovine

\* and \*\* mean significant differences of 0.05 and 0.01 between the  $S_u$  and  $S_b / * i ** znači da su za S_u i S_b izmjerene značajno različite vrijed$ nosti pri <math>p < 0.05 i p < 0.01



**Figure 2** CIELab values (a) and hardness (b) of heat-treated poplar wood at surface layer **Slika 2.** CIELab vrijednosti (a) i tvrdoća (b) površinskog sloja toplinski modificirane topolovine

degradation intensity difference between  $S_{b}$  and  $S_{a}$  increased with the increase of heat treatment temperature. Besides, the  $L^*$  value of  $S_u$  is lower than that of  $S_u$ in all heat-treated groups. Variance analysis shows that in all heat-treated groups (except  $H_{160-2}$ ), the  $\Delta L^*$  and  $\Delta E_{i}^{*}$  values are significantly different at the 0.05 level, which means the degree of thermal degradation of heat-treated poplar on the SL is different. When the heating temperature is higher than 190 °C, the uneven heating on the surface and inside leads to the difference in the degree of thermal degradation. Shi and Bao (2021) found that the extractives migrated from the core layer to the surface during the thermal modification process, which was also one of the reasons for the color difference of the wood surface layer. The possible reason for an inconsiderable difference of H<sub>160-2</sub> is

that the heat treatment is relatively low, and the *SL* is not significantly thermally degraded.

Figure 2 represents the  $H_R$  of heat-treated wood at the *SL*, which is the degree of surface plastic deformation to evaluate its ability to resist external force pressing. With the increase of the press-in depth, the ability of the compressed part to resist external force is enhanced due to the densification of the surface material. The  $H_R$ of the *SL* is calculated from the load when the press-in depth is 1 mm. After the heat treatment, the  $H_R$  of wood was reduced to different degrees, and the  $H_R$  value decreased with an increase in the heat treatment temperature in the duration of 2 hours. The  $H_R$  values of  $H_{160-2}$ ,  $H_{190-2}$ , and  $H_{220-2}$  decreased by 34 %, 50 %, and 65 %, respectively, compared with that of untreated wood. Previous studies stated that the hardness of wood is sensi-



**Figure 3** SEM images of heat-treated poplar surface and roughness of cell wall **Slika 3.** SEM slike površine toplinski modificirane topolovine i hrapavosti stanične stijenke

tive to the change in hemicellulose content (Konnerth et al., 2010; Borůvka et al., 2018). Hemicellulose acts as a cell wall-filling component and its degree of degradation increases with the increase of the degree of heat treatment. The  $H_p$  value of the SL decreased with the increase of heat treatment temperature when the duration was 4 hours, as can be seen in Figure 2. However, the  $H_p$  value of the SL on  $\rm H_{160\text{-}4},\, \rm H_{190\text{-}4},$  and  $\rm H_{220\text{-}4}$  was higher than that of  $H_{160-2}$ ,  $H_{190-2}$ , and  $H_{220-2}$ , respectively. It could be explained by the comprehensive effects of the changing on crystallinity and the cross-linked lignin network during the long duration of the heat treatment process. When poplar wood was treated at 160 °C, the degradation of wood cell wall components was relatively low, and the crystallinity of wood increased with the HT duration, which caused the increase in  $H_{R}$ . When the heating temperature is higher, the crystallinity of wood decreases with the heat treatment duration, and the relative content of lignin increases with the degradation of hemicellulose and cellulose. In addition, the degradation products of hemicellulose and lignin also form cross-linked lignin networks, which leads to an increase in  $H_p$  (Altgen *et al.*, 2018). Xi (2018) found that crystallinity and lignin contents also have a great influence on wood hardness.

Existing research also showed that the destruction of the cell wall structure reduces the hardness of the

heat-treated wood (Aytekin et al., 2021; Chu et al., 2020). The surface cell wall and the roughness of the heat-treated wood were observed by scanning electron microscope as shown in Figure 3. It can be seen that the untreated wood surface was smooth, and the cell wall structure was undamaged. When the poplar wood was treated at 160 °C, the surface and pit structure did not change significantly. The roughness analysis showed that the sample surface was relatively smooth. So, the low-temperature heat treatment did not have a significant indigenous effect on the cell wall structure. With the increase in heat treatment temperature, the surface of  $H_{190-2}$  and  $H_{220-2}$  was cracked and gradually expanded, and some cell walls were stratified. The pit membrane deforms, and the pits become hollow and rupture, resulting in cracks and shedding. The increase of burrs on the surface made the heat-treated wood rougher, and the cell wall structure was damaged and collapsed. The cell wall structure of  $H_{220-2}$  was significantly changed and severely damaged mainly due to further degradation of hemicellulose and partial pyrolysis of lignin at the high temperature of 220 °C (Budakçı et al., 2016). Wang (2017) found that the changes in the pit structure were due to a pressure gradient generated by the degradation of cell wall components, which resulted in the increase of internal pressure of wood.

#### 3.2 Chemical component changes of surface layer of heat-treated poplar wood

# 3.2. Promjene kemijskih komponenata u površinskom sloju toplinski modificirane topolovine

The change of the chemical composition is the basic reason for the change of properties of the heattreated wood. Table 3 documents the results of the wet chemical analysis of the surface layer (SL) of heattreated poplar wood.

Among all wood components, hemicellulose is the most severely degraded by thermal treatment, and it is the main reactive substance during the heat treatment process; this result is consistent with the study of Dubey et al. (2012). Hemicellulose showed a downward trend with the increase of heat treatment temperature and duration; the hemicellulose contents of H<sub>160.4</sub>, H<sub>190.4</sub>, and H<sub>220-4</sub> were 31.42 %, 21.75 %, and 15.12 %, respectively. On the contrary, the extractives content increases with the increase of heat treatment temperature and duration. Especially, the content of the extractives of  $H_{220.4}$  is 10.13%, which is 3.26 times higher than that of untreated wood. Hemicellulose degrades to produce small molecules (Candelier et al., 2020), which probably increases the extractives content. The content of cellulose and lignin are relatively stable, but when the heat treatment temperature is higher than 190 °C, cellulose also degrades to a certain extent (Chien et al., 2018), and lignin branched-chain breaks (Yang et al., 2021); the relative content of these two components increases because of the decrease of hemicellulose.

The FT-IR spectra of the chemical functional group changes of poplar wood before and after heat treatment are shown in Figure 4. The experiment mainly analyzes the change of characteristic peaks in the range of 1800 cm<sup>-1</sup>-750 cm<sup>-1</sup>. There is a subtle change of characteristic peak at 1424 cm<sup>-1</sup> after the HT process, which is assigned to the shear vibration peak of cellulose. Therefore, this peak could be chosen to normalize the spectrum (Lei et al., 2021). Table 4 represents the ratio of each peak.

As shown in Figure 4(a), the absorbance at 1735 cm<sup>-1</sup> is assigned to the hemicellulose acetyl non-conjugated carbonyl group (Wentzel et al., 2019); its peak is reached relatively smoothly with the increase of heat treatment temperature mainly due to the degradation of hemicellulose. The higher the heat treatment temperature, the more severe the degradation of the hemicellulose. The absorption peak at 1163 cm<sup>-1</sup> is attributed to the C-O-C stretching vibration of cellulose and hemicellulose. The increase in the intensity of this peak is mainly due to the formation of ether bonds by the hydroxyl groups on hemicellulose and cellulose during the heat treatment. However, the peak at 1163 cm<sup>-1</sup> was reduced for  $H_{220-2}$  and  $H_{220-4}$ , indicating that the ether bond broke at a high heat treatment temperature of 220 °C. The change of absorption peak near 1058 cm<sup>-1</sup>, which is attributed to the C-O shrinkage vibration of cellulose and hemicellulose, indicated that aldols products are formed

Table 3 Chemical components and XRD of surface layer of heat-treated poplar wood Tablica 3. Kemijske komponente i XRD površinskog sloja toplinski modificirane topolovine

Groups Skupine		Chemical of Promjena	compositio kemijskog		C <b>rystallization</b> Kristalizacija		
	Hemicellulose	Cellulose	Lignin	Benzene-alcohol extractives	$I_{002}(2\theta)$	$I_{am}(2\theta)$	CI <sub>XRD</sub>
Control	34.83	46.63	23.71	2.38	22.22	18.62	42.09
H <sub>160-2</sub>	32.79	47.22	23.50	2.24	22.20	18.48	44.82
H <sub>190-2</sub>	25.91	47.69	20.43	4.67	22.26	18.86	51.38
H <sub>220-2</sub>	17.05	44.19	26.81	8.30	22.24	18.44	56.35
H <sub>160-4</sub>	31.42	46.35	21.36	2.53	22.16	18.58	45.33
H <sub>190-4</sub>	21.75	49.60	20.83	5.14	22.26	18.52	51.27
H <sub>220-4</sub>	15.12	46.86	27.88	10.13	22.46	18.92	53.05

Table 4 Intensity ratio of each peak after normalizaton treatment at 1424 cm<sup>-1</sup> Tablica 4. Omjer intenziteta svakog vrha nakon normalizacije na 1424 cm<sup>-1</sup>

Groups Skupine	I1735	I1603	I1512	I1463	I1264	I1163	I1058
Control	0.865	0.801	0.737	0.974	1.840	0.904	2.560
H <sub>160-2</sub>	0.782	0.824	0.725	0.944	1.620	1.577	2.542
H <sub>190-2</sub>	0.773	0.864	0.733	0.962	1.553	1.538	2.659
H <sub>220-2</sub>	0.709	0.847	0.810	0.985	1.518	1.385	2.234
H <sub>160-4</sub>	0.816	0.810	0.741	0.966	1.425	1.598	2.534
H <sub>190-4</sub>	0.846	0.787	0.757	0.978	1.375	1.551	2.603
H <sub>220-4</sub>	0.707	0.900	0.833	1.013	1.360	1.387	2.280





Figure 4 Infrared spectra (a) and X-ray diffraction curves (b) of surface layer of heat-treated poplar Slika 4. Infracrveni spektar (a) i difrakcijska krivulja rendgenskih zraka (b) površinskog sloja toplinski modificirane topolovine

during the heat treatment process at relatively low temperature, while the aldols undergo condensation reaction at the high heat treatment temperature of  $220 \,^{\circ}$ C.

The stretching vibration peak at 1650 cm<sup>-1</sup> is assigned to the C-O group in lignin, mostly located on the propyl branch of lignin. Besides, absorbance at wave numbers 1603 cm<sup>-1</sup> and 1264 cm<sup>-1</sup> is assigned to the conjugate stretching vibration of the C=O and the benzene ring skeleton of lignin and the stretching vibration of the G ring and the acyl-oxygen bond C=O in the lignin (Gao, 2019). With the increase of heat treatment temperature, the peak intensity at 1650 cm<sup>-1</sup> and 1264 cm<sup>-1</sup> decreased, while the peak intensity at 1603 cm<sup>-1</sup> increased, which could also support the conclusion that the relative content of lignin increases. Furthermore, the cross-linking reaction of the lignin occurs at the same time as its thermal degradation. Absorbance at 1512 cm<sup>-1</sup> and 1463 cm<sup>-1</sup> is the benzene ring carbon skeleton vibration peak of lignin and the deformation vibration peak of C-H<sub>2</sub> in lignin and carbohydrates. The two peaks increased with the increase of the heat treatment temperature, indicating an increase in relative lignin content (Table 3). The result is consistent with that of wet chemical analysis discussed above.

The X-ray diffraction pattern is shown in Figure 4(b). Within the scanning interval of 0-35°, two prominent peaks appear on the curve. A peak valley at about  $2\theta=18^{\circ}$  represents the scattering intensity  $I_{am}$  of the diffraction angle of the amorphous region. The peak located near  $2\theta=22^{\circ}$  indicates the maximum intensity  $I_{002}$  of the diffraction angle of the crystal region. The crystallinity value can reflect the degree of crystal formation during the cellulose condensation process, and it is closely related to elastic modulus, impact toughness, and mechanical properties (Sun *et al.*, 2019). Table 4 lists the crystallization characteristics of *SL* of heattreated poplar calculated according to  $I_{am}$  and  $I_{002}$ .

After the heat treatment, the positions of the  $I_{002}$  diffraction peaks of the samples were all around 22°,

and the crystal layer distance did not change. It means that the heat treatment did not significantly change the crystallization characteristics. When the treatment duration was 2h, the crystallinity of heat-treated wood increased from 42.09 to 56.35, with the increase in heat treatment temperature. It is worth noting that the  $CI_{xRD}$ of H<sub>160-4</sub> increased slightly compared with that of H<sub>160-2</sub>, and there was no evident difference between  $\mathrm{H}_{\mathrm{190-2}}$  and H<sub>190-4</sub>. Nevertheless, a significant decline was found for  $H_{220.4}$  as the duration increased from 2h to 4h. The degradation of hemicellulose and amorphous-cellulose increases with the increase of heat treatment duration at a relatively low temperature of 160 °C, which accordingly increases the relative percentage of cellulose. Akguiil et al. (2007) found that the degradation products of hemicellulose could also be transformed into a crystalline area. With the increase of heat treatment temperature, part of the cellulose began to decompose, and the  $CI_{XRD}$  of  $H_{220.4}$  declined with the extension of heat treatment duration. Additionally, hemicellulose degradation generates acid substances, accelerating the deterioration of cellulose and lignin and decreases the wood crystallinity (Yang et al., 2021).

### 3.3 Correlation analysis and prediction of degradation intensity at surface layer

### 3.3. Korelacijska analiza i predviđanje intenziteta razgradnje površinskog sloja

Figure 5 represents the correlation analysis of the degradation characteristics, including the color, hardness, crystallization, and the change rate of chemical components.

As shown in Figure 5(a), the SPSS analysis results show that the color difference  $\Delta E_1^*$  and change rate of hemicellulose and benzene-alcohol extractives are all significantly correlated at the 0.05 statistical level. The hardness of wood was significantly correlated with the change rate of hemicellulose, while the relative crystallinity of cellulose was significantly negatively correlated with the change rate of hemicellu-



**Figure 5** Correlation matrix of chemical component change rate (a) and fitting curve of change rate of hemicellulose with color difference and hardness on surface layer of heat-treated poplar wood (b) **Slika 5.** Korelacijska matrica stupnja promjene kemijske komponente (a) i usklađivanje krivulja stupnja promjene hemiceluloze s razlikom u boji i s tvrdoćom površinskog sloja toplinski modificirane topolovine (b)

lose. The lower hemicellulose content, the higher crystallinity of wood; crystallinity also affects the hardness of wood. As shown in Figure 5(b), the  $H_R$  of the heat-treated poplar at the surface layer decreases with an increase in hemicellulose change rate; the functional equations of the fitted curve are *y*=*x*-0.40, (*R*<sup>2</sup>=0.993); *y*=-0.0068 · *x*+0.69, (*R*<sup>2</sup>=0.992).

The color difference  $\Delta E_1^*$  increases linearly with the changing rate of hemicellulose, and the fitted model is *y*=-0.0090·*x*+1.03, (*R*<sup>2</sup>=0.994). The degree of fitting is very good, indicating that the hardness and surface color difference of the heat-treated wood surface layer is highly linearly related to the change of hemicellulose content. Therefore, they could characterize the thermal degradation of the heat-treated wood to a certain extent. This result agrees with previous studies, wherein Costa *et al.* (2020) stated that the degradation of polysaccharides could be the reason for the increased modification intensity of the high-temperature treated wood.

## **4 CONCLUSIONS**

# 4. ZAKLJUČAK

The thermal modification intensity of the 160 °C, 190 °C, and 220 °C heat-treated wood at 0-3 mm surface layer was studied, and wet chemical analysis and correlation analysis were applied to reveal the relationship of surface color and hardness changes with the degree of thermal degradation of the treated wood regarding chemical changes.

The surface layer of the heat-treated wood becomes dark and consequently the surface hardness decreases with the increase of modification intensity, and the colors between the upper-surface and bottom-surface are significantly different. Heat treatment causes a change in the chemical composition of poplar wood, and the thermal mass loss of wood mainly occurs due to hemicellulose degradation, which significantly changes the cell wall structure and could be listed as the main reason for the increase in modification intensity.

The correlation analysis shows that the color difference  $(\Delta E_1^*)$  and surface hardness  $(H_R)$  have a high correlation with the change rate of hemicellulose, and a functional relationship,  $H_R$  and  $\Delta E_1^*$ , could be used as a simple and rapid method to characterize the thermal modification degree of the surface layer of heat-treated wood.

## Acknowledgements - Zahvala

This research was funded by "Natural Science Foundation of Anhui Province, grant number No. 2008085QC130, No. 2008085MC84; The University Synergy Innovation Program of Anhui Province, No. GXXT-2021-053; Science and Technology Major Project of Anhui province, No. 202003b06020010".

# **5 REFERENCES**

# 5. LITERATURA

- Akgül, M.; Gümüşkaya, E.; Korkut, S., 2007: Crystalline structure of heat-treated Scots pine [*Pinus sylvestris* L.] and Uludağ fir [*Abies nordmanniana (Stev.) subsp. bornmuelleriana (Mattf.)*] wood. Wood Science and Technology, 41(3): 281-289. https://doi.org/10.1007/s00226-006-0110-9
- Altgen, M.; Uimonen, T.; Rautkari, L., 2018: The effect of de- and re-polymerization during heat-treatment on the mechanical behavior of Scots pine sapwood under quasi-static load. Polymer Degradation and Stability, 147: 197-205. https://doi.org/10.1016/j.polymdegradstab.2017.12.007
- Aytekin, A.; Yazıcı, H., 2021: Physical, Anatomical and Photochemical Analyses of Some Exotic Wood Species Submitted to Heat Treatment. Journal of Renewable Materials, 9 (8): 1485-1501. https://doi.org/10.32604/ jrm.2021.015768

- Borůvka, V.; Eidler, A.; Holeček, T.; Dudík, R., 2018: Elastic and Strength Properties of Heat-Treated Beech and Birch Wood. Forests, 9 (4): 197. https://doi. org/10.3390/f9040197
- Budakçı, M.; Pelit, H.; Sönmez, A., 2016: The Effects of Densification and Heat Post-Treatment on Hardness and Morphological Properties of Wood Materials. BioResources, 11 (3): 7822-7838. https://doi.org/10.15376/biores.11.3.7822-7838
- Candelier, K.; Thévenon, M.; Collet, R.; Gérardin, P.; Dumarçay, S., 2020: Anti-fungal and anti-termite activity of extractives compounds from thermally modified ash woods. Maderas. Ciencia y Tecnología, 22 (2): 223-240. https://doi.org/10.4067/S0718-221X2020005000209
- Cao, Y. J.; Li, X. W.; Wang, S.; Li, Y. X.; Li, W. J., 2018: Effects of Thermal Treatment on Color in *Eucalyptus urophylla* Wood. Forestry and Environmental Science, 34 (01): 18-20.
- Chen, X. Z.; Zhang, A. W.; Chen, L. C.; Lu, R. J.; Li, K.; Li, Y. J., 2018: Effect of Hydrothermal Treatment on Color of *Pinus massoniana* Wood. Zhejiang For Sci Technol, 38 (06): 1-7. https://doi.org/10.3969/j. issn.1001-3776.2018.06.001
- Chien, Y.; Yang, T.; Hung, K.; Li, C.; Xu, J., 2018: Effects of heat treatment on the chemical compositions and thermal decomposition kinetics of Japanese cedar and beech wood. Polymer Degradation and Stability, 158: 220-227. https://doi.org/10.1016/j.polymdegradstab.2018.11.003
- Chu, D. M.; Mu, J.; Avramidis, S.; Rahimi, S.; Lai, Z. Y.; Ayanleye, S., 2020: Effect of heat treatment on bonding performance of poplar via an insight into dynamic wettability and surface strength transition from outer to inner layers. Holzforschung, 74 (8): 777-787. https://doi. org/10.1515/hf-2019-0145
- Chu, D. M.; Mu, J.; Avramidis, S.; Rahimi, S.; Liu, S. Q.; Lai, Z. Y., 2019: Functionalized Surface Layer on Poplar Wood Fabricated by Fire Retardant and Thermal Densification. Part 2: Dynamic Wettability and Bonding Strength. Forests, 10 (11): 982. https://doi.org/10.3390/ f10110982
- Chu, D. M., 2019: Manufacturing and characterizing of the surface functional layer on poplar using combined treatment of fire retardancy and thermal modification. Beijing Forestry University.
- Chu, D. M.; Mu, J.; Avramidis, S.; Rahimi, S.; Lai, Z. Y.; Ayanleye, S., 2020: Effect of heat treatment on bonding performance of poplar via an insight into dynamic wettability and surface strength transition from outer to inner layers. Holzforschung, 74 (8): 777-787. https://doi. org/10.1515/hf-2019-0145
- Costa, H. W. D.; Coldebella, R.; Andrade, F. R., 2020: Brittleness increase in Eucalyptus wood after thermal treatment. International Wood Products Journal, 11 (1): 38-42. https://doi.org/10.1080/20426445.2020.1719298
- Dong, Y. M.; Liu, X. Y.; Liu, R. H.; Wang, W.; Li, J. Z., 2020: Research Progress and Development Trend of Wood Chemical Cross-Linking Modification. China Forest Products Industry, 57 (11): 10-14. https://doi. org/10.19531/j.issn1001-5299.202011003
- Dubey, M. K., Pang, S., Walker, J., 2012: Changes in chemistry, color, dimensional stability and fungal resistance of *Pinus radiata* D. Don wood with oil heat-treatment. Holzforschung, 66 (1): 49-57. https://doi. org/10.1515/HF.2011.117
- 17. Fu, Z.; Zhou, F.; Gao, X.; Weng, X.; Zhou, Y., 2020: Assessment of mechanical properties based on the changes

of chromatic values in heat treatment wood. Measurement, 152: 107215. https://doi.org/10.1016/j.measurement.2019.107215

- Gao, Y. L., 2019: Study on the Changes and Its Mechanism of Moisture Adsorption and Absorption Properties of High Temperature Heat Treated Chinese Fir Wood. Chinese Academy of Forestry.
- Gérardin, P., 2016: New alternatives for wood preservation based on thermal and chemical modification of wood – a review. Annals of Forest Science, 73 (3): 559-570. https://doi.org/10.1007/s13595-015-0531-4
- Gu, L. B.; Ding, T., 2019: Jiang N. Development of wood heat treatment research and industrialization. Journal of Forestry Engineering, 4 (4): 1-11. https://doi. org/10.13360/j.issn.2096-1359.2019.04.001
- Guo, F.; Huang, R. F.; Lv, J. X.; Gao, Z. Q., 2016: Effect of Heat Treatment Parameters on Brittleness of Masson Pine Wood. China Wood Industry, 30 (3): 9-12. https:// doi.org/10.19455/j.mcgy.2016.03.003
- 22. Gurleyen, L.; Ayata, U.; Esteves, B.; Cakicier, N., 2017: Effects of heat treatment on the adhesion strength, pendulum hardness, surface roughness, color and glossiness of Scots pine laminated parquet with two different types of UV varnish application. Maderas. Ciencia y Tecnología, 19 (2): 213-214. https://doi.org/10.4067/S0718-221X2017005000019
- Hill, C.; Altgen, M.; Rautkari, L., 2021: Thermal modification of wood – a review: chemical changes and hygroscopicity. Journal of Materials Science, 56 (11): 6581-6614. https://doi.org/10.1007/s10853-020-05722-z
- Hughes, M.; Hill, C.; Pfriem, A., 2015: The toughness of hygrothermally modified wood COST Action FP0904 2010-2014: Thermo-hydro-mechanical wood behavior and processing. Holzforschung, 69 (7): 851-862. https:// doi.org/10.1515/hf-2014-0184
- Konnerth, J.; Eiser, M.; Jäger, A.; Bader, T. K.; Hofstetter, K.; Follrich, J.; Tersv, T.; Hansmann, C.; Wimmer, R., 2010: Macro- and micro-mechanical properties of red oak wood (*Quercus rubra* L.) treated with hemicelluloses. Holzforschung, 64 (4): 447-453. https://doi. org/10.1515/hf.2010.056
- Korkut, S., 2008: The effects of heat treatment on some technological properties in Uludağ fir (*Abies bornmuellerinana Mattf.*) wood. Building and Environment, 43 (04): 422-428.
- Lei, W. C.; Zhang, Y. M.; Yu, W. J.; Yu, Y. L., 2021: The adsorption and desorption characteristics of moso bamboo induced by heat treatment. Journal of Forestry Engineering, 6 (03): 41-46. https://doi.org/10.13360 /j. issn.2096-1359.202010008
- Lengowski, E. C.; Bonfatti, E. A.; Nisgoski, S., 2021: Properties of thermally modified teakwood. Maderas. Ciencia y Tecnología, 10 (23): 1-16. https://doi. org/10.4067/S0718-221X2021000100410
- 29. Li, T.; Song, J. W.; Zhao, X. P.; Yang, Z.; Pastel, G.; Xu, S. M.; Jia, C.; Dai, J. Q.; Chen, C. J.; Gong, A.; Jiang, F.; Yao, Y. G.; Fan, T. Z.; Yang, B.; Wågberg, L.; Yang, R. G.; Hu, L. B., 2018: Anisotropic, lightweight, strong and super thermally insulating nanowood with naturally aligned nanocellulose. Science Advances, 4 (03): 3724. https://doi.org/10.1126/sciadv.aar3724.
- Liu, G. J.; Hou, J.; Xu, J. M., 2021: A Review of the Research on Wood Color. World Forestry Research, 34 (04): 46-53. https://doi.org/10.13348/j.cnki.sjlyyj.2021.0019.y
- Mantanis, G. I., 2017: Chemical Modification of Wood by Acetylation or Furfurylation: A Review of the Present

Scaled-up Technologies. BioResources, 12 (2): 4478-4489. https://doi.org/10.15376/biores.12.2.4478-4489

- 32. Miklečić, J.; Jirouš-Rajković, V., 2016: Influence of Thermal Modification on Surface Properties and Chemical Composition of Beech Wood (*Fagus sylvatica L.*). Drvna industrija, 67 (1): 65-71. https://doi.org/10.5552/ drind.2016.1520
- 33. Nasir, V.; Nourian, S.; Avramidis, S.; Cool, J., 2019: Prediction of physical and mechanical properties of thermally modified wood based on color change evaluated by means of "group method of data handling" (GMDH) neural network. Holzforschung, 73 (4): 381-392. https:// doi.org/10.1515/hf-2018-0146
- 34. Primaningtyas, A.; Listyanto, T.; Lukmandaru, G.; Weasa, B. M., 2020: Modification of nanochitosan with NaCl and surfactant for wood preservative. IOP Conference Series: Earth and Environmental Science, 449 (1): 12012-12023. https://doi.org/10.1088/1755-1315/449/1/ 012023
- Rahimi, S.; Singh, K.; DeVallance, D., 2019: Effect of Different Hydrothermal Treatments (Steam and Hot Compressed Water) on Physical Properties and Drying Behavior of Yellow-Poplar (*Liriodendron tulipifera*). Forest Products Journal, 69 (1): 42-52. https://doi. org/10.13073/FPJ-D-18-00028
- 36. Ramdane, S.; Younsi, R.; Lachemet, A.; Kocaefe, D., 2015: Transient analysis of heat and mass transfer during heat treatment of wood including pressure equation. Thermal Science, 19 (2): 693-702. https://doi. org/10.2298/TSCI120309113Y
- Schneid, E.; Gonzalez De-Cademartori, P. H.; Gatto, D., 2014: The effect of thermal treatment on physical and mechanical properties of Luehea divaricata hardwood. Maderas. Ciencia y Tecnología, 16 (4): 413-422. https:// doi.org/10.4067/S0718-221X2014005000033
- Shi, Q.; Bao, F. C., 2021: Analyzing chromogenic group of heat treated wood by GC-MS. Forestry and Grassland Machinery, 2 (03): 16-23. https://doi.org/10.13594/j. cnki.mcjgjx.2021.03.003
- 39. Srivaro, S.; Lim, H.; Li, M.; Jantawee, S.; Tomad, J., 2021: Effect of compression ratio and original wood density on pressing characteristics and physical and mechanical properties of thermally compressed coconut wood. Construction and Building Materials, 299: 124272. https://doi.org/10.1016/j.conbuildmat.2021.124272
- Stehr, M.; Johansson, I., 2000: Weak boundary layers on wood surfaces. Journal of Adhesion Science and Technology, 14 (10): 1211-1224. https://doi. org/10.1163/156856100742168
- 41. Sun, H. Y.; Su, M. L.; Lu, J. X.; Zhao, R. J.; Ren, H. Q., 2019: Wang YR. Research progress on effect of microfibril angle andcrystalline area in cell wall on wood physical and mechanical properties. Journal of Northwest A & F University (Natural Science Edition), 47 (05): 50-58. https://doi.org/10.13207/j.cnki.jnwafu.2019.05.007
- 42. Wang, X. H., 2011: Thermal effect and mechanism of Eucalyptus pellita woodunder vaccum heat treatment. Chinese Academy of Forestry.

- 43. Wang, Z., 2017: Study on properties and mechanism of larch wood modified by vacuum heat treatment. Chinese Academy of Forestry.
- 44. Wentzel, M.; Rolleri, A.; Pesenti, H.; Militz, H., 2019: Chemical analysis and cellulose crystallinity of thermally modified Eucalyptus nitens wood from open and closed reactor systems using FTIR and X-ray crystallography. European Journal of Wood and Wood Products, 77 (4): 517-525. https://doi.org/10.1007/s00107-019-01411-0
- 45. Widmann, R.; Fernandez-Cabo, J. L.; Steiger, R., 2012: Mechanical properties of thermally modified beech timber for structural purposes. European Journal of Wood and Wood Products, 70 (6): 775-784. https://doi. org/10.1007/s00107-012-0615-x
- Wu, Y. Q., 2021: Newly advances in wood science and technology. Journal of Central South University of Forestry & Technology. 41 (01): 1-28. https://doi. org/10.14067/j.cnki.1673-923x.2021.01.001
- 47. Wu, Z. X.; Chen, Y. H.; Huang, C. J.; Li, J. P.; Bao, Y. J.; Sun, F. W., 2019: A Review of Effects of Heat Treatment on Wood Mechanical Properties. World Forestry Research, 32 (1): 59-64. https://doi.org/10.13348 /j.cnki. sjlyyj.2018.0104.y
- 48. Xi, E.; 2018: Dynamic relationship between mechanical properties and chemical composition distribution of wood cell walls. Wood Research, 63 (2): 179-192.
- 49. Xu, J. F; Qin, Y.; Liu, R.; Long, L.; Ma, E., 2021: Preparation and properties of high-performance fast-growing wood modified by exfoliated organo-montmorillonite in waterborne hyperbranched polyacrylate emulsion. Construction and Building Materials, 298: 123868. https://doi.org/10.1016/j.conbuildmat.2021.123868
- Yang, S.; Fu, Y. J.; Yan, T. T.; Chen, Y., 2021: Effect of high temperature heat-treatment on the chemical properties of *Swietenia macrophylla*. Journal of Forestry Engineering, 6 (02): 120-125. https://doi.org/10.13360 /j. issn.2096-1359.202006016
- Yang, Y.; Zhang, L Song, F. F.; Zhang, J. J.; Zhang, Z. F., 2020: Research Progress on High Value Utilization of Fast-growing Wood in Plantation. China Forest Products Industry, 57 (05): 53-55. https://doi.org/10.19531/j. issn1001-5299.202005012
- Yildiz, S.; Gezer, E. D.; Yildiz, U. C., 2006: Mechanical and chemical behavior of spruce wood modified by heat. Building and Environment, 41 (12): 1762-1766. https:// doi.org/10.1016/j.buildenv.2005.07.017
- 53. Yu, H. Q.; An, X.; Zhao, Y. K.; Li, X. L., 2020: Study on Hardness Test Methods and Standards of Wood and Wood-based Panels. China Wood Industry, 34 (05): 40-43. https://doi.org/10.19455/j.mcgy.20200509
- 54. Zhang, X. B.; Ziling, W.; Künzelb, H. M.; Mitterer, C.; Zhang, X., 2016: Combined effects of sorption hysteresis and its temperature dependency on wood materials and building enclosures. Part II: Hygrothermal modeling. Building and Environment, 10: 181-195. https://doi. org/10.1016/j.buildenv.2016.06.033

#### **Corresponding address:**

#### **DEMIAO CHU**

Anhui Agricultural University, School of Forestry & Landscape Architecture, Key Lab of State Forest and Grassland Administration on "Wood Quality Improvement & High Efficient Utilization, No. 130, Changjiang West Road, CHINA, e-mail: Demiaochu@ahau.edu.cn

Sefa Durmaz<sup>1</sup>, Ugur Aras<sup>2</sup>, Erkan Avci<sup>1</sup>, Yusuf Ziya Erdil<sup>1</sup>, Ilkay Atar<sup>3</sup>, Hulya Kalaycioglu<sup>2</sup>

# Influence of Zinc Oxide Nanoparticles on Flame Resistance in Wood Plastic Composites

# Utjecaj nanočestica cinkova oksida na vatrootpornost drvno-plastičnih kompozita

# **ORIGINAL SCIENTIFIC PAPER**

**Izvorni znanstveni rad** Received – prispjelo: 15. 6. 2023. Accepted – prihvaćeno: 30. 8. 2023. UDK: 674.049.3; 674.816 https://doi.org/10.5552/drvind.2023.0132 © 2023 by the author(s). Licensee University of Zagreb Faculty of Forestry and Wood Technology. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license.

**ABSTRACT** • The interest in wood plastic composites (WPCs) has increased in recent years. The utilization of environmentally friendly materials has been of great significance due to the overwhelming pressure on nature. As a widely used material, plastic is, however, easily combustible due to its structure. In this study, WPCs were reinforced with zinc oxide (ZnO) nanoparticles. The effect of higher content of ZnO nanoparticles (1, 3, 5, 10%) on WPCs thermal stability and fire performance was investigated. Thermogravimetric analysis (TGA) clearly demonstrated that nanoparticles acted as a shield, which inhibited heat transfer and increased the degradation temperature thanks to covering the surface of materials. Free radicals accelerated the thermal degradation of neat-HDPE (high-density polyethylene) by oxidative reactions, while ZnO nanoparticles reduced the degradation velocity. Moreover, the increase in nanoparticle content significantly affected the residue. The fire performance of WPCs was also investigated by the limit oxygen index (LOI) test. While neat-HDPE flamed with dripping, ZnO nanoparticles made flaming difficult for WPCs. Therefore, the LOI values increased with increasing nanoparticle content up to 28.5 %, which indicated the need for more oxygen. The improvement reached up to 54 % compared to neat HDPE. Moreover, the char forming was also improved, which helped enhance the fire resistance. The scanning electron microscope (SEM) investigation indicated that nanoparticles were well dispersed in the matrix. However, the tendency to agglomerate increased with the increase of concentration. The ability of carbonization of wood fiber surface during the combustion also contributed to improving thermal stability and fire performance.

**KEYWORDS:** ZnO nanoparticles; LOI test; TGA analysis; thermal resistance; WPC

**SAŽETAK** • Zanimanje za drvno-plastične kompozite (WPC) posljednjih je godina poraslo. Upotreba ekološki prihvatljivih materijala ima veliko značenje zbog golemog pritiska na okoliš. Plastika je široko rasprostranjen i vrlo često upotrebljavan materijal, ali je zbog svoje strukture lako zapaljiva. U ovom su istraživanju drvno-plastični kompoziti ojačani nanočesticama ZnO. Istraživan je utjecaj većeg udjela nanočestica cinkova oksida (ZnO) (1, 3, 5 i 10 %) na toplinsku stabilnost i vatrootpornost WPC-a. Termogravimetrijska analiza (TGA) jasno je pokazala da nanočestice djeluju kao štit koji inhibira prijenos topline i povećava temperaturu razgradnje. Nadalje, slobodni radikali u reakcijama oksidacije ubrzali su toplinsku razgradnju čistog HDPE-a (polietilena visoke gustoće), dok su nanočestice ZnO smanjile brzinu razgradnje. Štoviše, povećanje udjela nanočestica znatno je utjecalo na

<sup>&</sup>lt;sup>1</sup> Authors are researchers at Muğla Sıtkı Koçman University, Muğla, Turkey. https://orcid.org/0000-0002-3880-0033; https://orcid.org/0000-0002-1475-4028; https://orcid.org/0000-0003-3938-2168

<sup>&</sup>lt;sup>2</sup> Authors are researchers at Karadeniz Technical University, Trabzon, Turkey. https://orcid.org/0000-0002-1572-0727; https://orcid.org/0000-0002-1807-4353

<sup>&</sup>lt;sup>3</sup> Author is researcher at Kahramanmaraş Sütçü İmam University, Kahramanmaraş, Turkey. https://orcid.org/0000-0001-9527-1791

ostatak nakon razgradnje. Vatrootpornost WPC-a također je ispitana mjerenjem graničnog indeksa kisika (LOI). Dok je čisti HDPE gorio uz kapanje, nanočestice ZnO otežale su gorenje WPC-a. LOI vrijednosti rasle su s povećanjem udjela nanočestica do 28,5 %, što je upućivalo na veću potrebu za kisikom. Poboljšanje je iznosilo do 54 % u usporedbi s čistim HDPE-om. Štoviše, poboljšano je i stvaranje pougljenjenog sloja koji je pridonio povećanju vatrootpornosti. Istraživanje pretražnim elektronskim mikroskopom (SEM) pokazalo je da su se nanočestice dobro dispergirale u matrici. Međutim, tendencija aglomeraciji povećala se s povećanjem koncentracije nanočestica. Sposobnost karbonizacije površine drvnih vlakanaca tijekom gorenja također je pridonijela poboljšanju toplinske stabilnosti i vatrootpornosti WPC-a.

KLJUČNE RIJEČI: nanočestice ZnO; LOI test; TGA analiza; toplinska otpornost; WPC

# **1 INTRODUCTION**

# 1. UVOD

Climate change is increasing the pressure on the environment, driving manufacturers to produce greener products. Today, however, it is not only the desire for green materials that is more important, but also efficiency. The scarcity of raw materials also makes efficiency vital. Composite materials meet these requirements. Composites consist of two or more materials with different properties than the sum of the properties of two components. Wood-based composites are among the most popular composites because of their green label. Wood is a renewable, biodegradable, and abundant material, distinguishing it from other materials. The properties of wood vary between species, within the same species, and even more between different parts of the same tree. However, composites made from wood have homogeneous properties that attract designers, architects, engineers, etc. Today, woodbased composites are evaluated for everything from furniture to structures, car parts to unique designs.

WPCs, one of the wood-based composites, have existed for over 50 years in various applications: decks, fences, railing, garden furniture, doors, and window frames (Kim and Pal, 2010; Klyosov, 2007). Mixing wood and plastic creates a new material with superior properties. Plastic is one of the most remarkable inventions of the 20th century. Its high resistance to outdoor conditions makes it a more practical material against factors such as UV light, moisture, insects, fungi, etc. However, wood does not have the same advantages. On the contrary, it can be easily degraded by biotic and abiotic factors under the right conditions. Therefore, WPCs have become more desirable in recent years due to their high durability, high strength, low cost, noncorrosion, and environmental friendliness. Moreover, they can be considered an alternative material for most industries such as construction, car, architectural and structural design, aerospace, etc. (Erchiqui et al., 2020; Smith and Wolcott, 2006).

Composites are recognized as superior in comparison with their own constituents. However, there is still a need for improvement to overcome the existing shortcomings. Most research has focused on improving WPCs physical, mechanical, and weathering durability. However, the high flame sensitivity of polymer restricted WPC application, mainly indoors and outdoors, due to higher heat during thermal degradation. Heat or temperature differentiations affect WPCs thermal properties (Guo et al., 2019a). Free radicals initiate oxidation reactions during thermal degradation, decreasing the thermal stability of polymers (He et al., 2012). The chain scissions resulted in mass losses with increasing temperature due to thermaloxidative reactions. Eventually, most of the structure is converted to carbon monoxide (CO), carbon dioxide (CO<sub>2</sub>), and water (H<sub>2</sub>O) (Devi and Maji, 2012). In addition, the polymer threatens structural integrity, especially in structural design, due to its formability at low temperatures.

On the other hand, wood is a biopolymer consisting of cellulose, hemicellulose, and lignin. When exposed to heat or temperature, there are also some changes in the chemical and physical structure of wood. The cross-linked structure makes lignin more resistant to high temperatures, while hemicellulose is vulnerable (Lowden and Hull, 2013). Acetic and formic acid, furfural, and methanol formation occur during high-temperature exposure, which degrades cellulose and hemicellulose amorphous structures (Özgenç et al., 2017). Meanwhile, thermal degradation involves liquefaction, gasification, and pyrolysis (Mohan et al., 2006). Liquefaction gradually decreases and turns into some gases and solids; CO, CO<sub>2</sub>, and H<sub>2</sub>O, followed by gasification, which starts below 200 °C with non-combustible gases. Combustible gases also begin to accompany the process as the temperature rises, which starts the combustion of the combustible material. This produces char, which breaks down the wood.

The incompatibility between wood and polymer is also a crucial issue for the thermal stability of WPCs, as well as for their physical and mechanical properties. The bond between the polymer and wood flour, which are hydrophobic and hydrophilic, respectively, has a negative effect on the thermal expansion of WPCs (Yang *et al.*, 2005).

The effect of various fillers on the physicochemical and mechanical properties of polymer composites is determined by many factors. Moreover, some studies have also been carried out to improve the thermal stability of WPC with the reinforcement of synthetic fibers. Rasana *et al.* (2019) stated that glass fibers act as a thermal shield and inhibit degradation. The degradation of carbon fibers above 600 °C, which is higher than polyethylene and wood, improves the thermal stability of WPCs (Eibl, 2017; Guo *et al.*, 2019b; Yao *et al.*, 2018). As in previous studies, the thermal degradation of WPCs was retarded with the reinforcement of glass and carbon-woven fabrics (Durmaz *et al.*, 2021). Moreover, some studies also dealt with improving the thermal stability and fire performance of fibers with aerogels (He *et al.*, 2022; Zhang *et al.*, 2023).

Nanotechnology applications have been quite prevalent in recent years. The small size of nanoparticles surpasses their traditional counterparts and remarkably improves the materials properties (De Filpo et al., 2013). Nanomaterials improve the mechanical and physical properties of WPCs, and also significantly improve the thermal stability and fire performance (Chaharmahali et al., 2014). Metal oxide nanoparticles are the preferred nanoparticles for most applications. ZnO nanoparticle is one of the metal oxides evaluated in various applications due to its inherent properties (Franco-Urquiza et al., 2020). Zhao and Li (2006) highlighted that the thermal stability of polypropylene-based WPCs with ZnO nanoparticles was maintained even after exposure to UV light. Devi and Maji (2012) also found that the degradation temperature improved with the thermal shielding ability of ZnO nanoparticles. The small size of the nanoparticles allows them to cover the material surface, inhibiting the free radicals that occur during thermal degradation. As a result, the degradation temperature was improved. Jouyandeh et al. (2019) also found that increasing the content thickens the nanoparticle layer, which acts as an insulator and reduces heat transfer. Moreover, ZnO nanoparticles can absorb UV light between 200-380 nm (He et al., 2009). As in previous studies, ZnO nanoparticles improved weathering resistance with increasing content. The loss of mechanical properties was also reduced with decreasing crack formation (Durmaz et al., 2022).

High thermal stability and fire resistance are essential for the expansion of indoor and outdoor applications of WPCs. The main objective of this study was to improve the thermal strength and fire resistance of HDPE-based WPCs with ZnO nanoparticle reinforcement. For this purpose, the effect of ZnO nanoparticles at high contents on the WPCs thermal stability and fire performance was investigated. The distribution of ZnO nanoparticles was investigated by SEM analysis. The thermal behavior of the reinforced WPCs was investigated using TGA. The LOI test also revealed the fire behavior of the reinforced WPCs.

# 2 MATERIALS AND METHODS

2. MATERIJALI I METODE

# 2.1 Materials

#### 2.1. Materijali

The pine wood flour (Pinus sylvestris L.) with 40-60 mesh dimensions as a lignocellulosic filler and HDPE as a thermoplastic polymer were supplied from commercial suppliers (Ucar Plastic, İzmir, Turkey). HDPE is one of the world's most popular plastics due to its resistance to high temperatures, strong chemicals, and UV light. The fine-grain polymer (~200 mesh) was used to obtain a homogenous mixture. The melt flow index (MFI) and density of HDPE were 5.5 g/10 min (190 °C/2.16 kg) and 0.965 g/cm<sup>3</sup>, respectively. Maleic anhydride grafted polyethylene (Licocene PE MA 4351 Fine Grain) was used as a coupling agent to improve the bonding between the materials. The MAPE softening point and density were 123 °C and 0.99 g/ cm<sup>3</sup>, respectively. The ZnO nanoparticles with dimensions of 30-50 nm and a density of 5.5 g/cm<sup>3</sup> were used as a reinforcement agent (Nanografi, Ankara, Turkey).

### 2.2 Methods

#### 2.2. Metode

The wood flour, thermoplastic polymer, and nanoparticles were mixed in a mechanical mixer (1200 rev/min) to obtain a homogeneous mixture, according to the scheme in Table 1. Before manufacturing, wood flour was oven-dried until constant weight at 80 °C. Then, the homogeneous mixture was extruded with a single screw extruder (Teknomatik, Turkey). The screw speed was set to 40 rpm. The extruder temperature was from 180 to 195 °C. The extruded samples were cooled in water and pelletized. The pellets were dried at (102±2) °C and hot pressed at 180 °C for 15 min (CemilUsta SSP 125, Istanbul, Turkey). The reinforcement of ZnO nanoparticles decreased the melt flow index of the polymer. The pressure of the hot press was 2.35-

Table 1 Wood flour, polymer, and nanoparticle content (%) Tablica 1. Sadržaj drvnog brašna, polimera i nanočestica (%)

<b>Groups</b> <i>Grupe</i>	Wood/Polymer ratio	Nanoparticle content
-	Omjer drvo /	Sadržaj nanočestica
	polimer	
Neat-HDPE	0/100	-
WPC40	40/60	-
WPC50	50/50	-
WPC40-1	40/59	1
WPC40-3	40/57	3
WPC40-5	40/55	5
WPC40-10	40/50	10
WPC50-1	50/49	1
WPC50-3	50/47	3
WPC50-5	50/45	5
WPC50-10	50/40	10

2.55 MPa. The dimensions of the panels were 500 mm  $\times$  500 mm  $\times$  4 mm.

#### 2.3 Thermal properties

### 2.3. Toplinska svojstva

TGA indicates the difference in the weight of samples as a function of temperature or time. DTGA is also obtained by taking the first derivative of TGA curves as a temperature or time function. The WPC samples underwent grinding utilizing an IKA grinder with a 1 mm sieve. The thermal stability of WPCs was determined with Perkin–Elmer STA 6000 thermogravimetric analyzer 4000 (USA). Samples were heated from 30 °C to 600 °C with a heating rate of 10 °C/min under nitrogen gases. Two samples were tested for each group.

The LOI test is one of the most prominent test methods for characterizing the fire behavior of materials due to its ability to analyze small samples. The LOI test determines the minimum oxygen content for the continuing flame combustion of the samples. The LOI test was performed with the Dynisco LOI analyzer (Franklin, USA) (Figure 1) according to ASTM D 2863-19 (2019). The LOI was measured on five samples for each group (with one sample measuring 127 mm  $\times$  12.7 mm  $\times$  5 mm). Each sample was placed in the center of a glass tube in which a mixture of oxygen and nitrogen gas was released. After the gas mixture had expanded the glass tube, the sample was burned with a pilot fire. The sample was monitored during the 50 mm of burning or flaming for 180 s. The higher LOI values indicate resistance to flammability, while the lower LOI values are associated with flammable materials.

### 2.4 SEM analysis 2.4. SEM analiza

The WPC samples were investigated by a scanning electron microscope (Zeiss, Evo LS10). Before



Figure 1 LOI instrument Slika 1. LOI uređaj

the investigation, WPC samples were dried until constant weight and coated with gold (Emitech, SC7620).

#### 2.5 Statistical analysis

#### 2.5. Statistička analiza

A comprehensive statistical analysis was conducted to explore the data, employing a robust approach involving analysis of variance (ANOVA) followed by post-hoc Duncan testing (significance level set at p < 0.05). This rigorous statistical methodology allowed for a thorough examination of the dataset, and consequently meaningful conclusions could be drawn based on the observed differences among the variables.

# **3 RESULTS AND DISCUSSION**

3. REZULTATI I RASPRAVA

# **3.1 TGA analysis 3.1. TGA analiza**

The behavior of materials against gradual temperature increase is essential for applications. The effect of ZnO nanoparticles on the gradual temperature increase (10 °C/min) was examined by TGA analysis, as seen in Table 2. The degradation of WPCs occurred in three temperature stages, as shown in Figure 2. As the temperature increased, the weight loss of the WPCs increased. First, the moisture released from the samples accelerated after 50 °C, causing a slight weight loss (~3-4 %). The primary degradation (2<sup>nd</sup> stage) was initiated after 180 °C. It is well-known that the main wood cell wall components, hemicellulose, cellulose, and lignin, begin to degrade above this temperature (Jarukumjorn and Suppakarn, 2009; Rowell, 2012).

It is well known that the thermal stability of wood is lower than that of polymer. Therefore, the first degradation was initiated in the wood cell wall components (Ramesh et al., 2022). Hemicellulose is the most sensitive to thermal degradation among the other cell wall components - degradation of hemicellulose followed by cellulose and lignin. In addition, the increase in wood content has a negative effect on the weight loss of WPCs. As shown in Table 2, the higher the wood content, the higher the weight loss. As the wood content increases, the encapsulation of the wood fibers decreases, which weakens the thermal stability. Meanwhile, the carbonization of wood fiber improves the thermal stability of the polymer due to the thermal shielding effect (Guo et al., 2019a). In addition, the wood content also affects the residue. Wood can contain up to  $\sim 5$  % ash, contributing to the increase in residue.

The degradation temperature of the second stage varied from 175.19 to 204.24 °C, in which hemicelluloses started to degrade first, followed by others. The final degradation stage (above 390 °C) was the decomposition of HDPE. Under the inert atmosphere, the py-
<b>Groups</b> Grupe	<b>Stage</b> Stadij	T onset, ℃	T endset, ℃	T deg., °C	Weight loss, % Gubitak mase, %	Weight onset, % Masa na početku, %	Weight endset, % Masa na kraju, %
Neat-HDPE		327.29	526.47	473	96.70	98.0916	1.395
WDC40	2 <sup>nd</sup> stage	179.34	398.47	345.64	27.42	98.703	71.286
WPC40	3 <sup>nd</sup> stage	398.47	529.82	473.96	60.72	71.286	10.568
WDC50	2 <sup>nd</sup> stage	175.19	396.71	344.52	34.17	98.095	63.926
WPC50	3 <sup>rd</sup> stage	396.71	523.27	449.17	51.44	63.926	12.487
WDC40_1	2 <sup>nd</sup> stage	181.25	398.95	347.4	26.26	98.426	72.169
WPC40-1	3 <sup>rd</sup> stage	398.95	529.18	474.28	60.36	72.23	11.87
WDC40.2	2 <sup>nd</sup> stage	188.6	398.95	346.76	26.76	98.469	71.712
WPC40-3	3 <sup>rd</sup> stage	398.95	530.62	473.32	58.85	71.712	12.867
WDC40.5	2 <sup>nd</sup> stage	191.31	397.51	347.56	26.87	98.81	71.94
WPC40-5	3 <sup>rd</sup> stage	397.51	526.94	469.81	57.04	71.94	14.9
WDC40 10	2 <sup>nd</sup> stage	204.08	395.91	346.28	26.18	98.691	72.51
WPC40-10	3 <sup>rd</sup> stage	395.91	523.11	468.37	52.69	72.51	19.825
WDC50 1	2 <sup>nd</sup> stage	189.39	399.74	347.4	31.98	97.887	65.912
WPC30-1	3 <sup>rd</sup> stage	399.74	530.3	474.44	52.95	65.912	12.959
WDC50 2	2 <sup>nd</sup> stage	180.93	402.3	349.31	34.71	98.189	63.476
WPC50-5	3 <sup>rd</sup> stage	402.3	529.5	473.16	49.01	63.476	14.469
WDC50 5	2 <sup>nd</sup> stage	204.24	399.58	346.28	31.64	98.084	66.445
WPC30-3	3 <sup>rd</sup> stage	399.58	529.98	471.08	48.10	66.445	18.349
WDC50 10	2 <sup>nd</sup> stage	202.32	404.05	350.75	36.21	96.715	60.509
WPC50-10	3 <sup>rd</sup> stage	404.05	544.98	466.46	28.89	60.509	31.624

**Table 2** TGA values of WPCs**Tablica 2.** TGA vrijednosti WPC-a

rolytic degradation starts the cleavage reactions, resulting in rapid weight loss (Peters, 1979). Therefore, the neat HDPE was easily degraded due to its structure. The DTGA plot also showed the rapid degradation of neat HDPE, as seen in Figure 3. As in the previous study, we found that ZnO nanoparticles limit the interaction with active surface groups by covering the surface (Durmaz *et al.*, 2022). In addition, the nanoparticles cover the materials, which scavenges oxygen from the surface and inhibits scission reactions. The synergistic effect of carbonization of wood and nanoparticles limits the  $O_2$  penetration, creating a barrier to thermal degradation. Therefore, the degradation temperatures were increased due to the heat-shielding role of nanomaterials as well as the carbonization of wood fibers. The increased degradation temperature is crucial because it retards the softening of the matrix and ceases the relaxation process (Xian *et al.*, 2023).

In addition, ZnO nanoparticles inhibited the free radicals that cause thermal oxidative reactions, which reduced the degradation rate and improved thermal stability. Increasing the content of ZnO nanoparticles also



**Figure 2** TGA thermograms of WPCs **Slika 2.** TGA termogrami WPC-a



Figure 3 DTG thermograms of WPCs Slika 3. DTG termogrami WPC-a

affected the degradation rate. The nanoparticle layer becomes thicker with increasing nanoparticle content, which increases the heat requirement for free radicals to cause scission reactions. Similarly, Uyup et al. (2019) attributed the higher thermal stability to the higher nanoparticle content. Therefore, the lowest degradation rate was observed for WPC50-10, as shown in the DTGA plot. In addition, HDPE degrades to volatile compounds with no residue above 550 °C (Dorigato et al., 2012). However, the addition of wood and nanoparticles increased the residue, which is essential for fire performance, as discussed below. The highest residue was detected from ZnO50-10. As nanoparticle content increases, they tend to agglomerate, affecting the matrix thermal conductivity, which limits the polymer chain scission reactions (Anu and Pillai, 2022). Therefore, this situation could increase the char residue for WPC-10. Xu *et al.* (2019) also highlighted the contribution of ZnO nanoparticles to char residue, which limits heat transfer and oxygen penetration.

### 3.2 LOI test 3.2. LOI test

The effect of ZnO nanoparticles on the fire performance of WPCs was examined by the LOI test. The oxygen is necessary for continuing flaming combustion. Fire-resistant materials also need much more oxygen for flaming combustion than easily combustible materials. As seen in Figure 4, the LOI values of WPCs were between 18.5 to 28.5. Due to its structure, neat HDPE is recognized as a combustible and flammable droplet thermoplastic. Therefore, it can be easily flamed. Free radicals, such as alkyl or alkyl peroxide, are formed due to oxidative reactions during thermal



**Figure 4** LOI values of WPCs, with letters indicating significant group differences (p < 0.05) determined by Duncan test **Slika 4.** LOI vrijednosti WPC-a sa slovima koja označavaju značajne grupne razlike (p < 0.05) utvrđene Duncanovim testom



**Figure 5** LOI samples **Slika 5.** Uzorci nakon LOI testa

degradation, which initiates polymer chain degradation (He *et al.*, 2012). As the degradation progresses, the free radicals are transformed into CO,  $CO_2$ , and  $H_2O$  during degradation. Therefore, the fire resistance of neat HDPE was the lowest. In addition, the conversion to CO,  $CO_2$ , and  $H_2O$  during thermal degradation results in low char formation, which reduces the fire performance of the polymer. The reduced char formation also contributes positively to the polymer dripping during the LOI test.

The lignocellulosic materials are critical in reducing the heat release rate, heat release during combustion, and mass loss rate (Borysiak et al., 2006; Kozłowski and Władyka-Przybylak, 2008). Carbonization of wood fiber during carbonization limits O<sub>2</sub> penetration from the surface to the center of the fiber, which improves fire performance. In addition, carbonization of the wood fiber enhances the char formation of WPCs, which limits the dripping of WPCs, as seen in Figure 5. This phenomenon is not neglected for structural applications. The addition of wood flour to the polymer increased the oxygen demand, resulting in higher LOI values. In addition, the polymer penetrates the wood interstices during extrusion. Nanoparticles absorbed by the polymer and deposited in the wood cells provide a barrier that scavenges oxygen from the surface and inhibits oxidation reactions, thereby improving fire performance. The deposited nanoparticles also coat the material, limiting the interaction of components with O<sub>2</sub>, which is essential for continued flaming.

The reinforcement with ZnO nanoparticles increased the LOI values up to 28.5 %. Previous studies have also shown that nanoparticles inhibit heat transfer and reduce combustible volatile compounds (Sun *et al.*, 2012; Yin *et al.*, 2022). In addition, the burning drop was improved as the nanoparticles acted as a heat shield. While the combustion behavior of neat HDPE was flaming drips, the reinforced WPCs charred without dripping, which improved the fire resistance. Therefore, the fire classification was upgraded from "flammable or combustible material" to "limited fire retardant of fireresistant material" according to ISO 4589-1 (2017), as a result of reinforced wood fibers and nanoparticles. Adding nanoparticles to WPCs improves fire performance, which is critical for indoor applications.

### **3.3 SEM analysis 3.3. SEM analiza**

The distribution of ZnO nanoparticles in the matrix is crucial in improving fire and thermal performance. Therefore, the structure of WPC samples was investigated by SEM analysis, as seen in Figure 6. SEM images showed that the nanoparticles were welldispersed in the matrix. It was thought that the powder form of polyethylene played a key role in distribution. Mainly, the lowest content resulted in a good dispersion of nanoparticles (white circles). On the other hand, as nanoparticle content increased, the tendency to agglomerate increased (white arrows). As stated in previous studies, the good dispersion of nanoparticles helped to improve the technological properties of WPCs (Durmaz et al., 2022). Likewise, ZnO nanoparticles were well dispersed in the matrix and covered the surface of materials, as clearly seen in the highest content of nanoparticles (Figure 6d, h). As stated above, nanoparticles occurred in thicker layers on the materials with increasing content, which inhibited heat transfer for the scission reaction of the polymer. Therefore, the degradation temperature increased, as explained in the TGA analysis. Moreover, the oxygen penetration was limited. Significantly, the highest nanoparticle content contributed to char formation, which descended the dripping of WPCs during the combustion and improved the fire performance of WPCs.



**Figure 6** SEM images of WPCs: a) WPC40-1, b) WPC40-3, c) WPC40-5, d) WPC40-10, e) WPC50-1, f) WPC50-3, g) WPC50-5, h) WPC50-10 (White circles show ZnO nanoparticles, while white arrows show agglomeration of ZnO nanoparticles)

**Slika 6.** SEM slike WPC-a: a) WPC40-1, b) WPC40-3, c) WPC40-5, d) WPC40-10, e) WPC50-1, f) WPC50-3, g) WPC50-5, h) WPC50-10 (bijelim kružićima označene su nanočestice ZnO, a bijele strelice pokazuju aglomeraciju nanočestica ZnO)

#### 4 CONCLUSIONS

#### 4. ZAKLJUĆAK

The petroleum-based polymers and wood are susceptible to thermal degradation. Many prior investigations have predominantly focused on assessing the impact of relatively modest quantities of ZnO nanoparticles in WPCs. However, in this study, we have elevated the thermal stability and fire-resistant characteristics of HDPE-based WPCs by augmenting them with ZnO nanoparticles, with enhancements observed at levels of up to 10%. TGA analysis also clearly demonstrated the thermal behavior of neat HDPE. The gradual increase in temperature leads to the formation of free radicals on the chemical chains of the polymer, which ultimately causes chain scission and mass loss. TGA analysis demonstrated that wood thermal stability was lower compared to polymer. However, the degradation pattern was changed with the addition of wood flour. Although the wood fibers thermally degraded at lower temperatures, the char formation improved the thermal stability. In addition, ZnO nanoparticles cover the surface of materials and insulate them from heat by acting as a heat shield. Therefore, the degradation was slowed down, and the degradation temperature increased. The surface coverage of the nanoparticles also inhibited free radicals, a key factor in the degradation rate. There was a rapid degradation for neat HDPE, while it was very moderate for ZnO nanoparticles.

Oxygen is an essential factor for combustion, as determined by the LOI test for WPCs. Carbonization of wood fiber surfaces inhibits oxygen penetration, which improves fire resistance. Therefore, only wood flour significantly improved the LOI values. In addition, nanoparticle reinforcement also supported insulation and increased oxygen demand. Meanwhile, char formation affected the fire performance. Increasing the wood flour and nanoparticle content increased the char formation, which helped to improve the fire performance. Nanoparticle reinforcement also limited dripping. SEM images indicated that nanoparticles were well-dispersed in the matrix. However, the increase in the nanoparticle content increased the tendency for aggregation. On the other hand, WPCs containing the highest nanoparticles were well covered, which clearly explained why thermal and fire performance improved. Consequently, improving thermal stability and fire performance with ZnO nanoparticles could contribute to improving WPC applications on a large scale. As the application range of WPCs widens, thermal stability is vital for safety in use. The improved thermal resistance makes WPCs safer for many applications, such as siding, decking, playgrounds, gardening furniture, windows, doors, etc.

#### Acknowledgements - Zahvala

This research was funded by The Scientific and Technological Research Council of Türkiye (TUBITAK), grant number 1210909. A part of this study was presented at the 30th International Online Conference Research for Furniture Industry on 22-23 September 2022.

#### 5 REFERENCES 5. LITERATURA

1. Anu, M. A.; Pillai, S. S., 2022: Structure, thermal, optical and dielectric properties of SnO<sub>2</sub> nanoparticles-filled

HDPE polymer. Solid State Communications, 341: 114577. https://doi.org/10.1016/j.ssc.2021.114577

- Borysiak, S.; Paukszta, D.; Helwig, M., 2006: Flammability of wood-polypropylene composites. Polymer Degradation and Stability, 91 (12): 3339-3343. https://doi. org/10.1016/j.polymdegradstab.2006.06.002
- Chaharmahali, M.; Hamzeh, Y.; Ebrahimi, G.; Ashori, A.; Ghasemi, I., 2014: Effects of nano-graphene on the physico-mechanical properties of bagasse/polypropylene composites. Polymer Bulletin, 71 (2): 337-349. https:// doi.org/10.1007/s00289-013-1064-3
- De Filpo, G.; Palermo, A. M.; Rachiele, F.; Nicoletta, F. P., 2013: Preventing fungal growth in wood by titanium dioxide nanoparticles. International Biodeterioration and Biodegradation, 85: 217-222. https://doi.org/10.1016/j. ibiod.2013.07.007
- Devi, R. R.; Maji, T. K., 2012: Effect of nano-ZnO on thermal, mechanical, UV stability, and other physical properties of wood polymer composites. Industrial and Engineering Chemistry Research, 51 (10): 3870-3880. https://doi.org/10.1021/ie2018383
- Dorigato, A.; Pegoretti, A.; Frache, A., 2012: Thermal stability of high density polyethylene – fumed silica nanocomposites. Journal of Thermal Analysis and Calorimetry, 109 (2): 863-873. https://doi.org/10.1007/s10973-012-2421-4
- Durmaz, S.; Erdil, Y. Z.; Avci, E., 2021: Improvement of technological properties of wood plastic composites reinforced with glass and carbon fibre fabric. Polymer and Polymer Composites, 29 (9): 1457-1465. https://doi. org/10.1177/09673911211054266
- Durmaz, S.; Keles, O. O.; Aras, U.; Erdil, Y. Z.; Mengeloglu, F. 2022: The effect of zinc oxide nanoparticles on the weathering performance of wood-plastic composites. Coloration Technology, 139 (4): 430-440. https:// doi.org/10.1111/cote.12666
- Eibl, S., 2017: Potential for the formation of respirable fibers in carbon fiber reinforced plastic materials after combustion. Fire and Materials, 41 (7): 808-816. https:// doi.org/10.1002/fam.2423
- Erchiqui, F.; Kaddami, H.; Dituba-Ngoma, G.; Slaoui-Hasnaoui, F., 2020: Comparative study of the use of infrared and microwave heating modes for the thermoforming of wood-plastic composite sheets. International Journal of Heat and Mass Transfer, 158: 119996. https:// doi.org/10.1016/j.ijheatmasstransfer.2020.119996
- Franco-Urquiza, E. A.; May-Crespo, J. F.; Escalante Velázquez, C. A.; Pérez Mora, R.; González García, P., 2020: Thermal degradation kinetics of ZnO/polyester nanocomposites. Polymers, 12 (8): 1753. https://doi. org/10.3390/polym12081753
- Guo, Y.; Zhu, S.; Chen, Y.; Li, D., 2019a: Thermal properties of wood-plastic composites with different compositions. Materials, 12(6): 881. https://doi.org/10.3390/ ma12060881
- Guo, Y.; Liu, D.; Chen, Y.; Zhang, T.; Zhu, S., 2019b: Preparation and properties of carbon fiber and pine cone fiber reinforced high density polyethylene composites. Journal of Applied Polymer Science, 136 (14): 47304. https://doi.org/10.1002/app.47304
- He, J.; Shao, W.; Zhang, L.; Deng, C.; Li, C., 2009: Crystallization behavior and UV-protection property of PET-ZnO nanocomposites prepared by in situpolymerization. Journal of Applied Polymer Science, 114: 1303-1311. https://doi.org/10.1002/app.30614

- He, Q.; Yuan, T.; Zhu, J.; Luo, Z.; Haldolaarachchige, N.; Sun, L.; Guo, Z., 2012: Magnetic high density polyethylene nanocomposites reinforced with in-situ synthesized Fe@ FeO core-shell nanoparticles. Polymer, 53 (16): 3642-3652. https://doi.org/10.1016/j.polymer.2012.06.010
- 16. He, H.; Liu, J.; Wang, Y.; Zhao, Y.; Qin, Y.; Zhu, Z.; Wang, J., 2022: An ultralight self-powered fire alarm etextile based on conductive aerogel fiber with repeatable temperature monitoring performance used in firefighting clothing. ACS Nano, 16 (2): 2953-2967. https://doi. org/10.1021/acsnano.1c10144
- Jarukumjorn, K.; Suppakarn, N., 2009: Effect of glass fiber hybridization on properties of sisal fiber-polypropylene composites. Composites. Part B: Engineering, 40 (7): 623-627. https://doi.org/10.1016/j.compositesb.2009.04.007
- Jouyandeh, M.; Rahmati, N.; Movahedifar, E.; Hadavand, B. S.; Karami, Z.; Ghaffari, M.; Saeb, M. R., 2019: Properties of nano-Fe<sub>3</sub>O<sub>4</sub> incorporated epoxy coatings from Cure Index perspective. Progress in Organic Coatings, 133: 220-228. https://doi.org/10.1016/j.porgcoat.2019.04.034
- Kim, J. K.; Pal, K., 2010: Recent Advances in the Processing of Wood-Plastic Composites. 1<sup>th</sup> ed. Berlin, Springer Science and Business Media, pp. 1-22.
- Klyosov, A. A., 2007: Wood-plastic Composites. 1<sup>th</sup> ed. New Jersey: John Wiley and Sons, pp. 1-49.
- Kozłowski, R.; Władyka-Przybylak, M., 2008: Flammability and fire resistance of composites reinforced by natural fibers. Polymers for Advanced Technologies, 19 (6): 446-453. https://doi.org/10.1002/pat.1135
- Lowden, L. A.; Hull, T. R., 2013: Flammability behaviour of wood and a review of the methods for its reduction. Fire Science Reviews, 2: 1-19. https://doi. org/10.1186/2193-0414-2-4
- Mohan, D.; Pittman Jr, C. U.; Steele, P. H., 2006: Pyrolysis of wood/biomass for bio-oil: a critical review. Energy and Fuel, 20 (3): 848-889. https://doi.org/10.1021/ef0502397
- 24. Özgenç, Ö.; Durmaz, S.; Boyaci, I. H.; Eksi-Kocak, H., 2017: Determination of chemical changes in heat-treated wood using ATR-FTIR and FT Raman spectrometry. Spectrochimica Acta. Part A: Molecular and Biomolecular Spectroscopy, 171: 395-400. https://doi.org/10.1016/j. saa.2016.08.026
- Peters, E. N., 1979: Flame retardant thermoplastics. I. Polyethylene – red phosphorus. Journal of Applied Polymer Science, 24 (6): 1457-1464. https://doi.org/10.1002/ app.1979.070240607
- Ramesh, M.; Rajeshkumar, L.; Sasikala, G.; Balaji, D.; Saravanakumar, A.; Bhuvaneswari, V.; Bhoopathi, R., 2022: A critical review on wood-based polymer composites: processing, properties and prospects. Polymers, 14 (3): 589. https://doi.org/10.3390/polym14030589
- Rasana, N.; Jayanarayanan, K.; Deeraj, B. D. S.; Joseph, K., 2019: The thermal degradation and dynamic mechanical properties modeling of MWCNT/glass fiber multiscale filler reinforced polypropylene composites. Composites Science and Technology, 169: 249-259. https:// doi.org/10.1016/j.compscitech.2018.11.027
- Rowell, R. M., 2012: Handbook of Wood Chemistry and Wood Composites, 2<sup>nd</sup> ed. USA, CRC press, pp. 117-134.
- Smith, P. M.; Wolcott, M. P., 2006: Opportunities for wood/natural fiber-plastic composites in residential and industrial applications. Forest Products Journal, 56 (3): 4-11.

- 30. Sun, Q. F.; Lu, Y.; Xia, Y. Z.; Yang, D. J.; Li, J.; Liu, Y. X., 2012: Flame retardancy of wood treated by TiO<sub>2</sub>/ZnO coating. Surface Engineering, 28 (8): 555-559. https://doi.org/10.1179/1743294412Y.0000000027
- Uyup, M. K. A.; Khadiran, T.; Husain, H.; Salim, S.; Siam, N. A.; Hua, L. S., 2019: Resistance improvement of rubberwood treated with zinc oxide nanoparticles and phenolic resin against white-rot fungi, *Pycnoporus sanguineus*. Maderas. Ciencia y Tecnología, 21 (4): 457-466. http://dx.doi.org/10.4067/S0718-221X2019005000403
- Xian, Y.; Feng, T.; Li, H.; Xing, Z.; Wang, G.; Cheng, H., 2023: The Influence of Nano-CaCO<sub>3</sub> on the Mechanical and Dynamic Thermo-Mechanical Properties of Core – Shell Structured Wood Plastic Composites. Forests, 14 (2): 257. https://doi.org/10.3390/f14020257
- 33. Xu, B.; Ma, W.; Bi, X.; Shao, L.; Qian, L., 2019: Synergistic effects of nano-zinc oxide on improving the flame retardancy of EVA composites with an efficient triazinebased charring agent. Journal of Polymers and the Environment, 27: 1127-1140. https://doi.org/10.1007/s10924-019-01400-7
- 34. Yang, H. S.; Wolcott, M. P.; Kim, H. S.; Kim, H. J., 2005: Thermal properties of lignocellulosic filler-thermoplastic polymer bio-composites. Journal of Thermal Analysis and Calorimetry, 82: 157-160. https://doi.org/10.1007/ s10973-005-0857-5
- 35. Yao, S. S.; Jin, F. L.; Rhee, K. Y.; Hui, D.; Park, S. J., 2018: Recent advances in carbon-fiber-reinforced ther-

moplastic composites: A review. Composites. Part B: Engineering, 142: 241-250. https://doi.org/10.1016/j.compositesb.2017.12.007

- 36. Yin, L.; Gong, K.; Zhou, K.; Qian, X.; Shi, C.; Gui, Z.; Qian, L., 2022: Flame-retardant activity of ternary integrated modified boron nitride nanosheets to epoxy resin. Journal of Colloid and Interface Science, 608: 853-863. https://doi.org/10.1016/j.jcis.2021.10.056
- 37. Zhang, Q.; Ma, L.; Xue, T.; Tian, J.; Fan, W.; Liu, T., 2023: Flame-retardant and thermal-protective polyimide-hydroxyapatite aerogel fiber-based composite textile for firefighting clothing. Composites. Part B: Engineering, 248: 110377. https://doi.org/10.1016/j.compositesb.2022.110377
- Zhao, H.; Li, R. K., 2006: A study on the photo-degradation of zinc oxide (ZnO) filled polypropylene nanocomposites. Polymer, 47 (9): 3207-3217. https://doi. org/10.1016/j.polymer.2006.02.089
- \*\*\*ASTM D2863-19:2019 Standard Test Method for Measuring the Minimum Oxygen Concentration to Support Candle – Like Combustion of Plastics (Oxygen Index), ASTM International: West Conshohocken, PA, USA.
- \*\*\*ISO-4589-1:2017 Plastics Determination of burning behaviour by oxygen index.Part 1: General requirements. International Standards Organization (ISO): Geneva, Switzerland.

#### **Corresponding address:**

#### SEFA DURMAZ

Muğla Sıtkı Koçman University, Kavaklıdere Vocational School, Department of Forestry, Kavaklıdere, TURKEY, e-mail: sefadurmaz@mu.edu.tr

Seyedeh Masoomeh Zamani<sup>1</sup>, Reza Hajihassani<sup>2</sup>, Saman Ghahri<sup>2</sup>

# Bio-Durability and Engineering Characteristics of Heat-Treated Poplar Wood

## Biološka trajnost i tehnička svojstva toplinski modificirane topolovine

#### **ORIGINAL SCIENTIFIC PAPER**

Izvorni znanstveni rad Received – prispjelo: 15. 6. 2022. Accepted – prihvaćeno: 18. 10. 2023. UDK: 630\*84 https://doi.org/10.5552/drvind.2023.0041 © 2023 by the author(s). Licensee University of Zagreb Faculty of Forestry and Wood Technology. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license.

**ABSTRACT** • The aim of this study was to evaluate the effect of brown rot fungus <u>Coniophora puteana</u> activity on physical and mechanical properties as well as biological resistance of heat-treated poplar wood. Two poplar wood species (<u>Populus deltoids</u> and <u>Populus nigra</u>) were heat-treated by thermo-wood (Thermo-D) method. Control and heat-treated specimens were exposed to brown rot fungus <u>C. puteana</u> for 16 weeks. Physical and mechanical characteristics of specimens including density, compression strength parallel to the grain and impact strength were evaluated before and after exposure to fungus. Mass loss of specimens caused by fungal activity (MLF) was also calculated. In addition, the effect of thermal modification on laccase production by <u>C. puteana</u> was assayed. The highest mass loss due to fungal deterioration was observed in control specimens, coinciding with the highest substrate-enzyme interactions and constant decrease in detectable laccase levels. According to the results, thermal modification can be used effectively to protect poplar wood against brown rot fungus attack.

**KEYWORDS:** *poplar; thermo-wood; fungus; physical and mechanical properties; biological resistance* 

**SAŽETAK** • Cilj ovog istraživanja bio je procijeniti utjecaj gljive smeđe truleži <u>Coniophora puteana</u> na fizička i mehanička svojstva te na biološku otpornost toplinski modificirane topolovine. Dvije vrste topolovine (<u>Populus deltoids</u> i <u>Populus nigra</u>) toplinski su modificirane postupkom thermo wood (Thermo D). Kontrolni i toplinski modificirani uzorci bili su izloženi 16 tjedana gljivi smeđe truleži <u>C. puteana</u>. Prije i nakon izlaganja gljivama određena su fizička i mehanička svojstva uzoraka uključujući gustoću, čvrstoću na tlak paralelno s vlakancima i čvrstoću na udarac. Također je izračunan gubitak mase uzoraka kao posljedica aktivnosti gljiva (MLF). Osim toga, ispitan je utjecaj toplinske modifikacije na stvaranje lakaze zbog djelovanja gljive <u>C. puteana</u>. Najveći gubitak mase, tj. najveća razgradnja nastala djelovanjem gljive zabilježena je na kontrolnim uzorcima, a to se podudara s najjačim međusobnim djelovanjem supstrata i enzima te s konstantnim smanjenjem detektirane razine lakaze. Prema dobivenim rezultatima, toplinska se modifikacija može učinkovito iskoristiti za zaštitu topolovine od napada gljiva smeđe truleži.

KLJUČNE RIJEČI: topolovina; thermo-wood; gljive; fizička i mehanička svojstva; biološka otpornost

<sup>&</sup>lt;sup>1</sup> Author is assistant professor at Research Institute of Forests and Rangelands, Agricultural Research Education and Extension Organization (AREEO), Tehran, Iran.

<sup>&</sup>lt;sup>2</sup> Authors are assistant professors at Wood and Forest Products Science Research Division, Research Institute of Forests and Rangelands, Agricultural Research Education and Extension Organization, Tehran, Iran.

#### **1 INTRODUCTION**

#### 1. UVOD

Wood as a natural polymer has been widely used in structures for many years. However, some properties of this natural material such as moisture absorption, dimensional instability, biological degradation, weathering, etc., limit its use. So, numerous new methods of wood modification have been developed to remove these disadvantages and expand its functionality. Wood heat treatment is one of the modification methods, which is carried out by different processes. This kind of modification was first studied scientifically by Stamm and Hansen in the 1930s in Germany and in the 1940s by White in the United States. The most comprehensive research in this field has been carried out by VTT (International Thermo-Wood Association) in Finland, whose product is called thermo-wood. A temperature of 160-260 °C is usually used to produce thermowood (Militz, 2002). Thermal modifications have some positive and negative effects on wood. In some cases, thermal treatments lead to positive changes in chemical structure of wood, colour, dimensional stability, thermal insulation properties, biological resistance to biological degradation. However, these treatments also reduce some mechanical properties such as wood bending strength.

Heat treatment of wood in the temperature range of 140-260 °C with a long holding time causes an irreversible reduction of moisture absorption (González-Peña *et al.*, 2004; Obataya and Tomita, 2002); it leads to dimensional stability (Krause *et al.*, 2004), less moisture movement (Militz and Tjeerdsma, 2001), and improved biological resistance (Welzbacher and Rapp, 2004; Farahani *et al.*, 2001). However, one of the major drawbacks of these treatments is the reduction of wood mechanical properties (González-Peña and Hale, 2007).

Brown rot fungi have an enzyme system that produces hydrolytic and oxidative enzymes, which act on cell wall component degradation such as lignin, cellulose and hemicelluloses (Baldrian and Gabriel, 2003). Cellulases can be classified into three types: endoglucanases, exoglucanases and 1, 4- $\beta$ -glucosidases (Gielkens *et al.*, 1999). Lignin peroxidase (LiP), manganese peroxidase (MnP) and laccase (Lac) are the most widely distributed lignin-degrading enzymes, which are responsible for the biodegradation of lignin (Vicuna, 2000).

Evaluation of heat treatment effect on bio-durability of a wood species showed that it significantly increases wood durability against fungal action due to the reduction of hydroxyl groups (Mburu *et al.*, 2007). Kamperidou (2019) evaluated the biological durability of thermo-chemically modified pine and spruce wood against white and brown rot fungi. The results showed that thermal modification presented lower mass loss caused by used fungi compared to the unmodified wood. In other words, heat treatment increases the biological resistance of wood.

According to Gao *et al.* (2016), the colour of wood became darker and there was an improvement in dimensional stability as well as reduction in modulus of elasticity of heat-treated poplar wood at temperatures of 140 to 200 °C for 1-3 hours. In addition, biodurability of heat-treated specimen was better against white rot fungus than brown rot fungus.

Furthermore, Corleto *et al.* (2020) investigated the effect of thermal modification on properties of padauk wood, indicating that heat treatment at high temperature caused significant loss in bending strength and bending stiffness of padauk wood. Moreover, the results of chemical analysis showed that cellulose and lignin proportion increased, while that of hemicellulose decreased substantially following thermal modification. Another study showed that durability and dimensional stability of European birch, European aspen, Norway spruce as well as Scots pine improved by thermal modification (Karlsson *et al.*, 2011; Militz and Altgen, 2014).

Kaygin *et al.* (2009) investigated the effect of mass loss of mechanical properties of heat-treated Paulownia at 160, 180 and 200 °C. The results showed that mechanical properties including compression strength, modulus of elasticity, bending strength and impact strength decreased with increasing of heat treatment temperature. In fact, it was determined that, due to thermal modification, the mass loss significantly affected mechanic properties of Paulownia wood.

Heat treatment can be used effectively against fungal attack for Scotch pine, oak and beech wood species (Ayata *et al.*, 2017). At high temperature of wood thermal modification, the lignin content increases because of autocondensation and its higher thermal stability. There is a slight decrease in the extractives content. Moreover, the cellulose and its DP decrease, while cross-linking reactions occur in the cellulose (**Čabalová** *et al.*, 2019).

Brown rot fungi have the ability to degrade cellulose by producing extracellular fungal cellulose degrading enzymes. The extracellular enzymes produced by these fungi are greatly affected by the type of substrate being decayed. Our study focuses on measuring oxidative enzyme laccases secreted by *C. puteana* in the presence of poplar woods (control and heat-treated specimens) as substrates. Therefore, this study aimed to find the effect of heat treatment on biological resistance of thermo-wood against degradation by brown rot fungus *Coniophora puteana*. Meanwhile, physical and mechanical properties of thermo-wood and control specimen after exposure to *C. puteana* were investigated. In addition, levels of laccase enzyme on both substrates (control and modified specimens) were assayed.

#### 2 MATERIALS AND METHODS

#### 2. MATERIJALI I METODE

The lumbers of two poplar wood species (*Populus deltoids* and *Populus nigra*) were prepared and treated in Mazand wood company. After kiln loading, the temperature was rapidly raised to a level of around 100 °C and then increased steadily to 130 °C. Thereafter, the temperature inside the kiln was increased to around 212 °C and remained constant for 3 hours. The final stage was cooling and moisture conditioning by using water spray system. When the temperature reached 80 °C, re-moisturizing took place to bring the wood moisture content to a level of around 6 %. In order to evaluate biological resistance as well as physical and mechanical properties of treated timbers, six specimens were prepared in accordance with EN 113-2:2020 standard.

In this study, brown rot fungus Coniophora puteana was used to evaluate biological resistance of control and modified specimens. Modified and control specimens were exposed to fungal mycelium in Kolleflasks. As a nutrient medium for fungal mycelium, Malta Extract Agar (Merck KGaA, Darmstadt, Germany) was used. After preparing nutrient medium and sterilization in autoclave (temperature 120 °C, time 20 minutes, pressure 1.5 kg/cm<sup>2</sup>), small pieces of fungal mycelium were inoculated in Kolle flasks. Then the flasks with nutrient medium and mycelium were incubated in the climate chamber at a temperature of 22 °C and 65±5 % relative humidity for 2 to 3 weeks until the entire surface of the nutrient medium was overgrown by mycelium (as shown in Figure 1). Six modified and control specimens were placed in each flask and were incubated in the same climate chamber for another 16 weeks according to EN 113-2: (2020) standard (Figure 1). After 16 weeks, all Kolle flasks were taken out from the climate chamber.

Ligninolytic enzyme laccase was extracted at 25 °C in two extraction cycles, 5 and 24 hours, with 50 mM sodium acetate buffer (pH=5.5) with 0.1 g/L Polysorbate 20 (Tween<sup>®</sup> 20; from Sigma Aldrich: Steinheim, Germany). It should be noted that two extraction cycles of 5 and 24 hours were used to obtain the optimal time for maximum enzyme extraction. An amount of 1.5 gr chips prepared from decayed specimens were first soaked in 50 mL extraction buffer for 5 hours. The second extraction was performed with 25 mL extraction buffer for 24 hours. For this purpose, six repetitions were considered. Then, the supernatants collected from all extractions were filtered, centrifuged at 5.000 rpm for 15 minutes and used to test enzyme activity.

Laccase activity was assayed by measuring the oxidation of 2,6-DMP. An aliquot of enzyme solution was incubated in 1 ml of 5 mM of 2,6-Dimethoxyphenol (2,6-DMP) in 0.1 M sodium acetate buffer (pH= 3.6) at 30 °C. Absorbance was monitored at 469 nm in a spectrophotometer. One unit of laccase activity was defined as the amount of the laccase that oxidized 1  $\mu$ mol of 2,6-DMP per minute at 30 °C (Field *et al.*, 1993).

In order to evaluate the durability of wood specimens against brown rot fungus, the mass loss was calculated. After 16 weeks of incubation, the wood specimens were taken out of the Kolle flasks and carefully cleaned from the mycelium and then oven dried for 24 hours at the temperature of  $103\pm2$  °C until the constant mass was reached, and then weighed. The mass loss



Figure 1 Control and heat-treated specimens after exposure to brown rot fungi (Up: after 3 weeks, Down: after 16 weeks)
Slika 1. Kontrolni i toplinski modificirani uzorci nakon izlaganja gljivama smeđe truleži (gore: nakon tri tjedna, dolje: nakon 16 tjedana)

and the repellent effect of thermal modification on *MLF* of each specimen were calculated according to equation 1 and 2, respectively:

$$L = ((D_1 - D_2)/D_1) \times 100$$
 (1)

Where:

M.L – mass loss (%)

M

- $D_1$  oven dry mass of specimen before exposure to fungi (g)
- $D_2$  oven dry mass of specimen after exposure to fungi (g)

$$R.E = ((M.L_1 - M.L_2)/M.L_1) \times 100$$
(2)

Where:

- *R.E* repellent effect of thermal modification on *MLF* (%)
- $M.L_1$  mass loss of control specimen caused by fungi (%)  $M.L_2$  – mass loss of heat-treated specimen caused by
- fungi (%)

In this research, only the properties that could be determined based on the specimens that can be placed in Kolle flasks, based on standard dimensions, were considered. The physical and mechanical properties of control and heat-treated specimens were tested according to ASTM D 143-09. Also, the standard ASTM D256 was used to measure impact strength, because the dimensions of the specimens in this standard were suitable to be placed in the Kolle flask. Moreover, the standard EN 113 was used to evaluate bio-durability. It should be noted that 6 replications were considered for each parameter. The obtained data from the tests were statistically analyzed based on One-Way ANOVA method. This method was used to determine any statistically significant differences between the means of the two groups (control and modified specimens) for each wood species separately.

### 3 RESULTS AND DISCUSSION3. REZULTATI I RASPRAVA

In this research, the physical and mechanical properties of two wood species (*P. deltoids* and *P. ni*-

*gra*) were evaluated in two stages: the first after thermal modification and the second after fungal degradation to show the effect of heat treatment on fungal functionality as well as physical and mechanical properties. Table 1 summarizes the results of statistical analysis of One-Way ANOVA test for physical and mechanical properties of two poplar wood species, specifying significant levels. The numbers presented in Table 1 show whether the heat treatment had a significant effect on the investigated properties.

### **3.1. Laccase activity assessment 3.1. Procjena pojave lakaze**

The results of laccase evaluation of specimens after 16 weeks of exposure to C. puteana showed that the laccase value in controls was lower than in heattreated specimens in both poplar wood species (Figure 2). In fact, the reduction of this enzyme in controls compared to thermo-wood indicates greater consumption and effectiveness of this enzyme in untreated specimens. As heat treatment reduces the amount of hemicellulose and holocellulose, hydroxyl groups, which are an active site for chemical reactions, are reduced. As a result, the activity and effectiveness of laccase enzyme produced by brown rot fungi in modified wood is reduced (Gaff et al., 2019; Mburu et al., 2007; Wentzel et al., 2019). Faraz et al. (2003) reported that, when Eucalyptus grandis wood chips were exposed to Ceriporiopsis subvermisporaon, the greatest mass loss was directly associated with a significant reduction of laccase enzyme, which supports the results of this study.

### 3.2. Mass loss caused by fungi (*MLF*) 3.2. Gubitak mase zbog djelovanja gljiva (*MLF*)

The results revealed that heat treatment had a significant effect on the decay action of brown rot fungi, leading to mass loss (Table 1). The *MLF* in thermowood specimens were significantly lower than in control specimens. This indicates the inhibition of heat

		Afte Nako	r thermal modific	r <b>ation</b> kacije	After fungal degradation Nakon razgradnje uzrokovane gljivama				
S.V	<b>Wood</b> species Vrste drva	Density Gustoća	Compression strength parallel to grain Čvrstoća na tlak paralelno s vlakancima	<b>Impact</b> strength Čvrstoća na udarac	<b>Mass loss</b> Gubitak mase	<b>Density</b> Gustoća	Compression strength parallel to grain Čvrstoća na tlak paralelno s vlakancima	Impact strength Čvrstoća na udarac	
Heat treatment	Populus deltoids	0.008**	0.017*	0.025*	0.020*	0.284 ns	0.122 ns	0.016*	
modifikacija	Populus nigra	0.024*	0.231 ns	0.028*	0.001**	0.386 ns	0.200 ns	0.034*	

 Table 1
 Summarized statistical results of One-Way ANOVA for physical and mechanical properties of treated poplar wood

 Tablica 1.
 Sažeti statistički rezultati jednosmjerne ANOVA-e za fizička i mehanička svojstva toplinski modificirane topolovine

ns, \* and \*\* - Non-significant and significant at 5 % and 1 % probability levels, respectively / nije značajno i značajno je pri razinama vjerojatnosti od 5 %, odnosno 1 %



Figure 2 Laccase activity in heat-treated and control wood specimens during decay by Coniophora puteana Slika 2. Pojava lakaze u toplinski modificiranim i kontrolnim uzorcima drva tijekom truljenja uzrokovanoga gljivom Coniophora puteana



Figure 3 Effect of thermal modification on MLF in two wood species

Slika 3. Utjecaj toplinske modifikacije na MLF za dvije vrste drva

treatment on fungal activity (Figure 3). As indicated in Figure 3, the repellent effect of thermal modification on MLF obtained 72.28 and 82.69 % in P. deltoids and P. nigra, respectively. On the other hand, as shown in Figure 2, laccase activity in heat-treated specimens is greater than in controls, which is a sign of thermowood resistance against fungal degradation. Considering the substrate-enzyme interactions, it seems that in the control specimens, laccase enzyme had more interaction with the substrate and therefore the mass in these specimens was reduced to a greater extent compared to the treated specimens. According to the results of this study, there are other reports that show the deep adsorption of laccases in substrates suitable for their degradation (Tu et al., 2009). Moreover, improvement of bio-durability of thermo-wood against fungal degradation is related to the reduction of hydroxyl groups due to heat treatment (Mburu et al., 2007).

#### 3.3. Density

3.3. Gustoća

This physical property was evaluated as follows.

#### 3.3.1 After thermal modification

#### 3.3.1. Nakon toplinske modifikacije

The results showed that heat treatment had a significant effect on the density (Table 1), so that the thermo-wood process reduced the density of both wood species (Figure 4). The reduction of density due to heat treatment in P. deltoids and P. nigra was 8.81 and 5.17 %, respectively. The change of wood chemical structure at high temperature causes mass loss, which leads to density decrease. The reduction of density due to heat treatment is related to destruction and change of chemical structure of wood, which intensifies at high temperatures (Militz, 2002).

#### 3.3.2 After fungal degradation 3.3.2. Nakon razgradnje gljivama

After density assessment of control and heat-treated specimens in section 3.3.1, these specimens were exposed to brown rot fungus C. puteana. The results

Table 2 Average reduction values of physical and mechanical properties due to fungal degradation Tablica 2. Prosječne redukcijske vrijednosti fizičkih i mehaničkih svojstava zbog razgradnje gljivama

<b>Property</b> Svojstvo	Density / (	Gustoća, %	Compressi parallel to Čvrstoća na t s vlakan	on strength o grain, % tlak paralelno ccima, %	<b>Impact st</b> Čvrstoća na	<b>rength, %</b> a udarac, %
Wood species / vrste drva	P. deltoids	P. nigra	P. deltoids	P. nigra	P. deltoids	P. nigra
Control / kontrolni uzorak	3.27	5.94	6.04	11.71	4.71	7.45
Thermo-wood / toplinski tretirano drvo	2.23	5.52	3.69	9.28	6.41	21.88



Wood species / vrste drva

**Figure 4** Effect of thermal modification on wood density and its variation under fungal attack **Slika 4.** Utjecaj toplinske modifikacije na gustoću drva i promjene gustoće zbog napada gljiva

showed that heat treatment had no significant effect on fungal functionality in reduction of density in both poplar wood species (Table 1). As shown in Figure 4, the density difference between the control and thermo-wood specimens is little in both poplar species. However, heat treatment has been able to reduce the activity of brown rot fungus in both wood species, so that the density reduction due to fungal action in thermo-wood specimen was less than in controls (Table 2). This indicates that heat treatment has a repellent effect on fungal activity.

### **3.4. Compression strength parallel to grain** 3.4. Čvrstoća na tlak paralelno s vlakancima

This mechanical property was evaluated as follows.



The results showed that thermal modification had no significant effect on the compression strength parallel to the grain of *P. nigra* but it did in *P. deltoids* (Table 1). However, as indicated in Figure 5, heat treatment has improved this mechanical property in both wood species, so that the value of compression strength parallel to the grain in *P. deltoids* and *P. nigra* was 27.46 and 11.21 %, respectively (Figure 5). As a matter of fact, wood chemical structure changes at high temperature during thermal modification. Consequently, heat treatment causes cross-linking of lignin as well as increasing of lignin and cellulose values in wood structure, which leads to the improvement of compression



**Figure 5** Effect of thermal modification on compression strength parallel to grain and its variation under fungal attack **Slika 5.** Utjecaj toplinske modifikacije na čvrstoću na tlak paralelno s vlakancima i promjene čvrstoće zbog napada gljiva



**Figure 6** Effect of thermal modification on impact strength and its variation under fungal attack **Slika 6.** Utjecaj toplinske modifikacije na čvrstoću na udarac i promjene te čvrstoće zbog napada gljiva

strength parallel to the grain. In other words, the change of wood chemical components due to the heat treatment is an effective agent for changing physical and mechanical properties of wood (Gaff *et al.*, 2019; Wentzel *et al.*, 2019).

### 3.4.2 After fungal degradation

3.4.2. Nakon razgradnje gljivama

The results revealed that heat treatment had no significant effect on fungal efficiency in terms of compression strength parallel to the grain (Table 1). However, comparison of the results in Figure 5 shows that this kind of modification was able to reduce the activity of brown rot fungi in both wood species. Consequently, the reduction of compression strength parallel to the grain due to fungal activity in thermo-wood specimen was less than that in the control, which indicates a repellent effect of heat treatment on fungal activity (Table 2). As shown in Figure 2, laccase activity in heattreated specimens was more than that in the control group, which indicates bio-durability of thermo-wood. Heat treatment of wood at high temperatures reduces hydroxyl groups, which leads to the reduction of enzymatic hydrolysis (Mburu et al., 2007).

#### 3.5. Impact strength

3.5. Čvrstoća na udarac

This mechanical property was evaluated as follows:

#### 3.5.1 After thermal modification

#### 3.5.1. Nakon toplinske modifikacije

The results revealed that heat treatment had a significant effect on impact strength of both wood species (Table 1). Thermal modification had negative effect on this mechanical property in both species, as shown in Figure 5. The reduction of impact strength was 8.24 and 31.91 % in *P. deltoids* and *P. nigra*, respectively (Figure 6). The reduction of impact strength of thermowood might be related to brittle structure of heat-treated specimens. In fact, the reduction of this mechanical property is directly related to chemical structure changes, while mass loss of specimens occurs due to heating (Kaygin *et al.*, 2009).

#### 3.5.2 After fungal degradation

#### 3.5.2. Nakon razgradnje gljivama

The results showed that heat treatment had a significant effect on fungal action in terms of impact strength (Table 1). Although thermal modification can reduce the activity of the brown rot fungi, the reduction of impact strength in thermo-wood exposed to brown rot fungi was greater than in control group (Table 2). The effect of heat treatment on structural changes of wood as well as destruction of cellulose and holocellulose by brown rot fungi can be effective agents in the reduction of impact strength of thermo-wood specimens.

#### 4 CONCLUSIONS

#### 4. ZAKLJUČAK

The results of the present study revealed that the degradation of two poplar wood species (*P. deltoids* and *P. nigra*) by *C. puteana* could be affected by thermal modification. Although the activity of *C. puteana* and laccase production occurred in both control and heat-treated specimens, the lowest detectable laccase levels were observed in controls, coinciding with the highest mass loss. Considering the substrate-enzyme interactions, it seems that in the control specimens, laccase enzyme had more interaction with the substrate and therefore the mass in these specimens was reduced to a greater extent compared to the treated specimens. Therefore, the result of laccase activity assessment also showed that the laccase value in controls was less than

that in modified specimens. Overall, the results revealed that thermal modification improves bio-durability of both poplar wood species; however, it reduces some mechanical properties. On the other hand, this kind of modification prevents the reduction of mechanical properties due to biological degradation.

#### Acknowledgements - Zahvala

Authors would like to express their special thanks to Research Institute of Forests and Rangelands, Agricultural Research, Education and Extension Organization (AREEO) and Mazand Choob Aria Company for their support in this research.

#### **5 REFERENCES**

#### 5. LITERATURA

- Ayata, U.; Akcay, C.; Esteves, B., 2017: Determination of decay resistance against Pleurotus ostreatus and *Coniophora puteana* fungus of heat-treated scotch pine, oak and beech wood species. Maderas, Ciencia y Tecnología, 19 (3): 309-316. http://dx.doi.org/10.4067/S0718-221X2017005000026
- Baldrian, P.; Gabriel, J., 2003: Lignocellulose degradation by Pleurotus ostreatus in the presence of cadmium. FEMS Microbiology Letters, 220 (2): 235-240. https:// doi.org/10.1016/S0378-1097(03)00102-2
- Čabalová, I.; Zachar, M.; Kačík, F.; Tribulová, T., 2019: Impact of thermal loading on selected chemical and morphological properties of spruce Thermo Wood. BioResources, 14 (1): 387-400. https://doi.org/10.15376/biores.14.1.387-400
- Corleto, R.; Gaff, M.; Niemz, P.; Sethy, A. K.; Todaro, L.; Ditommaso, G.; Kamboj, G., 2020: Effect of thermal modification on properties and milling behaviour of African padauk (*Pterocarpus soyauxii* Taub.) wood. Journal of Materials Research and Technology, 9 (4): 9315-9327. https://doi.org/10.1016/j.jmrt.2020.06.018
- Farahani, M. R. M.; Hill, C. A. S.; Hale, M. D. C., 2001: The effect of heat treatment on the decay resistance of corsican pine sapwood. In: Proceedings 5<sup>th</sup> European Panel Products Symposium, pp. 303-308.
- Ferraz, A.; Córdova, A. M.; Machuca, A., 2003: Wood biodegradation and enzyme production by *Ceriporiopsis* subvermispora. Enzyme and Microbial Technology, 32 (1): 59-65. https://doi.org/10.1016/S0141-0229(02)00267-3
- Field, J. A.; Jong, E.; Feijoo-Costa, G.; Bont, J. A. M., 1993: Screening for ligninolytic fungi applicable to the biodegradation of xenobiotics. Trends in Biotechnology, 11: 44-49. https://doi.org/10.1016/0167-7799(93)90121-O
- Gaff, M.; Babiak, M.; Kačík, F.; Sandberg, D.; Turčani, M.; Hanzlík, P.; Vondrová, V., 2019: Plasticity properties of thermally modified timber in bending – the effect of chemical changes during modification of European oak and Norway spruce. Composites. Part B: Engineering, 165 (5): 613-625. https://doi.org/10.1016/j.compositesb.2019.02.019
- Gao, H.; Sun, M. Y.; Cheng, H. Y.; Gao, W. L.; Ding, X. L., 2016: Effects of heat treatment under vacuum on properties of poplar. BioResources, 11 (1): 1031-1043. https://doi.org/10.15376/biores.11.1.1031-1043

- Gielkens, M. M. C.; Dekkers, E.; Visser, J.; Graaff, L. H., 1999: Two cellubiohydrolase-encoding genes from *Aspergillus niger* require Dxylose and the xylanolytic transcriptional activator XlnR for their expression. Applied and Environmental Microbiology, 65 (10): 4340-4345. https://doi.org/10.1128/AEM.65.10.4340-4345.1999
- González-Peña, M. M.; Hale, M. D. C., 2007: The relationship between mechanical performance and chemical changes in thermally modified wood. In: Proceedings of 3<sup>rd</sup> European Conference on Wood Modification, pp. 169-172.
- González-Peña, M. M.; Breese, M. C.; Hill, C. A. S., 2004: Hygroscopicity in heat-treated wood: effect of extractives. In: Proceedings of 1<sup>st</sup> International Conference on Environmentally – Compatible Forest Products, pp. 105-119.
- Kamperidou, V., 2019: The biological durability of thermally-and chemically modified black pine and poplar wood against basidiomycetes and mold action. Forests, 10 (12): 1111-1128. https://doi.org/10.3390/f10121111
- Karlsson, O.; Sidorava, E.; Moren, T., 2011: Influence of heat transferring media on durability of thermally modified wood. BioResources, 6 (1): 356-372. https://doi. org/10.15376/biores.6.1.356-372
- Kaygin, B.; Gunduz, G.; Aydemir, D., 2009: The effect of mass loss on mechanical properties of heat-treated Paulownia wood. Wood Research, 54 (2): 101-108.
- Krause, A.; Hof, C.; Militz, H., 2004: Novel wood modification processes for window and cladding products. In: Proceedings of 35<sup>th</sup> Annual Meeting, International Research Group on Wood Protection, IRG/WP 04-40285.
- Mburu, F.; Dumarc, S.; Huber, F.; Petrissans, M.; Gérardin, P., 2007: Evaluation of thermally modified grevillea robusta heartwood as an alternative to shortage of wood resource in Kenya. Characterisation of physicochemical properties and improvement of bio-resistance. Bioresource Technology, 98 (18): 3478-3486. https://doi. org/10.1016/j.biortech.2006.11.006
- Militz, H.; Altgen, M., 2014: Processes and properties of thermally modified wood manufactured in Europe. American Chemical Society in Deterioration and Protection of Sustainable Biomaterials, ACS Symposium Series; American Chemical Society: Washington, DC, pp. 269-285. https://doi.org/10.1021/bk-2014-1158.ch016
- Militz, H.; Tjeerdsma, B., 2001: Heat treatment of wood by the PLATO process. In: Review on Heat Treatments of Wood. COST ACTION E22: Environmental Optimisation of Wood Protection, pp. 27-38.
- Militz, H., 2002: Thermal treatment of wood. European processes and their background. In: Proceedings of 33<sup>rd</sup> Annual Meeting IRG/WP 02-40241, 12-17 May, Cardiff-Wales, 4: 1-17.
- Obataya, E.; Tomita, B., 2002: Hygroscopicity of heattreated wood II. Reversible and irreversible reductions in the hygroscopicity of wood due to heating. Mokuzai Gakkaishi, 48 (4): 288-295.
- Tu, M.; Pan, X.; Saddler, J. N., 2009: Adsorption of cellulase on cellulolytic enzyme lignin from lodgepole pine. Journal of Agricultural and Food Chemistry, 57 (17): 7771-7778. https://doi.org/10.1021/jf901031m
- Vicuna, R., 2000: Ligninolysis: a very peculiar microbial process. Molecular Biotechnology, 14: 173-176. https:// doi.org/10.1385/MB:14:2:173
- Welzbacher, C. R.; Rapp, A. O., 2004: Determination of the water sorption properties and preliminary results from field tests above ground of thermally modified ma-

terial from industrial scale processes. In: Proceedings of 35<sup>th</sup> Annual Meeting, International Research Group on Wood Protection, IRG/WP 04-40279.

- Wentzel, M.; Fleckenstein, M.; Hofmann, T.; Militz, H., 2019: Relation of chemical and mechanical properties of Eucalyptus nitens wood thermally modified in open and closed systems. Wood Material Science & Engineering, 14 (3): 165-173. https://doi.org/10.1080/17480272.2018. 1450783
- 26. \*\*\*EN 113:1997 Wood preservatives. Test method for determining the protective effectiveness against wood destroying basidiomycetes.
- 27. \*\*\* ASTM D 143-09:2014 Standard methods of testing small clear specimens of timber.
- 28. \*\*\* ASTM D 256:2018 Standard test methods for determining the Izod pendulum impact strength of plastics.

#### **Corresponding address:**

#### **REZA HAJIHASSANI**

Wood and Forest Products Science Research Division, Research Institute of Forests and Rangelands, Agricultural Research Education and Extension Organization (AREEO), Tehran, IRAN, e-mail: Reza.hajihassani@gmail.com

Seyyed Khalil Hosseinihashemi<sup>1</sup>, Mohammad-Ali Akhoundi<sup>1</sup>, Younes Shirmohammadli<sup>2</sup>, Nadir Ayrilmis<sup>3</sup>

## Influence of Pre-Impregnation Acetic Anhydride and Heating Time on Mechanical and Physical Properties of Wood-Plastic Composites

Utjecaj predimpregnacije anhidridom octene kiseline i vremena zagrijavanja na mehanička i fizička svojstva drvno-plastičnih kompozita

#### **ORIGINAL SCIENTIFIC PAPER**

**Izvorni znanstveni rad** Received – prispjelo: 26. 5. 2023. Accepted – prihvaćeno: 13. 11. 2023. UDK: 630\*84; 674.816 https://doi.org/10.5552/drvind.2023.0117 © 2023 by the author(s). Licensee University of Zagreb Faculty of Forestry and Wood Technology. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license.

**ABSTRACT** • This research aimed to assess the influence of different pre-impregnation times (PITs) (60, 180, and 300 min), heating or reaction times (H/RTs) (60, 90, and 120 min), and chemical modification of wood flour (WF) on the mechanical and physical properties of wood-plastic composites (WPCs). The study employed acetylated beech (Fagus orientalis L.) flour as the filler and polypropylene (PP) as the matrix phase producing of WPC samples through melt compounding and injection molding. The resulting composites underwent testing for their physical and mechanical properties. The findings revealed that WPCs derived from acetylated wood with <u>PITs</u> of 60 min and H/RTs of 60 min exhibited the highest mechanical properties, except for the bending modulus. Moreover, the lowest water absorption (WA) was observed in the PITs-H/RTs combination of 60-120 min, while the lowest thickness swelling (TS) occurred in the PITs-H/RTs combination of 300-60 min. The simultaneous utilization of pre-impregnation and reaction times demonstrated a synergistic effect on the physical and mechanical properties. Consequently, the chemical modification of wood flour and the application of suitable reaction times improved the interfacial adhesion, thereby enhancing the overall performance of the WPCs.

**KEYWORDS:** WPCs; wood flour; acetylation; chemical modification; mechanical properties

**SAŽETAK** • Cilj ovog istraživanja bio je procijeniti utjecaj različitih vremena predimpregnacije (PIT) (60, 180 i 300 min), vremena zagrijavanja (H/RT) (60, 90 i 120 min) i kemijske modifikacije drvnog brašna (WF) na mehanička i fizička svojstva drvno-plastičnih kompozita (WPC). U istraživanju je za proizvodnju uzoraka WPC-a taljenjem i injekcijskim prešanjem upotrijebljeno drvno brašno acetilirane bukovine (<u>Fagus orientalis</u> L.) kao punilo i polipropilen (PP) kao matrica. Zatim su ispitana fizička i mehanička svojstva dobivenih kompozita. Rezultati su

<sup>&</sup>lt;sup>1</sup> Authors are associate professor and MSc at Department of Wood Science and Paper Technology, Karaj Branch, Islamic Azad University, Karaj, Iran. https://orcid. org/0000-0001-6236-0376; https://orcid.org/0009-0009-4485-6170

<sup>&</sup>lt;sup>2</sup> Author is PhD at Department of Civil and Environmental Engineering, Faculty of Engineering, The University of Auckland, New Zealand. https://orcid.org/0000-0002-4695-1368

<sup>&</sup>lt;sup>3</sup> Author is professor at Department of Wood Mechanics and Technology, Forestry Faculty, Istanbul University-Cerrahpasa, Bahcekoy, Sariyer, 34473, Istanbul, Turkey. https://orcid.org/0000-0002-9991-4800

pokazali da drvno-plastični kompoziti dobiveni od acetiliranog drva s vremenom predimpregnacije i zagrijavanja od 60 min imaju najbolja mehanička svojstva, osim modula na savijanje. Nadalje, najmanje upijanje vode (WA) zabilježeno je za kombinaciju PIT-H/RT 60-120 min, dok je najmanje debljinsko bubrenje (TS) izmjereno za kombinaciju PIT-H/RT 300-60 min. Istodobno je primjena predimpregnacije i zagrijavanja pokazala sinergijski utjecaj na fizička i mehanička svojstva. Posljedično su kemijska modifikacija drvnog brašna i odabir odgovarajućih vremena zagrijavanja poboljšali adheziju između faza, a time i svojstva drvno-plastičnih kompozita.

KLJUČNE RIJEČI: WPC; drvno brašno; acetilacija; kemijska modifikacija; mehanička svojstva

#### **1 INTRODUCTION**

#### 1. UVOD

Chemical modification is considered an active approach as the transformation occurs in the macromolecules of the wood cell wall, utilizing one or more chemicals that react to form a material integrated into the wood cell wall structure (Rowell, 2005). This alteration of the cell wall macromolecules in solid wood leads to a reduction in dimensional stability and moisture content, while enhancing resistance to biological degradation (Hill, 2006; Lyon *et al.*, 2008; Ermeydan, 2014; Thybring and Fredriksson, 2021; Zhou and Liu, 2022). A prominent challenge arises when hydrophilic lignocellulosic fibers are used as reinforcements in a hydrophobic synthetic matrix, as the lack of compatibility between the two materials results in disparate behavior and poor adhesion (Özmen *et al.*, 2013a).

The properties and performance of wood and wood-based composites were improved by chemical modification, which is defined as chemical reaction between some reactive site of a lignocellulose material and a simple single chemical reagent, with or without catalyst, to form a covalent bond between them. Acetylation is a reaction that leads to a change in the chemical and physical properties of the wood substrate and results in the substitution of hydrophilic hydroxyl groups with hydrophobic acetyl groups (Rowell, 2006a, 2006b; Rowell, 2007; Bodîrlău *et al.*, 2012).

The poor compatibility between the reinforcing material and the matrix contributes to a decline in the mechanical properties of the composite (Özmen *et al.*, 2013b; Kallakas *et al.*, 2015). To address this issue and enhance the interfacial adhesion between the polymer matrix and wood flour (WF), various chemical modifications such as alkaline treatment, acetylation, epoxidation, benzoylation, silane treatment, or other WF modifications have been employed (Rowell, 2006a; Farsi, 2010; Gwon *et al.*, 2010; Tesinova, 2011; Teacă *et al.*, 2014; Kallakas *et al.*, 2015; de Prá Andrade and Poletto, 2021).

Chemical modification of *Abies alba* L. softwood samples was conducted by succinic anhydride (SA) and maleic anhydride (MA) reagents in the presence of xylene as solvent (reaction time 1 h and 90 °C), where MA exhibited lower reactivity towards wood than SA, presumably due to chemical structure (Teacă *et al.*, 2014).

The conventional acetylation process involves the impregnation of dried wood with liquid-phase acetic anhydride, followed by the application of external heat (Hung *et al.*, 2016). Acetylation of wood is a widely used method for modifying wood with acetic anhydride, which leads to the esterification of accessible hydroxyl groups in the cell wall with acetyl groups (Figure 1).

Larsson-brelid *et al.* (2006) conducted a study on WPCs using acetic anhydride-modified solid wood. Previous research has demonstrated that WPC materials reinforced with acetylated wood fibers exhibit improved mechanical properties, reduced moisture content, and enhanced resistance to brown-rot decay compared to unmodified WPC materials (Abdul Khalil *et al.*, 2002; Segerholm *et al.*, 2005).

The properties of acetylated wood are influenced by the specific method of acetylation. Factors such as reaction temperature, reaction time, type of catalysts, and their quantities significantly affect the extent of fiber degradation during the treatment process (Rowell, 1983). The weight percent gain (*WPG*) increases as the reaction temperature and time increase. For instance, in the chemical modification of pine wood using various chemicals, such as liquid acetic anhydride, the *WPG* of acetylated pine wood increases with higher temperature values (from 100 °C to 160 °C) and longer reaction times (from 60 min to 300 min) (Rowell, 1983).

This study aimed to investigate the mechanical and physical properties of polypropylene composites (WPCs) reinforced with acetylated and unmodified beech WF. Three different pre-impregnation times (*PITs*) and three different heating or reaction times (*H*/RTs) were evaluated during the acetylation process using acetic anhydride. The obtained results were compared with those of unmodified WPCs.



**Figure 1** Anhydride modification scheme, where  $R = CH_3$  (acetic anhydride) (Papadopoulos *et al.*, 2019) **Slika 1.** Shema modifikacije anhidrida, pri čemu je  $R = CH_3$  (anhidrid octene kiseline) (Papadopoulos *et al.*, 2019.)

#### 2 MATERIALS AND METHODS

2. MATERIJALI I METODE

#### 2.1 Materials

#### 2.1. Materijali

Beech wood (*Fagus orientalis* L.) was sourced from the Kelardasht forest in northern Iran. The wood was initially cut into small pieces and then further reduced in size using a hammer mill. The resulting WF had a particle size ranging from 0.25 mm to 0.42 mm. Before impregnation and reaction processes, the wood samples were subjected to oven drying at (103±2) °C for 24 hours. Subsequently, the samples were allowed to cool to room temperature in a desiccator containing phosphorous pentoxide before their weight was determined. The moisture content of WF before reaction process was 0 %. Ovendried WF was stored in sealed plastic bags.

Polypropylene (PP) with a melt flow index (MI) of 10 g/10 min and a density of 0.95 g/cm<sup>3</sup> was supplied by the Tabriz Petrochemical Company in Iran.

The coupling agent used in this study was maleated polypropylene (MAPP), obtained from Eastman Chemical Products, Inc.; as Epolene G-3003TM polymer with 8 % acid anhydride and a molecular weight of 103,500 (Nourbakhsh and Ashori, 2008).

Acetic anhydride (AA), a reagent required for the acetylation process, was purchased from Merck Company in Germany. Its molecular weight, purity, density, and melting point were 102.1 g/mol, 96 %, 1.08 g/cm<sup>3</sup>, and -73.1 °C, respectively (Kown and Nadir, 2015).

#### 2.2 Acetylation of beech wood flour

#### 2.2. Acetilacija drvnog brašna bukovine

A total of 500 g of oven-dried beech wood flour was placed in 2-L Erlenmeyer flasks along with acetic anhydride (96 %) for pre-impregnation. Catalysts were not used during the pre-impregnation process, which took place at room temperature for 60, 180, or 300 minutes as pre-impregnation times (*PITs*). The acetic anhydride treatment of the WF was conducted through a reaction at  $(103\pm2)$  °C for 60, 90, or 120 minutes as heating or reaction times (*H/RTs*). These reaction times were determined based on previous studies by Serin (2005), Dizman (2005), and Cavdar *et al.* (2014), respectively. After the reaction, the WF was thoroughly washed with semi-warm deionized water to remove any residual chemicals and by-products. The acetylated beech WF was then dried in an oven at (103±2) °C for 24 hours. Subsequently, it was placed in a desiccator containing phosphorous pentoxide until it reached room temperature, and the weight percentage gain (*WPG*) was determined using Eq. 1:

$$WPG(\%) = [(W2 - W1)/W1] * 100$$
 (1)

In the equation, *W1* represents the weight of the sample before treatment, and *W2* represents the weight of the sample after treatment.

#### 2.3 Production of WPCs

#### 2.3. Proizvodnja WPC-a

The composition of acetic anhydride-treated WF/ PP composites and the corresponding weight percentage gain (*WPG*) are presented in Table 1.

Before the sample preparation, both unmodified and modified beech wood flours were subjected to drying in an oven at a temperature of  $(103\pm2)$  °C for 24 hours. The mixing of the components was conducted using a Hake internal mixer (HBI System 90, USA) at a temperature of 180 °C and a rotation speed of 60 rpm.

The process involved feeding the PP into the mixing chamber, followed by the addition of the coupling agent (PP-g-MA) once the PP had melted. After five minutes, the WF was introduced, and the total mixing time was 11 minutes. The resulting mixture was then ground using a pilot scale grinder (Wieser, WGLS 200/200 Model, Germany). The obtained granules were subsequently dried at a temperature of 70 °C for 24 hours. Test samples were prepared by injection

<b>WPC or Treatment code</b> <i>WPC ili oznaka tretmana</i>	WF, wt %	PP, wt %	MAPP, wt %	PITs, min	<i>H/RTs</i> , min
PITs60-H/RTs60	60	40	2	60	60
PITs60- H/RTs90	60	40	2	60	90
<i>PITs</i> 60- <i>H</i> / <i>RTs</i> 120	60	40	2	60	120
<i>PITs</i> 180- <i>H/RTs</i> 60	60	40	2	180	60
<i>PITs</i> 180- <i>H/RTs</i> 90	60	40	2	180	90
<i>PITs</i> 180- <i>H</i> / <i>RTs</i> 120	60	40	2	180	120
<i>PITs</i> 300- <i>H</i> / <i>RTs</i> 60	60	40	2	300	60
<i>PITs</i> 300- <i>H</i> / <i>RTs</i> 90	60	40	2	300	90
<i>PITs</i> 300- <i>H/RTs</i> 120	60	40	2	300	120
Control	60	40	2	-	-

 Table 1 Procedure of chemical modification of wood flour

 Tablica 1. Postupak kemijske modifikacije drvnog brašna

*PITs* – Pre-impregnation times, *H/RTs* – Heating or reaction times, WF – Wood flour, PP – Polypropylene, MAPP – Maleic anhydride polypropylene / *PITs* – *vrijeme predimpregnacije*, *H/RTs* – *vrijeme zagrijavanja ili reakcije*, *WF* – *drvno brašno*, *PP* – *polipropilen*, *MAPP* – *anhidrid maleinske kiseline* – *polipropilen* 



Figure 2 Mechanical testing samples Slika 2. Uzorci za mehanička ispitivanja

molding according to the ASTM D 3641 (2015) standard using an injection molding machine at a temperature of 185 °C and a pressure of 10 MPa (Iman machine, Iran). Finally, the specimens were stored in controlled conditions with a relative humidity of 50 % and a temperature of 23 °C for at least 40 hours prior to testing. A sample product ready for mechanical testing is shown in Figure 2. Three replicated samples were tested for each property at each treatment level.

#### 2.4 Measurements

#### 2.4. Mjerenja

#### 2.4.1 Mechanical properties

#### 2.4.1. Mehanička svojstva

Flexural and tensile tests were carried out in accordance with ASTM D 790 (2007) and D 638 (2007) standards, respectively, using an Instron machine (Model 1186, England). The tests were performed at a crosshead speed of 5 mm/min. For the Izod impact test, a Zwick impact tester (Model SIT 20 D, Santam Co., Iran) was employed. To prepare the samples, a notch was made at the center of one longitudinal side, in accordance with the specifications provided in ASTM D 256 (2007).

#### 2.4.2 Physical properties 2.4.2. Fizička svojstva

Water absorption (*WA*) measurements were conducted following the guidelines outlined in ASTM D 570-98 (2010) standard. To determine the water absorption of the acetylated wood-plastic composite (WPC) specimens, three specimens of each WPC type were first dried in an oven at ( $103\pm2$ ) °C for 24 hours. The dried specimens were weighed with a precision of 0.001 g and then immersed in distilled water at room temperature for a period of 24 hours. After the immersion time elapsed, the specimens were carefully removed from the water and any excess moisture on the surface was removed using blotting paper. The weight of the specimens was then measured again after 24 hours of immersion in water at room temperature. The mean percentages of water absorption (WA) and thickness swelling (TS) over time were calculated for each composite (both modified and unmodified) and for each of the 10 formulations using Eqs. 2 and 3 as follows:

$$WA(t) \% = [(Wt - Wo) / Wo] \times 100$$
(2)

$$TS(t) \% = [(Tt - To) / To] \times 100$$
 (3)

Where WA (t) represents the water absorption at time *t*, Wo is the initial weight of the specimens, Wt is the weight of the specimens at time *t*, TS(t) denotes the thickness swelling at time *t*, To is the initial thickness of the specimens, and Tt is the thickness of the specimens at time *t*.

#### 2.5 Statistical analysis 2.5. Statistička analiza

The statistical analysis was performed using the SPSS software (Version 24) through the implementation of a general linear model (Univariate). To understand the combined effect of *PITs* (min) and *H/RTs* (min), a two-way ANOVA (analysis of variance) technique was used in the experiments. To assess the statistical significance at a significance level of p<0.05, the Duncan multiple range test was employed.

#### 3 RESULTS AND DISCUSSION 3. REZULTATI I RASPRAVA

### 3.1 Weight percent gain

#### 3.1. Povećanje mase

The result of the reaction between beech wood flour hydroxyl groups and acetic anhydride (AA), in other words the weight percentage gain of the acetic anhydride chemically modified WF is presented in Figure 3.

The individual effects of pre-impregnation times and heating/reaction times on the weight percent gain (*WPG*) of PP composites filled with AA-modified WF are summarized in Table 2.

Statistically, results showed that the interaction between the variables *PITs* and *H/RTs* was negative and had a significant effect on the weight percent gain of samples.

According to a statistical analysis, the individual effect of variables *PITs* and *H/RTs* showed significant effects on the *WPG* measured. The interaction between *PITs* and *H/RTs* was negative and indicated a significant effect on the *WPG* factor within the range of 95 % confidence for the experimental WPC samples made from AA-treated wood flour investigated.

The highest (7.18 %) and lowest (0.69 %) *WPG* levels were achieved for 300-120 min and 180-60 min of *PITs-H/RTs*, respectively (Figure 3).

**Table 2** Individual effect of pre-impregnation times and heating/reaction times on weight percent gain of PP composites filled with AA-modified WF

**Tablica 2.** Individualni utjecaj vremena predimpregnacije i vremena zagrijavanja/reakcije na povećanje mase PP kompozita ispunjenih AA modificiranim WF-om

	Pre-impregnation times / Vremena predimpregnacije							
Property / Svojstvo	60 min	180 min	300 min					
Weight percent gain, % / povećanje mase, %	1.48	4.31						
	Heating/reaction times / Vremena zagrijavanja/reakcije							
Property / Svojstvo	60 min	90 min	120 min					
Weight percent gain, % / povećanje mase, %	1.04	3.98	3.92					



**Figure 3** Interaction effect of pre-impregnation (*P1Ts*) and heating or reaction times (*H/RTs*) on weight percent gain of PP composites filled with AA-modified WF; Groups with the same letters in each row indicated no statistical difference (p < 0.05) between samples according to Duncan's multiple range test

**Slika 3.** Utjecaj interakcije predimpregnacije (*PITs*) i vremena zagrijavanja ili reakcije (*H/RTs*) na povećanje mase PP kompozita ispunjenih AA modificiranim WF-om; prema Duncanovu testu, grupe s istim slovima u svakom stupcu nisu pokazale statistički značajnu razliku (p < 0.05) među uzorcima

#### 3.2 Mechanical properties

#### 3.2. Mehanička svojstva

The mechanical properties of PP composites filled with acetic anhydride WF and Duncan's multiple range test results are shown in Figures 4, 5, 6, 7, and 8.

The individual effects of pre-impregnation times and heating/reaction times on the mechanical properties of PP composites filled with AA-modified WF are summarized in Tables 3 and 4, respectively.

### **3.2.1 Flexural strength** 3.2.1. Čvrstoća na savijanje

Statistically, it was shown that the interaction between the variables *PITs* and H/RTs was negative and exhibited a significant effect on the flexural strength

**Table 3** Individual effect of pre-impregnation times on mechanical properties of PP composites filled with AA-modified WF**Tablica 3.** Individualni utjecaj vremena predimpregnacije na mehanička svojstva PP kompozita ispunjenih AA modificiranimWF-om

Mechanical properties	Pre-impregnation times / Vremena predimpregnacije					
Mehanička svojstva	60 min	180 min	300 min			
Flexural strength, MPa / čvrstoća na savijanje, MPa	55.68 (2.4) <sup>b</sup>	52.02 (1.3) <sup>a</sup>	52.46 (3.4) <sup>a</sup>			
Flexural modulus, MPa / modul savijanja, MPa	4648 (2262) <sup>a</sup>	5996 (417) <sup>a</sup>	4958 (2239) ª			
Tensile strength, MPa / vlačna čvrstoća, MPa	35.73 (4.1) <sup>b</sup>	29.87 (2.2) <sup>a</sup>	29.66 (4.2) <sup>a</sup>			
Tensile modulus, MPa / modul elastičnosti pri vlačnom naprezanju, MPa	5561 (2387) <sup>b</sup>	4373 (1348) <sup>ab</sup>	3285 (1400) <sup>a</sup>			
Impact strength, J/m / čvrstoća na udarac, J/m	0.42 (0.1) <sup>a</sup>	0.41 (0.1) <sup>a</sup>	0.38 (0.0) <sup>a</sup>			

Groups with the same letters in each row indicated no statistical difference ( $p \le 0.05$ ) between samples according to Duncan's multiple rang test. The values in parentheses are standard deviations.

Prema Duncanovu testu, grupe s istim slovima u svakom retku nisu pokazale statistički značajnu razliku (p < 0.05) među uzorcima. Vrijednosti u zagradama standardna su odstupanja.

**Table 4** Individual effect of heating/reaction times on mechanical properties of PP composites filled with AA-modified WF

 **Tablica 4.** Individualni utjecaj vremena zagrijavanja/reakcije na mehanička svojstva PP kompozita ispunjenih AA modi 

 ficiranim WF-om

Mechanical properties	Heating/reaction times / Vremena zagrijavanja/reakcije					
Mehanička svojstva	60 min	90 min	120 min			
Flexural strength, MPa / čvrstoća na savijanje, MPa	53.55 (3.4) <sup>a</sup>	52.92 (3.0) <sup>a</sup>	53.69 (2.7) <sup>a</sup>			
Flexural modulus, MPa / modul savijanja, MPa	4007 (2376) <sup>a</sup>	5334 (1657) <sup>ab</sup>	6261 (381) <sup>b</sup>			
Tensile strength, MPa / vlačna čvrstoća, MPa	33.26 (4.5) <sup>a</sup>	30.03 (3.1) <sup>a</sup>	31.97 (5.4) <sup>a</sup>			
Tensile modulus, MPa / modul elastičnosti pri vlačnom naprezanju, MPa	5351 (1907) <sup>b</sup>	3655 (2491) <sup>a</sup>	4214 (930) ab			
Impact strength, J/m / čvrstoća na udarac, J/m	0.40 (0.1) <sup>a</sup>	0.42 (0.0) <sup>a</sup>	0.39 (0.0) <sup>a</sup>			

Groups with the same letters in each row indicated no statistical difference (p < 0.05) between samples according to Duncan's multiple rang test. The values in parentheses are standard deviations.

Prema Duncanovu testu, grupe s istim slovima u svakom retku nisu pokazale statistički značajnu razliku (p < 0.05) među uzorcima. Vrijednosti u zagradama standardna su odstupanja.



**Figure 4** Interaction effect of pre-impregnation (*PITs*) and heating or reaction times (*H/RTs*) on flexural strength of PP composites filled with AA-modified WF; Groups with the same letters in each column indicated no statistical difference (p < 0.05) between samples according to Duncan's multiple range test

**Slika 4.** Utjecaj interakcije predimpregnacije (*PITs*) i vremena zagrijavanja ili reakcije (*H/RTs*) na čvrstoću na savijanje PP kompozita ispunjenih AA modificiranim WF-om; prema Duncanovu testu, grupe s istim slovima u svakom stupcu nisu pokazale statistički značajnu razliku (p < 0.05) među uzorcima

(*FS*) of WPC samples. According to a statistical analysis, the individual effect of variable *PITs* showed a significant effect on the *FS* measured (Table 3), while the individual effect of variable *H/RTs* had no significant effect on the *FS* measured (Table 4). The interaction between *PITs* and *H/RTs* was negative and showed a significant effect on the *FS* factor within the range of 95% confidence for the experimental WPC samples made from AA-treated wood flour investigated (Figure 4).

The test results showed that the *PITs* and *H/RTs*, as reaction variables of chemical modification with AA, affected the mechanical and physical properties of the composite samples. Flexural strength (*FS*) values of samples increased with the decrease in pre-impregnation times considering the individual effect of *PITs*, while the decreases were not significant considering the individual effect of *H/RTs*. The *FS* values were determined to be between 55.39 MPa (control) and 57.9

MPa of the samples produced with WF modification at 60 minutes of *PITs* (Figure 4).

The greatest increase in FS was found in the samples modified with 60-60 min of *PITS-H/RTs* in the interaction effect of reaction variables (57.90 MPa) (Figure 4).

#### 3.2.2 Flexural modulus

#### 3.2.2. Modul savijanja

Statistically, it was shown that the interaction between the variables *PITs* and *H/RTs* was positive and had no significant effect on the flexural modulus (*FM*) of WPC samples (Figure 5).

According to a statistical analysis, the individual effect of variable H/RTs showed significant effect on the *FM* measured (Table 4), while the individual effect of variable *PITs* had no significant effect on the *FM* measured (Table 3). The interaction between *PITs* and



**Figure 5** Interaction effect of pre-impregnation (*PITs*) and heating or reaction times (*H/RTs*) on flexural modulus of PP composites filled with AA-modified WF; Groups with the same letters in each column indicated no statistical difference (p < 0.05) between samples according to Duncan's multiple range test

**Slika 5.** Utjecaj interakcije predimpregnacije (*PITs*) i vremena zagrijavanja ili reakcije (*H/RTs*) na modul savijanja PP kompozita ispunjenih AA modificiranim WF-om; prema Duncanovu testu, grupe s istim slovima u svakom stupcu nisu pokazale statistički značajnu razliku (p < 0.05) među uzorcima

*H/RTs* was positive and showed no significant effect on the *FM* factor within the range of 95 % confidence for the experimental WPC samples made from AA-treated wood flour investigated (Figure 5).

The use of different heating or reaction times (H/RTs) in the chemical modification process with AA led to an increase in the flexural modulus (FM). Among the various H/RTs tested, the highest increase in FM was observed with 120-minute H/RTs (6261 MPa) as individual effect; however, there was a significant difference in FM values between the 60-minute and 120-minute H/RTs. The greatest value in FM was found in the unmodified or control samples (6545 MPa) (Figure 5).

The lowest values in *FS* and *FM* were found in the samples modified with 300-90 min (49.71 MPa) and 300-60 min (2934 MPa) of *PITS-H/RTs* in the interaction effect of reaction variables, respectively (Figures 4 and 5). Compared to the control group, the WPCs produced in 60-60 min and 300-120 min of *PITs-H/RTs* provided about 4% higher *FS* and 2% lower *FM* values, respectively.

### **3.2.3 Tensile strength 3.2.3. Vlačna čvrstoća**

Statistically, it was shown that the interaction between the variables *PITs* and *H/RTs* was positive and had no significant effect on the tensile strength (*TS*) of WPC samples. According to a statistical analysis, the individual effect of variable *PITs* showed a significant effect on the *TS* measured (Table 3), while the individual effect of variable *H/RTs* had no significant effect on the *TS* measured (Table 4). The interaction between *PITs* and *H/RTs* was positive and showed no significant effect on the *TS* factor within the range of 95 % confidence for the experimental WPC samples made from AA-treated wood flour investigated (Figure 6).

The tensile strength (*TS*) values varied depending on the reaction variables, ranging from 29.62 MPa to 38.39 MPa in the control samples and samples produced by modifying WF in 60 min of *PITs* and 60 min of *H/RTs*, respectively (Figure 6).

The highest *TS* value was obtained from samples produced through a 60-minute pre-impregnation time (*PIT*). Remarkable changes were observed in *TS* values between the control samples and the modified ones.

#### 3.2.4 Tensile modulus

#### 3.2.4. Modul elastičnosti pri vlačnom naprezanju

Statistically, it was shown that the interaction between the variables *PITs* and *H/RTs* was negative and had a significant effect on the tensile modulus (*TM*) of WPC samples. According to a statistical analysis, the individual effect of variables *PITs* and *H/RTs* showed significant effects on the *TM* measured (Tables 3 and 4). The interaction between *PITs* and *H/RTs* was negative and showed a significant effect on the *TM* factor within the range of 95 % confidence for the experimental WPC samples made from AA-treated wood flour investigated (Figure 7).

In addition, the tensile modulus of the samples also exhibited an increase caused by variations in both reaction variables, namely *PITs* and *H/RTs*, during the acetylation of WF with AA. This increase in the *TM* can be attributed to the enhanced stiffness of WPCs (Figure 7). The increasing severity of pre-



**Figure 6** Interaction effect of pre-impregnation (*PITs*) and heating or reaction times (*H/RTs*) on tensile strength of PP composites filled with AA-modified WF; Groups with the same letters in each column indicated no statistical difference (p < 0.05) between samples according to Duncan's multiple range test

**Slika 6.** Utjecaj interakcije predimpregnacije (*PITs*) i vremena zagrijavanja ili reakcije (*H/RTs*) na vlačnu čvrstoću PP kompozita ispunjenih AA modificiranim WF-om; prema Duncanovu testu, grupe s istim slovima u svakom stupcu nisu pokazale statistički značajnu razliku (p < 0.05) među uzorcima

impregnation times and heating/reaction times negatively affected the tensile modulus and other mechanical properties.

### **3.2.5** Impact strength 3.2.5. Čvrstoća na udarac

Statistically, it was shown that the interaction between the variables *PITs* and *H/RTs* was positive and had no significant effect on the impact strength (*IS*) of WPC samples. According to a statistical analysis, the individual effect of variables *PITs* and *H/RTs* showed no significant effects on the *IS* measured (Tables 3 and 4). The interaction between *PITs* and *H/RTs* was positive and showed no significant effect on the IS factor within the range of 95 % confidence for the experimental WPC samples made from AA-treated wood flour investigated (Figure 8).

The impact strength (*IS*) values of samples increased with the decrease in pre-impregnation times in the individual effect of *PITs*, while the values did not



**Figure 7** Interaction effect of pre-impregnation (*PITs*) and heating or reaction times (*H/RTs*) on tensile modulus of PP composites filled with AA-modified WF; Groups with the same letters in each column indicated no statistical difference (p < 0.05) between samples according to Duncan's multiple range test

**Slika 7.** Utjecaj interakcije predimpregnacije (*PITs*) i vremena zagrijavanja ili reakcije (*H/RTs*) na modul elastičnosti pri vlačnom naprezanju PP kompozita ispunjenih AA modificiranim WF-om; prema Duncanovu testu, grupe s istim slovima u svakom stupcu nisu pokazale statistički značajnu razliku (p < 0,05) među uzorcima



**Figure 8** Interaction effect of pre-impregnation (*PITs*) and heating or reaction times (*H/RTs*) on impact strength of PP composites filled with AA-modified WF; Groups with the same letters in each column indicated no statistical difference (p < 0.05) between samples according to Duncan's multiple range test

**Slika 8.** Utjecaj interakcije predimpregnacije (*PITs*) i vremena zagrijavanja ili reakcije (*H/RTs*) na čvrstoću na udarac PP kompozita ispunjenih AA modificiranim WF-om; prema Duncanovu testu, grupe s istim slovima u svakom stupcu nisu pokazale statistički značajnu razliku (p < 0,05) među uzorcima

show significant changes for the individual effect of H/RTs. The *IS* values were found to be between 0.35 J/m (control) and 0.46 J/m from the samples produced by modifying WF in 60 min of *PITs* and 60 min of *H/RTs* (Figure 8).

Figure 4 clearly shows that the acetylation of beech WF with acetic anhydride had a positive effect on flexural strength, while the variation in the pre-impregnation and heating or reaction times had no positive effect on flexural modulus in comparison with unmodified or control samples (Figure 5).

The greatest increase in TS was found for samples modified with 60-60 min of *PITS-H/RTs* in the interaction effect of reaction variables (38.39 MPa) (Figure 6), and the greatest value in *TM* was also found in samples modified with 60-60 min of *PITS-H/RTs* in the interaction effect of reaction variables (7374 MPa) (Figure 7). The lowest values in *TS* and *TM* were found in samples modified with 300-90 min (27.95 MPa) and (1521 MPa) of *PITS-H/RTs* in the interaction effect of reaction variables, respectively.

Compared to the control group, composites produced with 60-60 min of *PITs-H/RTs* WF provided about 22 % and 26 % higher *TS* and *TM* values, respectively (Figures 6 and 7).

The greatest increase in the *IS* was found in samples modified with 60-60 min of *PITS-H/RTs* in the interaction effect of reaction variables (0.46 J/m), while the lowest values in the IS were found in the unmodified samples (0.35 J/m), 180-60 (0.37 J/m), and 300-60 min (0.37 J/m) of *PITS-H/RTs* in the interaction effect of reaction variables, respectively (Figure 8). Compared to the control group, the WPCs produced in 60-60 min of *PITs-H/RTs* WF provided about 23 % higher impact strength values.

#### **3.3 Physical properties 3.3. Fizička svojstva**

The individual effects of pre-impregnation times and heating/reaction times on the physical properties of PP composites filled with AA-modified WF are summarized in Tables 5 and 6, respectively.

**Table 5** Individual effect of pre-impregnation times on physical properties of PP composites filled with AA-modified WF**Tablica 5.** Individualni utjecaj vremena predimpregnacije na fizička svojstva PP kompozita ispunjenih AA modificiranimWF-om

Physical properties	Pre-impregnation times / Vremena predimpregnacije				
Fizička svojstva	60 min	180 min	300 min		
Water absorption (24 h), % / upijanje vode (24 h), %	1.86 (1.5) <sup>b</sup>	1.06 (0.2) <sup>a</sup>	1.66 (0.5) <sup>b</sup>		
Thickness swelling (24 h), % / debljinsko bubrenje (24 h), %	0.69 (4.3) <sup>a</sup>	0.19 (4.4) <sup>a</sup>	0.65 (4.1) <sup>a</sup>		

Groups with the same letters in each row indicated no statistical difference ( $p \le 0.05$ ) between samples according to Duncan's multiple rang test. The values in parentheses are standard deviations.

Prema Duncanovu testu, grupe s istim slovima u svakom retku nisu pokazale statistički značajnu razliku (p < 0.05) među uzorcima. Vrijednosti u zagradama standardna su odstupanja.

Physical properties	rties Heating/Reaction times / Vremena za					
Fizička svojstva	60 min	90 min	120 min			
Water absorption (24 h), % / upijanje vode (24 h), %	2.29 (1.3) <sup>b</sup>	1.20 (0.3) <sup>a</sup>	1.08 (0.2) <sup>a</sup>			
Thickness swelling (24 h), % / debljinsko bubrenje (24 h), %	-0.59 (5.7) <sup>a</sup>	1.19 (1.1) <sup>a</sup>	0.94 (4.4) <sup>a</sup>			

Table 6 Individual effect of heating/reaction times on physical properties of PP composites filled with AA-modified WF	
Tablica 6. Individualni utjecaj vremena zagrijavanja/reakcije na fizička svojstva PP kompozita ispunjenih AA modificiranim WF-	-om

Groups with the same letters in each row indicated no statistical difference (p < 0.05) between samples according to Duncan's multiple rang test. The values in parentheses are standard deviations.

Prema Duncanovu testu, grupe s istim slovima u svakom retku nisu pokazale statistički značajnu razliku (p < 0.05) među uzorcima. Vrijednosti u zagradama standardna su odstupanja.

The water absorption (*WA*) values of samples decreased with the increase in pre-impregnation times and heating or reaction times for the individual effects of *PITs* and *H/RTs* (Tables 5 and 6), where the values showed significant differences for both reaction variables. The same happened for thickness swelling (*TSw*), where the values showed significant differences in the individual effect of both reaction variables, but without special order. The greatest decrease in water absorption (*WA*) of 24 h was found for samples modified with 60-120 min of *PITS-H/RTs* (0.94 %) in the interaction effect of reaction variables (Table 7).

In contrast, the greatest increase value in *WA* of 24 h was found for samples modified with 60-60 min of *PITS-H/RTs* in the interaction effect of reaction variables (3.45 %). Compared to the control group, composites produced in 60-120 min and 60-60 min of *PITs-H/RTs* provided about 25 % lower and 63% higher *WA* values, respectively (Table 7).

The results showed that a significant reduction in *TSw* and *WA* properties of the WPCs was produced by using acetylated wood. This phenomenon may be explained by the fact that the amount of hydroxyl groups was substituted by acetyl moieties following the acetylation treatment, which leads to the decrease in the hydrophilicity of wood particles (Hung *et al.*, 2016). The greatest decrease in thickness swelling (*TSw*) of 24 h was found in samples modified with 180-60 min of *PITS-H/RTs* in the interaction effect of reaction variables (-2.58 %). In contrast, the greatest increase value in *WA* of 24 h was found in the samples modified with 60-60 min of *PITS-H/RTs* in the interaction effect of reaction variables (2.43 %). Compared to the control group, composites produced in 180-60 min and 60-60

min of *PITs-H/RTs* provided about 197 % lower and 8 % higher *TSw* values, respectively (Table 7).

The acetylation process significantly impacted the mechanical properties of wood-polymer composites (WPCs), including flexural strength (FS), tensile strength (TS), tensile modulus (TM), and impact strength (IS), with variations observed based on weight percentage gain (WPG) and pre-impregnation and heating or reaction times. Lower levels of WPG and shorter reaction times resulted in improved properties, indicating enhanced interfacial adhesion and polymer-WF interaction. However, higher levels of WPG and longer reaction times exhibited poor interfacial adhesion. Previous studies have highlighted the better bonding achieved between acetylated WF and the polymer matrix, although this improvement did not necessarily translate into improved mechanical properties of the composites.

This study introduced the use of acetic anhydride (AA) in different pre-impregnation and heating or reaction times, leading to enhancements in flexural strength, tensile strength, tensile modulus, and impact strength of WPCs. Similar results were found in previous studies (Abdul Khalil et al., 2002; Segerholm et al., 2007; Kwon and Ayrilmis, 2015). Acetylated wood particles contribute to the sufficient interfacial adhesion with the polymer matrix, which leads to a more efficient transfer of stress along the wood particle/polymer interface as compared to the control wood/polymer (Hung et al., 2016). However, no significant change was observed in the flexural modulus. This can be attributed to the altered failure mode, shifting from fiber pullout to fiber breakage, caused by improved bonding between the polymer and WF. Acetylated WF

**Table 7** Individual effect of pre-impregnation and heating or reaction times on physical properties of PP composites filled with AA-modified WF

**Tablica 7.** Individualni utjecaj vremena predimpregnacije i zagrijavanja ili reakcije na fizička svojstva PP kompozita ispunjenih AA modificiranim WF-om

<b>Physical properties</b>		Treatment code (PITs/HRTs) / Oznaka tretmana (PIT/HRT), min									
Fizička svojstva	60-60	60-90	60-120	180-60	180-90	180-120	300-60	300-90	300-120	Control	
$W_A(2A_b) = 0/$	3.45°	1.19ª	0.94ª	1.12ª	0.95ª	1.11ª	2.32 <sup>b</sup>	1.46 <sup>ab</sup>	1.20ª	1.26ª	
WA (24 II), 70	(0.4)	(0.1)	(0.1)	(0.1)	(0.4)	(0.1)	(0.0)	(0.2)	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	(0.3)	
$TS_{\rm eff}(241) 0/$	2.43ª	0.00ª	-0.35ª	-2.58ª	1.85ª	1.30 <sup>a</sup>	-1.63ª	1.73ª	1.87ª	2.65ª	
<i>I SW</i> (24 II), 70	(0.0)	(0.0)	(0.8)	(0.8)	(0.8)	(0.3)	(0.7)	(0.1)	1/, IIIII           00         300-120           .b         1.20a           0         (0.1)           a         1.87a           0         (0.8)	(0.6)	

WA – Water absorption; TSw – Thickness swelling. Groups with the same letters in each row indicated no statistical difference (p < 0.05) between samples according to Duncan's multiple range test. The values in parentheses are standard deviations.

WA - upijanje vode; TSw - debljinsko bubrenje. Prema Duncanovu testu, grupe s istim slovima u svakom retku nisu pokazale statistički značajnu razliku (p < 0,05) među uzorcima. Vrijednosti u zagradama standardna su odstupanja.

exhibited higher flexural strength compared to unmodified WF, with a slight reduction in flexural modulus potentially attributed to the weight gain caused by acetylation. The density is one of the significant parameters affecting the mechanical properties.

The compatibility between WF and the polymer matrix improved due to reduced polarity after acetylation, resulting in increased mechanical properties. However, excessive acetylation levels could hinder effective stress transmission from the matrix to the fibers, leading to reduced composite properties. The flexural strength, tensile strength, and moduli, as well as impact strength, were significantly increased in samples treated with AA, even with lower WPG compared to other treatments. However, a linear correlation between pre-impregnation and reaction times and mechanical properties was not observed. Inadequate washing to remove unreacted chemicals and by-products after modification may have contributed to the reduction and fluctuation in properties. The application of a Soxhlet cleaning procedure after acetylation was found to have a positive effect on the mechanical properties of WPCs compared to simple water washing.

The success of chemical modification in WF for WPCs depends on various factors, such as the method used, reaction time and temperature, and removal of unreacted chemicals. Substituting hydroxyl groups in WF with acetyl groups increased the hydrophobicity of the WF surface, improving compatibility with the polymer matrix. Additionally, the upward trend observed in flexural strength, tensile strength, and moduli, and impact strength with increasing *WPG* may be attributed to reduced moisture sorption and increased brittleness of the samples. The use of anhydrides and higher reaction temperatures influenced the cross-linking of lignin in WF during the formation of cellulose-reinforced thermoplastic composites.

As a result of the chemical reaction between wood and acetic anhydride as the reagent, an increase in dimensions of the reacted wood species has occurred, because of swelling of the wood cell wall. In most cases, chemically modified wood has a lower capacity for water absorption, with lower equilibrium moisture content at a specified atmospheric relative humidity, compared to unmodified wood (Teaca *et al.*, 2014).

#### 4 CONCLUSIONS

#### 4. ZAKLJUČAK

In this research, WPCs were made from acetylated wood flour. The highest and lowest *WPG* levels were achieved for 300-120 min for *PITs-H/RTs* and 180-60 min for *PITs-H/RTs*, respectively. The test results revealed that the chemical modifications of WF in different PITs and H/RTs improved the mechanical and physical properties of the WPCs. The highest mechanical properties were found in the WPCs produced with wood acetylated at PITs-H/RTs for 60-60 min with the exception of impact strength, while the least WA and TSw were found in the PITs-H/RTs for 60-120 min and PITs-H/RTs for 300-60 min, respectively. The WPCs produced with wood acetylated at PITs-H/RTs for 180-60 min affected the physical characteristics, while the positive effect did not affect all the mechanical properties compared to unmodified or control samples. Regarding the physical and mechanical properties of the WPCs, it was observed that surface modification by both anhydride acetic and reaction times improved the tensile and flexural strengths as well as WA and TSw of the composites. This was because the surface modification could improve the compatibility between wood flour and PP matrix, leading to fewer microvoids and fiber-PP debonding in the interphase region. Simultaneous use of pre-impregnation and reaction times had a synergic effect on the physical and mechanical properties.

#### Acknowledgements – Zahvala

The authors are grateful for the support of the Department of Wood Science and Paper Technology, Karaj Branch, Islamic Azad University.

#### 5 REFERENCES 5. LITERATURA

J. LITERATORA

- Abdul Khalil, H. P. S.; Rozeman, H. D.; Ismail, H.; Rosfaizal.; Ahmad, M. N., 2002: Polypropylene (PP) – Acacia mangium composites: the effect of acetylation on mechanical and water absorption properties. Polymer-Plastics Technology and Engineering, 41 (3): 453-468. https://doi.org/10.1081/PPT-120004362
- Cavdar, A. D.; Mengeloglu, F.; Karakus, K.; Tomak, E. D., 2014: Effect of chemical modication with maleic, propionic and succinic anhydrides on some properties of wood flour filled HDPE composites. BioResources, 9 (4): 6490-6503. https://doi.org/10.15376/biores.9.4.6490-6503
- de Prá Andrade, M.; Poletto, M., 2021: Wood treatments and interfacial bonding in wood-plastic composites. In: Mavinkere Rangappa, S.; Parameswaranpillai, J.; Kumar, M. H.; Siengchin, S. (eds.), Wood Polymer Composites. Composites Science and Technology. Springer, Singapore. https://doi.org/10.1007/978-981-16-1606-8\_3
- Dizman, E., 2005: The effects of chemical modification on the physical, mechanical and biological properties of alder and spruce particleboards. MSc Thesis, Institute of Natural Science, Karadeniz Technical University, Trabzon.
- Ermeyda, M. A., 2014: Wood cell wall modification with hydrophobic molecules. PhD Thesis, Institutional Repository of the University of Potsdam, Mai.
- Farsi, M., 2010: Wood-plastic composites: influence of wood flour chemical modification on the mechanical performance. Journal of Reinforced Plastics and Composites, 29 (4): 3587-3592. https://doi.org/10.1177/073168 4410378779

- Gwon, J. G.; Lee, S. Y.; Chun, S. J.; Doh, G.-H.; Kim, J.-H., 2010: Effect of chemical treatments of wood fibers on the physical strength of polypropylene based composites. Korean Journal of Chemical Engineering, 27 (2): 651-657. https://doi.org/10.1007/s11814-010-0058-1
- Hill, C. A. S., 2006: Wood Modification Chemical, Thermal and Other Processes. John Wiley & Sons, England.
- Hung, K.-C.; Wu, T.-L.; Chen, Y.-L.; Wu, J.-H., 2016: Assessing the effect of wood acetylation on mechanical properties and extended creep behavior of wood/recycled-polypropylene composites. Construction and Building Material, 108: 139-145. https://doi.org/10.1016/j. conbuildmat.2016.01.039
- Kallakas, H.; Shamim, M. A.; Olutubo, T.; Poltimae, T.; Suld, T. M., 2015: Effect of chemical modification of wood flour on the mechanical properties of wood-plastic composites. Agronomy Research, 13 (3): 639-653.
- Kwon, J. H.; Ayrilmis, N., 2015: Combined effect of acetylation and wax emulsion on physical and mechanical properties of particleboard. European Journal of Wood and Wood Products, 73: 845-847. https://doi. org/10.1007/s00107-015-0957-2
- Larsson-Brelid, P.; Segerholm, B. K.; Westin, M.; Walinder, M. E. P., 2006: Wood plastic composites from modified wood. Part 1: Conceptual idea, mechanical and physical properties. The 37<sup>th</sup> Annual Meeting, IRG/WP 06-40338, Tromsø, Norway.
- Lyon, F.; Thevenon, M. F.; Pizzi, A.; Tondi, G.; Despres, A.; Gril, J.; Rigolet, S., 2008: Wood preservation by a mixed anhydride treatment: Using simple models of polymeric wood compounds for a 13C NMR investigation. IRG 39<sup>th</sup> Annual Conference, IRG/WP 08-30457, Istanbul, Turkey.
- Nourbakhsh, A.; Ashori, A., 2008: Highly fiber-loaded composites: Physical and mechanical properties. Polymers & Polymer Composites, 16 (5): 343-347.
- Özmen, N.; Cetin, N. S.; Mengeloglu, F.; Birinci, E.; Karakus, K., 2013a: Effect of vinyl acetate modification on the properties of wood-plastic composites. BioResources, 8 (1): 753-767. https://doi.org/10.15376/ biores.8.1.753-767
- Özmen, N.; Cetin, N. S.; Mengeloglu, F.; Birinci, E., 2013b: Vinyl acetate modified Scots pine reinforced HDPE composites: Influence of various levels of modification on mechanical and heat properties. BioResources, 8 (1): 1361-1373. https://doi.org/10.15376/biores. 8.1.1361-1373
- Papadopoulos, A. N.; Bikiaris, D. N.; Mitropoulos, A. C.; Kyzas, G. Z., 2019: Nanomaterials and chemical modifications for enhanced key wood properties: A Review. *Nanomaterials*, 9 (4): 607. https://doi.org/10.3390/ nano9040607
- Rowell, R. M., 2007: Homopolymers, Blends and Composites. In Handbook of Engineering Biopolymers: edited by Fakirov, S.; Bhattacharya, D., Munich: Hanser Publications, Germany, pp. 673-691.

- Rowell, R. M., 2006a: Chemical modification of wood: A short review. Wood Material Science and Engineering, 1 (1): 29-33. https://doi.org/10.1080/17480270600670923
- Rowell, R. M., 2006b: Acetylation of wood Journey from analytical technique to commercial reality. Forest Product Journal, 56 (9): 4-12.
- Rowell, R. M., 2005: Handbook of Wood Chemistry and Wood Composites. Boca Raton: Taylor and Francis, Florida.
- Rowell, R. M., 1983: Chemical modification of wood: A review. Forest Products Abstracts, 6 (12): 363-382.
- 23. Segerholm, K.; Walinder, M.; Larsson, B. P.; Walkenstorm, P.; Westin, M., 2005: Wood plastic composites made from acetylated wood: Effects on water vapor sorption behavior and durability. In: Proceedings of the 9<sup>th</sup> European Panel Products Symposium, Llandudno, Wales, UK, pp. 233-242.
- 24. Segerholm, B. K.; Walkenström, P.; Nyström, B.; Wålinder, M. E. P.; Brelid, P. L., 2007: Micromorphology, moisture sorption and mechanical properties of a biocomposite based on acetylated wood particles and cellulose ester. Wood Material Science & Engineering, 2 (3-4): 106-117. https://doi.org/10.1080/17480270801906175
- Serin, Z. O., 2005: The effects of succinic anhydride modification on properties of medium density fiberboard (MDF). PhD Thesis, Institute of Natural Science, Karadeniz Technical University, Trabzon.
- Teacă, C.-A.; Bodirlau, R.; Spiridon, I., 2014: Maleic anhydride treatment of softwood – Effect on wood structure and properties. Cellulose Chemistry and Technology, 48 (9-10): 863-868.
- Tesinova, P., 2011: Advances in Composite Materials Analysis of Natural and Man-Made Material. InTech, p. 584.
- Thybring, E. E.; Fredriksson, M., 2021: Wood modification as a tool to understand moisture in wood. Forests, 12 (3): 372. https://doi.org/10.3390/f12030372
- Zhou, T.; Liu, H., 2022: Research progress of wood cell wall modification and functional improvement: A review. Materials, 15: 1598. https://doi.org/10.3390/ma15041598
- \*\*\*ASTM D256:2007 Standard test method for determining the izod pendulum impact resistance of plastics. ASTM International, West Conshohocken, PA.
- \*\*\*ASTM D 638:2007 Standard test method for tensile properties of plastics. ASTM International, West Conshohocken, PA.
- 32. \*\*\*ASTM D 790:2007 Standard test method for flexural properties of unreinforced and reinforced plastics and electrical insulating materials. ASTM International, West Conshohocken, PA.
- 33. \*\*\*ASTM D 570-98:2010 Standard Test Method for Water Absorption of Plastics. ASTM International, West Conshohocken, PA.
- \*\*\*ASTM D 3641:2015 Standard practice for injection molding test specimens of thermoplastic molding and extrusion materials. ASTM International, West Conshohocken, PA.

**Corresponding address:** 

#### SEYYED KHALIL HOSSEINIHASHEMI

Department of Wood Science and Paper Technology, Karaj Branch, Islamic Azad University, Karaj, Iran, e-mail: hashemi@kiau.ac.ir

Mesut Uysal, Cagatay Tasdemir, Dogan Memis<sup>1</sup>

# Effect of Epoxy Resin Reinforcement on Screw Withdrawal Strength of Fiberboard and Particleboard Used in Furniture Industry

Utjecaj ojačanja epoksidnom smolom na izvlačnu silu vijka u ploča vlaknatica i iverica koje se rabe u proizvodnji namještaja

#### **ORIGINAL SCIENTIFIC PAPER**

Izvorni znanstveni rad Received – prispjelo: 3. 8. 2023. Accepted – prihvaćeno: 10. 11. 2023. UDK: 674.812 https://doi.org/10.5552/drvind.2023.0146 © 2023 by the author(s). Licensee University of Zagreb Faculty of Forestry and Wood Technology. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license.

**ABSTRACT** • The study aimed to increase the screw withdrawal strength of medium density fiberboard and particleboard used in furniture strength by using epoxy resin in the screw pilot hole. Therefore, the effects of pilot hole diameters, screw diameter, and amount of epoxy resin on screw withdrawal strength of medium density fiberboard and particleboard from face and edge were investigated. According to TS EN 13446, 50 mm × 50 mm specimens were cut from commercial medium density fiberboard and particleboard boards. A static load was applied parallel to the screw direction. The screw withdrawal strength of medium density fiberboard was higher than the screw withdrawal strength of particleboard because of its density. Besides, the screw withdrawal strength of medium density fiberboard and particleboard samples with a 3.5 mm screw diameter was higher compared to those with a 4.5 mm screw diameter. A decrease in pilot hole diameter and an increase in the amount of epoxy resin provided higher screw withdrawal strength of materials. Using 20 % epoxy resin of the volume of the pilot hole resulted in two times better screw withdrawal strength values. The study showed that a higher amount of epoxy resin, smaller pilot hole diameter, and smaller screw diameter contribute to better screw withdrawal strength of both medium density fiberboard and particleboard from the face and edge.

**KEYWORDS:** *epoxy resin; reinforcement; screw withdrawal strength; medium density fiberboard; particle- board* 

**SAŽETAK** • Cilj ovog istraživanja bio je povećati izvlačnu silu vijka u ploča vlaknatica i iverica koje se rabe u proizvodnji namještaja, i to upotrebom epoksidne smole u pilot-rupama za vijke. Stoga je ispitan utjecaj promjera pilot-rupa, promjera vijka i količine epoksidne smole na izvlačnu silu vijka na plohi i rubu ploče vlaknatice i iverice. Uzorci dimenzija 50 mm × 50 mm prema TS EN 13446 izrađeni su od komercijalne srednje guste ploče vlaknatice i ploče iverice. Statičko opterećenje djelovalo je paralelno na smjer vijka. Zbog razlika u gustoći ploča

<sup>&</sup>lt;sup>1</sup> Authors are assistant professor, assistant professor and research assistant at Bursa Technical University, Department of Forest Industry Engineering, Bursa, Turkey. https://orcid.org/0000-0003-0114-3030; https://orcid.org/0000-0002-7161-630X; https://orcid.org/0000-0002-3907-0896

izvlačna sila vijka bila je veća za srednje gustu ploču vlaknaticu nego za ploču ivericu. Osim toga, izvlačna sila vijka promjera 3,5 mm u srednje gustoj ploči vlaknatici i ploči iverici bila je veća od izvlačne sile vijka promjera 4,5 mm. Smanjenje promjera pilot-rupe i povećanje količine epoksidne smole omogućilo je veću otpornost materijala na izvlačenje vijaka. Upotrebom 20 % epoksidne smole u odnosu prema volumenu pilot-rupe rezultiralo je dvostruko boljim vrijednostima izvlačne sile vijka pridonose boljoj izvlačnoj sili vijka na plohi i rubu srednje guste ploče vlaknatice i ploče iverice.

KLJUČNE RIJEČI: epoksidna smola; ojačanje; izvlačna sila vijka; srednje gusta ploča vlaknatica; ploča iverica

#### **1 INTRODUCTION**

#### 1. UVOD

Joints are the weakest component of furniture construction (Smardzewski, 2009). In the case of any joint failure, all construction fails. Therefore, the reusability of the furniture decreases over time due to the fact that furniture joints loosen or cannot provide its serviceability. Especially, in case furniture, screw joints are a typical example where a furniture member or screw is not damaged, but the screw hole enlarge. Besides, screws are used for demountable furniture to be easily assembled and disassembled. However, screw holes enlarge after screwing off, so the screw pilot hole diameter increases and the joint does not have the initial strength after re-assembly. Larger screws could be preferred for re-assembly efforts, which increases the strength of the screw joints because the pilot hole diameter was increased in that screw withdrawn from the material (Uysal et al., 2015). However, the use of larger joints may not be suitable for wood-composite materials, especially edgewise, because of their thickness and internal bond strength. On the contrary, the use of adhesive in screw pilot holes is another method to increase the screw withdrawal strength (SWS) of materials. Epoxy resin (EP) is a better selection to reinforce screw withdrawal strength compared to other adhesives because it provides higher mechanical properties for adhesives containing EP as reinforcement.

An increase in joint strength decreases the environmental impact of furniture because its serviceability and life span are increased since the rate of discarded furniture in landfills decreases while the rate of reusability increases. Besides, the repairability of the furniture construction with larger screws or the use of adhesive could increase the life span of the furniture. Remanufacturing furniture is another way to increase furniture life span and decrease its environmental impact before being discarded in landfills. However, the remanufacturing market of furniture is responsible for 0.4 % of the furniture industry, and the number of employees is less compared to employees in the reuse network (Parker *et al.*, 2015). In addition, furniture remanufacturing requires more energy than furniture reuse, according to the product recovery hierarchy in the end-of-life options (Östlin *et al.*, 2009). Hence, increasing the service life of furniture with screw joints could be provided by increasing the *SWS* of materials used in construction, so the second users can reuse furniture after it fulfills its service life for the first users.

Screw-type joints are mainly used in demountable furniture for easy assembly and disassembly. Therefore, it should be considered how joint strength would be increased after the first use or dissembling. Fasteners like screws, minifixes, confirmats, etc. are effective joinery systems to join elements made of wood-based composites and are used in case furniture. The strength of joints depends on the material, joinery system, screw pilot hole, number of fasteners and distance between fasteners. However, it is significant to have a self-locking system for each fastener to hold material. In addition, freezing and heating significantly affect material withdrawal capacity (Gašparík et al., 2023). Thus, increasing the SWS of materials comes into prominence to improve the overall strength of furniture joints. It depends on their density and internal bonding strength, diameter of pilot hole, length of screw penetration, diameter of screw shank and thread, and filling material injected in a pilot hole (Smardzewski et al., 2016). Besides, thread geometry such as thread height, flank distance, teeth angle, and thread angle significantly affect the SWS of the material (Hoelz et al., 2022).

Fiberboard (MDF) and particleboard (PB) are widely used as raw materials for Ready-to-Assembly (RtA) furniture. Screw-type fasteners are not only used in joints but also in handles, hinges and locks (Örs *et al.*, 1998). Studies related to joints with screws have been conducted with the technological development in wood composites and engineered wood since the strength design of furniture was defined in the 1940s (Eckelman, 1973; Smardzewski, 2009; Uysal and Haviarova, 2019). Such studies will continue to be conducted to enhance the strength of screw joints as long as the furniture is the greatest need of people's living spaces and as long as new technological developments emerge in the field. The *SWS* of material increased by 9-45 % and 7-39 % with injected polyvinyl acetate ad-

hesive in pilot holes for MDF and PB compared to those without adhesive. In addition, an increase in the thread depth of the screw provided higher SWS (Örs et al., 1998). The correlation between SWS and internal bonding strength is better than that of density and was studied to predict the internal bond strength of PB from its SWS (Semple and Smith, 2005). The larger-diameter screws had a greater SWS than smaller-diameter screws, but thread depth is more critical than screw diameter for the SWS of materials (Abu and Ahmad, 2015). Polyvinyl acetate and polyurethane adhesive were used in screw pilot holes to increase joint strength in RtA furniture (Imirzi et al., 2016). The type of material and type of screw affected the SWS of the material (Wolpiuk and Sydor, 2016). The SWS of a material increases with higher material density (Bal et al., 2017; Jivkov et al., 2017). The SWS of wood-based materials was inversely proportional to the modulus of elasticity because of high stiffness (Yunus et al., 2019). The effects of screw type, screw orientation, and adhesive on the SWS of MDF and PB were studied. Results showed that the use of adhesive in pilot holes increased the SWS of both MDF and PB (Yorur et al., 2020). The pilot hole diameter and density effects on the SWS of MDF were studied. Results revealed that pilot diameter had more impact on SWS than material density. Besides, coarse thread screws have lower SWS than fine thread screws; namely, the distance between screw threads is also a significant factor. SWS from the face was greater than SWS from the edge because the face density of MDF is higher than the core density. SWS is also affected by the adhesive composition used in particleboard (Farajollah Pour et al., 2022). The reinforcement type in a material increases its SWS due to higher density (Perçin and Uzun, 2022). Different size and amount of waste melamine impregnated paper (WMIP) as adhesive in particle boards were studied (Başboğa et al., 2023). SWS of particleboard increased with narrower size and higher amount of WMIP.

EP was recommended for frame construction with biscuit joints (Yildirim *et al.*, 2017). EP provides a more durable adhesion with its mechanical adhesion force. It achieves more and faster penetration to the adhesion surface pores on the material than other adhesives due to its higher elasticity (Karaman *et al.*, 2017). Also, an increase in the mechanical interlocking mechanism between wood material and the surface of metal fasteners with EP enhances the *SWS* of wood-based composite materials. In light of this information, the present study was carried out with the idea of adding EP as reinforcement to the screw joint to increase the *SWS* of materials. In doing so, the effects of pilot hole diameters, screw diameter, and amount of EP on *SWS* of MDF and PB from face and edge were investigated.

#### 2 MATERIALS AND METHODS

#### 2. MATERIJALI I METODE

#### 2.1 Design of experiment

#### 2.1. Postavke eksperimenta

In this study, MDF and PB materials, widely used in panel-type furniture and commercially produced, were investigated. A complete four-factor factorial experiment with five replications in each sample group was conducted to determine the *SWS* of the MDF and PB. For this purpose, two materials (MDF and PB), two screw diameters (3.5 mm and 4.5 mm), three pilothole diameters (80 %, 90 % and 100 % of screw diameter) and four levels of the EP amount (no-resin, 20 %, 40 % and 60 % of volume of the pilot-hole) factors were selected for edge and face withdrawal from specimens. In doing so, 480 specimens in 96 sample groups were tested to examine the effect of the EP on the *SWS* of the materials.

### 2.2 Preparation of specimens

#### 2.2. Priprema uzoraka

According to TS EN 13446, all specimens were cut into 50 mm  $\times$  50 mm nominal dimensions from 18 mm  $\times$  2100 mm  $\times$  2800 mm of MDF and PB boards obtained from a local store in the Inegol furniture cluster in Turkey. Table 1 shows the average density and moisture content of the MDF and PB used in the study. The dimensions of the specimens and orientations for the pilot holes are given in Figure 1.a and b. All pilot holes were drilled according to screw diameter and depth of screw penetration in MDF and PB. The depth of the screw penetration is equal to 8d (d: screw diameter) or at most 30 mm for *SWS* from the edge and 16 mm for *SWS* from the face of the materials. Besides, the depth of the pilot hole was 70 % of the depth of the screw penetration.

**Table 1** Average density and moisture content of materials**Tablica 1.** Prosječna gustoća i sadržaj vode u pločama

Matarial	<b>Density</b> / <i>Gustoća</i> , g/cm <sup>3</sup>							ture conter	1t, %
Material Air-dry		Oven-dry			Sadržaj vode, %				
Materijal	Mean	SD*	CoV*	Mean	SD*	CoV*	Mean	SD*	CoV*
MDF	0.70	0.004	0.57 %	0.67	0.004	0.60 %	6.98	0.07	1.01 %
PB	0.62	0.006	0.91 %	0.59	0.006	1.01 %	8.78	0.03	0.30 %

\*SD - standard deviation, CoV - coefficient of variation / \*SD - standardna devijacija, CoV - koeficijent varijacije

Screw orientations Orijentacija vijka	For 3.5 × 45 mm screw (4.5 × 45 mm screw) Za vijak 3.5 × 45 mm (vijak 4.5 × 45 mm)							
	<b>Depth of pilot hole, mm</b> Dubina pilot-rupe, mm	<b>Diameter of pilot hole, mm</b> <i>Promjer pilot-rupe,</i> mm	<b>Amount of EP, μl</b> Količina epoksidne smole, μl					
			20 %	40 %	60 %			
Edge / rub	19.60 (21.00)	2.80 (3.60)	26 (50)	50 (100)	75 (150)			
		3.15 (4.05)	32 (60)	60 (120)	90 (180)			
		3.50 (4.50)	40 (80)	80 (160)	115 (240)			
Face / ploha	11.20 (11.20)	2.80 (3.60)	20 (32)	40 (65)	60 (100)			
		3.15 (4.05)	25 (40)	50 (80)	75 (120)			
		3.50 (4.50)	30 (50)	60 (100)	90 (150)			

 Table 2 Dimensions of pilot hole and amount of EP for each sample group

 Tablica 2. Dimenzije pilot-rupe i količina epoksidne smole za svaku skupinu uzoraka

\* Values in parathesis are for screws with a diameter of 4.5 mm. / \* Vrijednosti u zagradama odnose se na vijke promjera 4,5 mm.

 Table 3 Length, penetration depth, diameter, and root diameter of screws used in the study

 Tablica 3. Duljina, dubina prodiranja, promjer i promjer unutarnjeg navoja vijaka upotrijebljenih u istraživanju

		<b>3.5 × 45 mm screw (4.5 × 45 mm screw)</b> / Vijak 3,5 × 45 mm (vijak 4,5 × 45 mm)*						
		Length	Penetration depth	Diameter	Root diameter			
		Duljina	Dubina prodiranja	Promjer	Unutarnji promjer			
Edea	Mean	44.14 (44.21)	27.88 (29.24)	3.54 (4.46)	2.33 (2.80)			
Rub	SD	0.39 (0.35)	0.73 (0.93)	0.07 (0.04)	0.04 (0.02)			
	CoV	0.88 % (0.78 %)	2.61 % (3.19 %)	1.86 % (0.84 %)	1.79 % (0.60 %)			
<b>F</b>	Mean	44.15 (44.19)	15,96 (16.05)	3.53 (4.48)	2.30 (2.80)			
Ploha -	SD	0.40 (0.34)	0.67 (0.69)	0.07 (0.02)	0.02 (0.05)			
	CoV	0.90 % (0.76 %)	4.21 % (4.29 %)	2.02 % (0.41 %)	0.78 % (1.78 %)			

\* Values in parathesis are for screws with a diameter of 4.5 mm. / \* Vrijednosti u zagradama odnose se na vijke promjera 4,5 mm.



**Figure 1** Pilot hole position for a) face withdrawal and b) edge withdrawal

**Slika 1.** Položaj pilot-rupe: a) za izvlačenje iz plohe, b) za izvlačenje iz ruba

Table 2 shows pilot hole dimensions and the amount of EP for each sample group. Epoxy and hardener, obtained from a local store, were mixed to a 5/2 ratio for 3 minutes. Epoxy and hardener mix were exposed to air for at most 15 minutes before the screws were driven. After each specimen was drilled for a pilot hole and an appropriate amount of EP was injected, 3.5 and 4.5 mm diameter screws were driven into pilot holes to the above length. Table 3 shows the average length, penetration depth, diameter, and root diameter of the screws used in the study.

### 2.3 Determination of screw withdrawal strength

#### 2.3. Određivanje izvlačne sile vijka

All tests for the *SWS* were conducted on the SHI-MADZU universal test machine according to TS EN 13446. Withdrawal loads were obtained by applying a withdrawal load parallel to the screw axis from the edge and face of specimens with a rate of 0.8 mm/min and 2 mm/min, respectively, and continued until the ultimate load was reached (Figure 2). *SWS*, f (MPa), was calculated by using Eq. 1.

$$f = \frac{F_{\rm ult}}{d \cdot l_{\rm p}} \tag{1}$$

 $F_{ult}$  – Ultimate withdrawal load (N) d – Diameter of screw tread (mm)  $l_p$  – depth of screw penetration (mm)



**Figure 2** Test configuration for a) edge withdrawal and b) face withdrawal

**Slika 2.** Ispitna konfiguracija: a) za izvlačenje iz plohe, b) za izvlačenje iz ruba

#### 2.4 Statistical analysis

#### 2.4. Statistička analiza

Data collected for all specimens were checked for determining statistical significance through Fourway ANOVA analysis, and Tukey pairwise comparisons were carried out in Minitab Statistical Analysis Software (Minitab Inc. 2013).

#### 3 RESULTS AND DISCUSSION

3. REZULTATI I RASPRAVA

### 3.1 Screw withdrawal strength from the edge

#### 3.1. Izvlačna sila vijka na rubu

Results for *SWS* from the edge are given in Figure 3. According to test results, the average *SWS* from the edge of MDF was 9.26 MPa with a standard deviation of 0.56 MPa provided that the pilot hole was 80 % of the 3.5 mm screw diameter. An increase in the pilot hole diameter from 80 % to 90 % and 100 % decreased *SWS* of the MDF by 7.89 MPa and 3.42 MPa with standard deviations of 1.32 MPa and 0.47 MPa, respec-

tively. If 4.5 mm diameter screws were used, the average SWS from the edge of MDF was 8.05 MPa with a standard deviation of 0.50 MPa, 6.80 MPa with a standard deviation of 0.28 MPa and 2.88 MPa with a standard deviation of 0.13 MPa for 80 %, 90 % and 100 % pilot hole of 4.5 mm-screw-diameter, respectively. If PB was used, the average SWS from the edge was 6.91 MPa with a standard deviation of 0.63 MPa, 4.92 MPa with a standard deviation of 0.44 MPa, and 3.50 MPa with a standard deviation of 0.22 MPa for 80 %, 90 % and 100 % pilot hole of 3.5 mm-screw-diameter, respectively. Moreover, those of 4.5 mm-screw diameter had 6.27 MPa with a standard deviation of 0.69 MPa, 5.17 MPa with a standard deviation of 0.78 MPa, and 2.80 MPa with a standard deviation of 0.19 MPa, respectively.

It could be observed that the average *SWS*s from the of MDF are higher than those of PB not only because of internal bond strength but also because the density of the material directly affected the *SWS* (Semple and Smith, 2005; Bal *et al.*, 2017; Jivkov *et al.*, 2017; Başboğa *et al.*, 2023). Besides, statistical analysis results



**Figure 3** Effect of amount of EP on *SWS* from edge of material for a) 3.5 mm and b) 4.5 mm screw diameter **Slika 3.** Utjecaj količine epoksidne smole na izvlačnu silu vijka na rubu materijala: a) za promjer vijka od 3,5 mm, b) za promjer vijka od 4,5 mm

shown in Table 4 and Table 5 documented that the type of material significantly affected SWS from the edge (pvalue = 0.000 < 0.05), and that the mean values of SWS from the edge of MDF and PB were significantly different from each other ( $\alpha = 0.05$ ), respectively. Furthermore, larger screw diameter levels yielded decreasing SWS values from the edge of both MDF and PB. Although larger screws provide higher SWS for materials, screw diameter affected SWS when drilling a pilot hole but not the screw insertion technique (Abu and Ahmad, 2015). Therefore, 4.5 mm screw sizes might have an adverse impact on SWS because of using EP. The average SWS from the edge for a 3.5 mm-screw diameter was 11.3 % higher than those of 4.5 mm. Also, Table 4 and Table 5 showed that screw diameter had a significant impact on the SWS from the edge (p-value = 0.000 <0.05) and that the mean values of SWS from the edge for 3.5 mm and 4.5 mm screw diameter were significantly different ( $\alpha = 0.05$ ), respectively.

Pilot hole diameter is vital to the strength of joints with screws because the difference between its shank diameter and screw teeth length affects SWS depending on the material. Therefore, it should be close to the shank diameter so that the screw is easily driven into the material without damaging it (Eckelman, 2003). On the other hand, joinery systems with screws allow easily assembled and disassembled furniture construction. However, every effort to disassemble furniture construction enlarges the pilot hole, and re-assembling the furniture results in lower strength of joints. The test results showed that larger pilot hole diameters decreased the SWS of materials, while the increased levels of EP injected in the pilot hole enhanced SWS. The amount of EP was increased by 20 %, 40 % and 60 % of the pilot hole diameter. In the case of MDF samples with a 3.5 mm screw diameter and 60 % EP, SWS from the edge was increased by 119.67 %, 156.56 %, and 425.59 % for 80 %, 90 % and 100 % pilot hole diameters, respectively. Those with 40 % and 20 % EP,

increased *SWS* as much as 69.49 %, 67.33 %, 292.66 % and 79.32 %, 111.55 % and 322.94 %, respectively.

Furthermore, SWS was increased by 94.24 % when the pilot hole diameter increased from 80 % to 100 %, and the amount of EP increased from 0 to 60 %. SWS increase in those with 40 % and 20 % EP was 45.08 % and 56.27 %, respectively. Not only did both the pilot hole (p-value =0.000) diameter and amount of EP (p-value = 0.000) have a significant effect on SWS from the edge, but the interaction of pilot hole diameter and amount of EP (p-value = 0.000) also significantly affected SWS from the edge (Table 4). Furthermore, as shown in Table 5, based on Tukey pairwise comparisons, the mean values of SWS from the edge for pilot hole diameter and amount of EP significantly differed at the 95 % confidence level, respectively. An SWS increase should be achieved with a screw pilot hole close to the root diameter (Farajollah Pour et al., 2022). The pilot hole diameter should be at least 70 % of the screw diameter (Eckelman, 2003). The holding mechanism of the screw on materials depends on the screw thread rather than the whole screw diameter. In this study, specimens with 70 % pilot hole diameter and EP failed in the case of drilling screws, so these sample groups were omitted from the experiment.

As can be observed in Table 4, interactions of (i) screw diameter and pilot hole diameter, (ii) screw diameter, amount of EP, and material, (iii) screw diameter, amount of EP, and pilot hole diameter, (iv) screw diameter, material, and pilot hole diameter, (v) amount of EP, material, and pilot hole diameter, and (vi) screw diameter, amount of EP, material, and pilot hole diameter, and either hole diameter amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, and pilot hole diameter, amount of EP, material, amount of EP

### 3.2 Screw withdrawal strength from the face

#### 3.2. Izvlačna sila vijka na plohi

Test results for *SWS* from the face are given in Figure 4. According to test results, the average *SWS* from the face of MDF was 11.80 MPa with a standard

Tablica 4. Cetverosnijema AnovA za izvlačnu snu vijka na rubu						
Source / Izvor	DF	Adj SS	Adj MS	F-Value	<i>P</i> -Value	
Screw diameter (D) / promjer vijka (D)	1	89.59	89.59	50.09	0.000	
Epoxy resin (EP) / epoksidna smola (EP)	3	3068.02	1022.67	571.79	0.000	
Material (M) / materijal (M)	1	746.91	746.91	417.61	0.000	
Pilot hole (P) / pilot-rupa (P)	2	206.88	103.44	57.84	0.000	
D*EP	3	73.63	24.54	13.72	0.000	
D*M	1	20.85	20.85	11.66	0.001	
EP*M	3	199.32	66.44	37.15	0.000	
EP*P	6	86.00	14.33	8.01	0.000	
M*P	2	55.96	27.98	15.64	0.000	
Error	217	388.11	1.79			
Lack-of-Fit	25	88.71	3.35	2.11	0.003	
Pure Error	192	304.40	1.59			
Total	239	4935.27				

**Table 4** Four-way ANOVA for *SWS* from the edge **Tablica 4**. Četverosmierna ANOVA za izvlačnu silu vijka na rubu

Variables / Varijable	N	Average Prosjek	Grouping / Grupiranje				
Motorial / matavital	MDF	120	13.19	А			
Material / materijal	PB	120	9.67		В		
	3.5 mm	120	12.04	А			
Screw diameter / promjer vijka	4.5 mm	120	10.82		В		
	60 %	60	15.37	А			
Amount of ED / koližing anakaidaa guala	40 %	60	12.88		В		
Amount of EP / koncina epoksiane smole	20 %	60	11.81			C	
	Control	60	5.66				D
	80 %	80	12.41	А			
Pilot hole diameter / promjer pilot-rupe	90 %	80	11.71		В		
	100 %	80	10.18			C	

**Table 5** Tukey pairwise comparisons for *SWS* from the edge of material **Tablica 5.** Tukey usporedbe parova za izvlačnu silu vijka na rubu materijala

deviation of 0.85 MPa if the pilot hole was 80 % of the 3.5 mm screw diameter. Increasing pilot hole diameter from 80 % to 90 % and 100 % decreased *SWS* of the MDF by 9.45 MPa and 7.34 MPa with standard deviations of 0.85 MPa and 0.31 MPa, respectively. If 4.5 mm diameter screws were used, the average *SWS* from the face of MDF was 9.71 MPa with a standard deviation of 0.47 MPa, 7.96 MPa with a standard deviation

of 0.85 MPa and 5.26 MPa with a standard deviation of 0.31 MPa for 80 %, 90% and 100 % pilot hole of 4.5 mm-screw-diameter, respectively. If PB was used, the average *SWSs* from the face were 8.97 MPa with a standard deviation of 0.57 MPa, 6.49 MPa with a standard deviation of 0.72 MPa, and 5.93 MPa with a standard deviation of 0.60 MPa for 80 %, 90 % and 100 % pilot hole of 3.5 mm-screw-diameter, respec-



**Figure 4** Effect of amount of EP on *SWS* from the face of material for a) 3.5 mm and b) 4.5 mm screw diameter **Slika 4.** Utjecaj količine epoksidne smole na izvlačnu silu vijka na plohi materijala: a) za promjer vijka od 3,5 mm, b) za promjer vijka od 4,5 mm

tively. Furthermore, those of 4.5 mm-screw diameters had 7.02 MPa with a standard deviation of 0.35 MPa, 5.93 MPa with a standard deviation of 0.22 MPa, and 4.68 MPa with a standard deviation of 0.69 MPa, respectively.

Four-Way ANOVA results presented in Table 6 showed that all independent variables, namely, screw diameter (p-value = 0.000), material type (p-value = 0.000), amount of EP (*p*-value = 0.000), and pilot hole diameter, significantly affected the mean values of SWS from the face of specimens. In terms of the impact of material type on SWS from the face, a similar pattern was obtained with those from the edge; namely, the mean SWS from the face of MDF (11.80 MPa) was higher than that of PB (8.97 MPa) because MDF is a denser material with a higher internal bonding strength than PB. Tukey pairwise comparisons identified and documented the presence of SWS from the face differences across various levels of each independent factor. Table 7 demonstrated that the mean SWS from the face of MDF was significantly different from that of PB ( $\alpha$ = 0.05). Table 5 also showed that changing screw diameter levels resulted in significantly different mean SWS from the face values. Besides, the mean SWS from the face for 3.5 mm and 4.5 mm screw diameters were significantly different from each other ( $\alpha = 0.05$ ). As given in Table 6, the interaction of screw diameter and material also significantly affected the average SWS from the face (p-value = 0.000).

A smaller pilot hole diameter and higher amount of EP resulted in higher mean *SWS* from the face of both MDF and PB. According to Table 6, the interaction of pilot hole diameter and amount of EP (*p*-value = 0.001) significantly impacted *SWS* from the face. The average *SWS* from the face of MDF increased by 69.94 %, 73.09 %, and 101.28 % for 80 %, 90 % and 100 % pilot hole diameter when using a 3.5 mm screw diameter, and the amount of EP was 60 % of pilot hole diameter, respectively. For the same pilot hole diameters, the increase in SWS from the face for the samples with 4.5 mm screw diameter and the amount of EP of 60 % of pilot hole diameter was 68.74 %, 46.85 % and 117.86 %, respectively. Similar increases in the mean SWS from the face of MDF and PB were observed for the samples with the amount of EP equal to 20 % and 40 % of the pilot hole diameter. According to Tukey pairwise comparison results, the mean SWS from the face for the samples with the amount of EP equal to 60 % of the pilot hole diameter and without resin was significantly different. However, there was not enough statistical evidence to claim that the mean SWS from the face values for the groups with an amount of EP equal to 20 % and 40 % of the pilot hole diameter (Table 7) were different. Furthermore, Table 7 also demonstrated that the mean SWS from the face values for 80 %, 90 % and 100 % pilot hole diameters were significantly different from each other. Table 6 also showed that interactions of (i) screw diameter, amount of EP, and pilot hole diameter and (ii) all parameters had no significant effects on the SWS from the face.

#### 4 CONCLUSIONS 4. ZAKLJUČAK

This study investigated the effect of EP in the pilot hole on the *SWS* of MDF and PB. For this purpose, three pilot hole diameters, four amounts of EP and two screw diameters were investigated.

Internal bond strength and density of material have a significant effect on *SWS*. No matter what pilot

Source / Izvor		Adj SS	Adj <i>MS</i>	F-Value	<i>P</i> -Value
Screw diameter (D) / promjer vijka (D)		613.29	613.29	522.98	0.000
Epoxy resin (EP) / epoksidna smola (EP)	3	1602.82	534.27	455.60	0.000
Material (M) / materijal (M)		61.67	61.67	52.59	0.000
Pilot hole (P) / pilot-rupa (P)	2	193.89	96.94	82.67	0.000
D*EP	3	58.40	19.46	16.60	0.000
D*M	1	51.42	51.42	43.84	0.000
D*P	2	10.50	5.25	4.48	0.012
EP*M	3	57.38	19.13	16.31	0.000
EP*P	6	28.09	4.68	3.99	0.001
M*P	2	83.46	41.73	35.59	0.000
D*EP*M	3	10.27	3.43	2.92	0.035
D*M*P	2	10.75	5.37	4.58	0.011
EP*M*P	6	15.95	2.66	2.27	0.039
Error	204	239.23	1.17		
Lack-of-Fit	12	18.22	1.52	1.32	0.210
Pure Error	192	221.01	1.15		
Total	239	3037.11			

 Table 6 Four-way ANOVA for SWS from the face

 Tablica 6. Četverosmjerna ANOVA za izvlačnu silu vijka na plohi
Variables / Varijable		N	<b>Average</b> Prosjek	Grouping / Grupiranje		
Material / materijal	MDF	120	12.32	А		
	PB	120	11.31		В	
Screw diameter / promjer vijka	3.5 mm	120	13.41	А		
	4.5 mm	120	10.21		В	
Amount of EP / količina epoksidne smole	60 %	60	14.48	А		
	40 %	60	12.80		В	
	20 %	60	12.43		В	
	Control	60	7.54			С
Pilot hole diameter / promjer pilot-rupe	80 %	80	12.98	А		
	90 %	80	11.67		В	
	100 %	80	10.79			С

**Table 7** Tukey pairwise comparisons for SWS from the face of material

 **Tablica 7.** Tukey usporedbe parova za izvlačnu silu vijka na plohi materijala

hole diameter, amount of EP and screw diameter were used, the SWS of MDF was higher than the SWS of PB. An increase in screw diameter from 3.5 to 4.5 mm significantly affects the SWS of MDF and PB. In the edge and face withdrawal of screws from materials, smaller screw diameter has higher SWS. Increase in pilot hole diameter decreases the SWS of materials. 80 % pilot hole diameter had greater SWS than that of 90 % and 100 % because it was the closest to the root diameter. In addition, an increase in the amount of EP in the pilot hole increases the SWS of the material. EP dispenses more and faster in material pores, so it works better after drilling screws in pilot holes. If a piece of furniture with screw-type fasteners were disassembled, its pilot hole would be close to 100 % of or larger than the screw diameter. Using EP in the pilot hole for the following assembly would be beneficial. SWS of materials with 80 % pilot hole diameter and no-EP increased by 40 % compared to that of 100 % pilot hole with 60 % EP of the volume of the pilot hole.

The study showed that the existence of EP in the pilot hole increases the *SWS* of both MDF and PB. In this way, it could be used to repair furniture with screw-type fasteners. Moreover, it has a longer life span before being discarded in landfills and increases the reusability and repairability of case furniture to decrease its environmental impacts.

### Acknowledgements - Zahvala

This research was funded by the Scientific Research Projects Coordinatorship at Bursa Technical University (Project no: 211N018, 2021).

## 5 REFERENCES 5. LITERATURA

- Abu, F.; Ahmad. M., 2015: Effects of screw insertion on screw withdrawal strength. International Journal of Advances in Applied Sciences, 2: 25-29.
- 2. Bal, B. C.; Akcakaya, E.; Gundes, Z., 2017: Screw-holding capacity of melamine-faced fiberboard and particle-

board used in furniture production. Mugla Journal of Science and Technology, 4: 49-52.

- Başboğa, İ. H.; Taşdemir, Ç.; Yüce, Ö.; Mengeloğlu, F., 2023: Utilization of different size waste melamine impregnated paper as an adhesive in the manufacturing of particleboard. International Journal of Adhesion and Adhesives, 120: 103275. https://doi.org/10.1016/j.ijadhadh.2022.103275
- Eckelman, C. A., 1973: Holding strength of screws in wood and wood-based materials. Purdue University Agricultural Experiment Station Research Bulletin, 895: 1-15.
- 5. Eckelman, C. A., 2003: Textbook of product engineering and strength design of furniture. Purdue University, West Lafayette, USA.
- Farajollah Pour, M.; Hatefnia, H.; Dorieh, A., 2022: Research on medium density fiberboard (MDF) behavior against screw axial withdrawal: Impact of density and operational variables. Structures, 39: 194-206. https:// doi.org/10.1016/j.istruc.2022.03.025
- Gašparík, M.; Karami, E.; Kytka, T., 2023: The influence of freezing and heating on the nail withdrawal capacity of Norway spruce and European larch wood. European Journal of Wood and Wood Products, 81: 387-398. https://doi.org/10.1007/s00107-022-01898-0
- Hoelz, K.; Dörner, P. T.; Hohlweg, J.; Matthiesen, S., 2022: Influence of thread parameters on the withdrawal capacity of wood screws to optimize the thread geometry. European Journal of Wood and Wood Products, 80: 529-540. https://doi.org/10.1007/s00107-022-01792-9
- Imirzi, H. O.; Ozkaya, K.; Efe, H., 2016: Determination of the strength of L-type corner joints obtained from wood-based board materials using different joining techniques. Forest Products Journal, 66: 214-224. https://doi. org/10.13073/FPJ-D-14-00081
- Jivkov, V.; Kyuchukov, B.; Simeonova, R.; Marinova, A., 2017: Withdrawal capacity of screws and confirmat into different wood-based panels. In: Proseedings of the XX-VIII<sup>th</sup> International Conference on research for furniture Industry, Poznan, Poland, pp. 68-82.
- Karaman, A.; Yildirim, M. N.; Guven, S.; Yesil, H., 2017: Determination of tensile and compressive strength of Ttype furniture assemblies prepared with wood biscuits and polyvinylacetate (PVAc-D4) and epoxy adhesives. Journal of Advanced Technology Sciences, 6: 17-27.
- Örs, Y.; Özen, R.; Doğanay, S., 1998: Screw holding ability (strength) of wood materials used in furniture manufacture. Turkish Journal of Agriculture and Forestry, 22: 29-34.

- Östlin, J.; Sundin, E.; Björkman, M., 2009: Product lifecycle implications for remanufacturing strategies. Journal of Cleaner Production, 17: 999-1009. https://doi. org/10.1016/j.jclepro.2009.02.021
- Parker, D.; Riley, K.; Robinson, S., 2015: Remanufacturing market study. In: Eur. remanufacturing netw (online). https://www.remanufacturing.eu/assets/pdfs/remanufacturing-market-study.pdf (Accessed Jul. 1, 2023).
- Perçin, O.; Uzun, O., 2022: Screw withdrawal strength of heat-treated and laminated veneer lumber reinforced with carbon and glass fibers. BioResources, 17: 2486-2500. https://doi.org/10.15376/biores.17.2.2486-2500
- Semple, K. E.; Smith, G. D., 2005: Prediction of internal bond strength in particleboard from screw withdrawal resistance models. Wood and Fiber Science, 38: 256-267.
- Smardzewski, J.; Rzepa, B.; Kilic, H., 2016: Mechanical properties of externally invisible furniture joints made of wood-based composites. BioResources, 11: 1224-1239. https://doi.org/10.15376/biores.11.1.1224-1239
- Smardzewski, J., 2009: The reliability of joints and cabinet furniture. Wood Res, 54: 67-76.
- Uysal, M.; Haviarova, E., 2019: Lower tolerance limits for screw withdrawal strength in wood. Wood and Fiber Science, 51: 375-386. https://doi.org/10.22382/wfs-2019-036
- Uysal, M.; Haviarova, E.; Eckelman, C. A., 2015: A comparison of the cyclic durability, ease of disassembly, repair and reuse of parts of wooden chair frames. Materials

& Design, 87: 75-81. https://doi.org/10.1016/j.matdes.2015.08.009

- Wolpiuk, M.; Sydor, M., 2016: Practical screw withdrawal strength in chosen wood-based composites. Annals of WULS Forestry and Wood Technology, 96: 310-314.
- Yildirim, M. N.; Karaman, A.; Aykan, R.; Yesil, H., 2017: Effect of epoxy resin on bending momentum in L-type corner joins. Artvin Coruh University Journal of Forestry Faculty, 18: 211-217. https://doi.org/10.17474/artvinofd.307974
- Yorur, H.; Birinci, E.; Gunay, M. N.; Tor, O., 2020: Effects of factors on direct screw withdrawal resistance in medium density fiberboard and particleboard. Maderas. Ciencia y Tecnología, 22: 375-384. https://doi.org/10.4067/S0718-221X2020005000311
- 24. Yunus, N. Y. M.; Amali, N. W. A.; Tamat, N. S. M.; Rahman, W. M. N. W. A., 2019: Flexural influence on screw withdrawal behavior of selected commercial particleboard. International Journal of Advanced Engineering and Technology, 9: 5948-5951. https://doi.org/10.35940/ ijeat.A3033.109119
- 25. \*\*\*Minitab Inc. 2013: MINITAB statistical software. Version Release 16 10.
- \*\*\*TS EN 13446:2002 Wood-based panels. Determination of withdrawal capacity of fasteners. Turkish Standards Institution, Ankara, TURKEY, 2002.

#### **Corresponding address:**

#### Assist. Prof. MESUT UYSAL, PhD

Bursa Technical University, Yildirim, Bursa, TURKEY, e-mail: mesut.uysal@btu.edu.tr

# Drvo topole *Populus* spp.

# MIKROSKOPSKA OBILJEŽJA

Drvo topole rastresito je porozno. Traheje su mu sitne (promjera od 80 do 100 µm), brojne i guste (od 40 do 180 na 1 mm<sup>2</sup> poprečnog presjeka). Raspoređene su pojedinačno, u parovima i kratkim radijalnim nizovima te u skupinama. Perforacija članaka traheja potpuna je. Volumni udio traheja iznosi 22 do 44 %. Staničje drvnih trakova je homogeno. Drvni su traci isključivo jednoredni, visoki od 10 do 15 stanica. Gustoća drvnih trakova je od 8 do 13 na milimetar tangentnog smjera, a udio im je 10 do 14 %. Drvna su vlakanca libriformska, duljine od 0,3 do 2,1 mm. Debljina staničnih stijenki kreće se od 2,2 do 4,7 µm, a širina lumena od 11,5 do 23,5 µm. Udio drvnih vlakanaca je od 56 do 63 %. Uzdužni je parenhim zanemariv. Iako su makroskopski slični, drvo roda Populus može se strukturom drvnih trakova mikroskopski razlikovati od drva roda Salix (drvo roda Salix ima heterocelularne, a drvo roda Populus homocelularne drvne trakove).

# RELEVANTNE SPOZNAJE O STRUKTURI I VARIJACIJAMA DRVA TOPOLE

Određena anatomska obilježja drva topole poput duljine i promjera drvnih vlakanaca, debljine stanične stijenke (Ehrenberg, 1970.) te udjela različitih elemenata drva (Harris, 1970.) prepoznati su kao važna svojstva već u ranom razvoju programa oplemenjivanja šumskog drveća.

S obzirom na komercijalno rasprostranjenu specifičnu namjenu sirovine, postojanje traheja u drvu mnogih vrsta listača, pa tako i topola, veliki je problem u proizvodnji papira (Lundqvist, 2002.). Iako je uloga traheja u živom stablu nezamjenjiva, iz perspektive proizvodnje papira one su iznimno nepoželjne. U razvijanju vrsta predviđenih za takvu namjenu, osobito pri oplemenjivanju brzorastućih klonova i hibrida topola, treba težiti postizanju što manje veličine i broja traheja. Sve to upućuje na važnost istraživanja koja će obuhvatiti karakteriziranje stanica drva (drvnih vlakanaca i traheja), odnosno određivanje njihovih dimenzija, rasporeda i udjela u strukturi drva.

Pregled istraživanja varijabilnosti unutar i između godova drva topole donosi ove spoznaje: a) da je rast duljine drvnih vlakanaca drva topole u prvih deset do dvadeset godina starosti drveta izrazito pojačan (Boyce i Kaiser, 1961.); b) da za promjer drvnih vlakanaca i tra-

# **MICROSCOPIC CHARACTERISTICS**

Poplar wood is diffuse porous. Vessels are small (diameter 80 to 100 µm), numerous and dense (40 to 180 on 1 mm<sup>2</sup> of cross section). They are distributed individually, in pairs and in short radial rows. Perforation of vessel members is simple. Wood rays are exclusively uniseriate, 10 to 15 cells in height. The density of wood rays is 8 to 13 per tangential mm, and the volume fraction 10 to 14 %. Wood fibres are libriform, with a length of 0.3 to 2.1 mm. The cell wall thickness ranges from 2.2 to 4.7 µm, and lumen diameter from 11.5 to 23.5 µm. The volume fraction of wood fibres is from 56 to 63 %. Axial parenchyma is negligible. Although they are macroscopically similar, wood of the genus Populus can be microscopically distinguished from wood of the genus Salix according to the structure of the wood rays (wood of the genus Salix has heterocellular, and wood of the genus Populus homocellular wood rays).

## RELEVANT KNOWLEDGE ABOUT POPLAR WOOD STRUCTURE AND VARIATIONS

Certain anatomical characteristics of poplar wood, such as fibre length and fibre diameter, cell wall thickness (Ehrenberg, 1970) and proportions of different wood elements (Harris, 1970) were recognized as important already in the early development of forest tree breeding programmes.

Considering the commercially widespread specific use of the raw material, in many hardwood species, such as poplar, the presence of vessels represents a major problem in paper production (Lundqvist, 2002). Although their function in the living tree is indispensable, from the perspective of paper production they are extremely undesirable. When developing species intended for such a purpose, especially when breeding fast-growing poplar clones and hybrids, the aim should be to achieve the smallest possible size and number of vessels. All the above points to the importance of research that includes the characterization of wood cells (fibres and vessels), that is, the determination of their dimensions, distribution and proportion in the wood structure.

A review of the research on the intra- and interring variability of poplar wood provides knowledge on: (a) a pronounced growth in the fibre length of poplar wood in the first ten to twenty years of age (Boyce heja te za debljine njihovih stijenki vrijedi trend suprotan onome koji se odnosi na rast duljine vlakanaca (Bendtsen, 1978.); c) da se broj traheja u smjeru od srčike prema kori smanjuje, a njihov promjer između godova u istom smjeru raste (Kroll i dr., 1992.); d) da su udjeli stijenki stanica drva, udjeli lumena traheja i drvnih vlakanaca te udjeli drvnih trakova pokazali najmanje promjene duž radijusa (Pezlen, 1994.); e) da su radijalne varijacije određene rastom duljine vlakanaca i starošću drva, dok se dvostruka debljina stijenki i promjer lumena vlakanaca mijenjaju prema promjenjivom modelu među godovima (Ištok i dr., 2017.); f) da je odnos širine godova i duljine te promjera vlakanaca u drvu topole značajno negativan (Zhang i dr., 2022.).

Jedno od najvažnijih svojstava drva za tradicionalnu upotrebu topole jest mali udio tenzijskog drva. To se drvo po nizu biokemijskih, anatomskih i mehaničkih obilježja razlikuje od normalnog drva nastalog zbog nepostojanja mehaničkog podražaja, kao i od suprotnog drva koje se nalazi na donjoj strani nagnutog stabla (Pilate i dr., 2004.). Rane i nešto novije spoznaje o toj temi upućuju na zaključak: a) da je udio nepoželjnoga tenzijskog drva u križancima topole velik (Šefc i dr., 2009.); b) da su želatinozna vlakanca sadržana u tenzijskom drvu relativno uža od vlakanaca u normalnom drvu uočenih na poprečnom presjeku (Jouarez i dr., 2001.); c) da se od normalnog drva do drva s udjelom blagoga i jakog tenzijskog drva smanjila debljina stanične stijenke, isključujući želatinozni sloj (Fang i dr., 2008.); d) da je udio traheja u tenzijskom drvu manji (De Zio i dr., 2020.).

# LITERATURA / REFERENCES

- Bendtsen, B. A., 1978: Properties of wood from improved and intensively managed trees. Forest Products Journal, 28 (10): 61-72.
- Boyce, S. G.; Kaiser, M., 1961: Environmental and genetic variability in the length of fibers of eastern cottonwood. Tappi, 44: 363-366.
- De Zio, E.; Montagnoli, A.; Karady, M.; Terzaghi, M.; Sferra, G.; Antoniadi, I.; Scippa, G. S.; Ljung, K.; Chiatante, D.; Trupiano, D., 2020: Reaction Wood Anatomical Traits and Hormonal Profiles in Poplar Bent Stem and Root. Front. Plant Sci., 11: 1-16. doi: 10.3389/fpls.2020.590985
- Ehrenberg, C., 1970: Breeding for stem quality. Unasylva, 24: 23-31.
- Fang, C. H.; Guibal, D.; Clair, B.; Gril, J.; Liu, Y.-M.; Liu, S.-Q., 2008: Influence of growth stress level on wood properties in Poplar I-69 *Populus deltoides* Bartr.cv."Lux" ex I-69/55. Annals of Forest Science, 65: 307-315.
- Harris, J. M., 1970: Breeding to improve wood quality. Unasylva, 24: 32-36.
- Ištok, I.; Šefc, B.; Hasan, M.; Popović, G.; Sedlar, T., 2017: Fiber Characteristics of White Poplar (*Populus alba* L.) Juvenile Wood along the Drava River. Drvna industrija, 68 (3): 241-247. doi:10.5552/drind.2017.1729
- Jourez, B.; Riboux, A.; Leclercq, A., 2001: Anatomical characteristics of tension wood and opposite wood in young inclined stems of poplar (populus euramericana cv "ghoy"), IAWA J. 22: 133-157.

and Kaiser, 1961); (c) decrease in the number of vessels in the direction from the pith to the bark, while their diameter increased with cambial age in the same direction (Kroll *et al.* 1992); (d) proportions of cell walls, proportions of vessel and fibre lumens, and proportions of wood rays that showed the least changes along the radius (Pezlen, 1994); (e) the variation in radial pattern was characterized by increase in fibre length with cambial age, while double cell wall thickness and fibre lumen diameter varied in non-consistent radial pattern (Ištok *et al.* 2017); (f) a significantly negative relationship of tree-ring width to fibre length and fibre width in poplar wood (Zhang *et al.* 2022).

One of the most important wood properties for the traditional use of poplar is a small proportion of tension wood. Tension wood differs from normal wood formed in the absence of mechanical stimulus, and from opposite wood located on the lower side of the inclined stem, in a number of biochemical, anatomical and mechanical characteristics (Pilate et al. 2004). Early and more recent knowledge on the topic refers to: (a) large amount of undesired tension wood (Sefc et al. 2009); (b) gelatinous fibres present in tension wood were relatively narrower than the opposite wood fibres observed on cross section (Jouarez et al. 2001); (c) from normal wood to mild and severe tension wood, the thickness of cell wall excluding G-layer decreased (Fang et al. 2008); (d) tension wood was characterized by a low number of vessels (De Zio et al. 2020).

- Kroll, E. R.; Ritter, C. D.; Khuan, C. A., 1992: Anatomical and physical properties of balsam poplar (*Populus balsamifera* L.) in Minnesota. Wood and Fiber Science, 24 (1): 13-24.
- Lundqvist, S. O., 2002: Efficient wood and fiber characterization – a key factor in research and operation. Ann. For. Sci., 59: 491-501.
- 11. Peszlen, I., 1994: Influence of age on selected anatomical properties of populus clones. IAWA J., 15 (3): 311-321.
- Pilate, G.; Déjardin, A.; Laurans, F.; Leplé, J.-C., 2004: Tension wood as a model for functional genomics of wood formation. New Phytologist, 164: 63-72. doi: 10.1111/j.1469-8137.2004.01176.x
- Šefc, B.; Trajković, J.; Govorčin, S.; Despot, R.; Hasan, M., 2009: Selected tree characteristics and wood properties of two poplar clones. Wood Research, 54 (1): 15-22.
- Zhang, Y.; Fang, S.; Tian, Y.; Wang, L.; Lv, Y., 2022: Responses of radial growth, wood density and fiber traits to planting space in poplar plantations at a lowland site. J. For. Res. 33: 963-976. https://doi.org/10.1007/s11676-021-01382-0
- Wagenführ, R.; Scheiber, C., 2006: HOLZATLAS. VEB Fchbuchverlag, Leipzig, pp. 87-89.

doc. dr. sc. Iva Ištok

# Upute autorima

#### Opće odredbe

Časopis Drvna industrija objavljuje znanstvene radove (izvorne znanstvene radove, pregledne radove, prethodna priopćenja), stručne radove, izlaganja sa savjetovanja, stručne obavijesti, bibliografske radove, preglede te ostale priloge s područja biologije, kemije, fizike i tehnologije drva, pulpe i papira te drvnih proizvoda, uključujući i proizvodnu, upravljačku i tržišnu problematiku u drvnoj industriji. Predaja rukopisa podrazumijeva uvjet da rad nije već predan negdje drugdje radi objavljivanja ili da nije već objavljen (osim sažetka, dijelova objavljenih predavanja ili magistarskih radova odnosno disertacija, što mora biti navedeno u napomeni) te da su objavljivanje odobrili svi suautori (ako rad ima više autora) i ovlaštene osobe ustanove u kojoj je istraživanje provedeno. Cjelokupni sadržaj Drvne industrije dostupan je za skidanje s interneta, tiskanje, daljnju distribuciju, čitanje i ponovno korištenje bez ograničenja sve dok se naznače autor(i) i originalni izvor prema Creative Commons Attribution 4.0 International License (CC BY). Autor(i) zadržavaju izdavačka prava bez ograničenja.

Znanstveni i stručni radovi objavljuju se na engleskom jeziku, uz sažetak na hrvatskome. Također, naslov, podnaslovi i svi važni rezultati trebaju biti napisani dvojezično. Uredništvo osigurava inozemnim autorima prijevod na hrvatski. Ostali se članci uglavnom objavljuju na hrvatskome. Znanstveni i stručni radovi podliježu temeljitoj recenziji najmanje dvaju recenzenata. Izbor recenzenata i odluku o klasifikaciji i prihvaćanju članka (prema preporukama recenzenata) donosi Urednički odbor.

Svi prilozi podvrgavaju se jezičnoj obradi. Urednici će od autora zahtijevati da tekst prilagode preporukama recenzenata i lektora, te zadržavaju i pravo da predlože skraćivanje ili poboljšanje teksta. Autori su potpuno odgovorni za svoje priloge. Podrazumijeva se da je autor pribavio dozvolu za objavljivanje dijelova teksta što su već negdje objavljeni te da objavljivanje članka ne ugrožava prava pojedinca ili pravne osobe. Radovi moraju izvještavati o istinitim znanstvenim ili tehničkim postignućima. Autori su odgovorni za terminološku i metrološku usklađenost svojih priloga. Radovi se šalju elektronički putem poveznice http://journal.sdewes.org/drvind

#### Upute

Predani radovi smiju sadržavati najviše 15 jednostrano pisanih A4 listova s dvostrukim proredom (30 redaka na stranici), uključujući i tablice, slike te popis literature, dodatke i ostale priloge. Dulje je članke preporučljivo podijeliti na dva ili više nastavaka. Tekst treba biti u *doc formatu*, u potpunosti napisan fontom *Times New Roman* (tekst, grafikoni i slike), normalnim stilom, bez dodatnog uređenja teksta.

Prva stranica poslanog rada treba sadržavati puni naslov, ime(na) i prezime(na) autora, podatke o zaposlenju autora (ustanova, grad i država) te sažetak s ključnim riječima (duljina sažetka približno 1/2 stranice A4).

Posljednja stranica treba sadržavati titule, zanimanje, zvanje i adresu (svakog) autora, s naznakom osobe s kojom će Uredništvo biti u vezi. Znanstveni i stručni radovi moraju biti sažeti i precizni. Osnovna poglavlja trebaju biti označena odgovarajućim podnaslovima. Napomene se ispisuju na dnu pripadajuće stranice, a obrojčavaju se susljedno. One koje se odnose na naslov označuju se zvjezdicom, a ostale uzdignutim arapskim brojkama. Napomene koje se odnose na tablice pišu se ispod tablica, a označavaju se uzdignutim malim pisanim slovima, abecednim redom.

Latinska imena trebaju biti pisana kosim slovima (italicom), a ako je cijeli tekst pisan kosim slovima, latinska imena trebaju biti podcrtana.

U uvodu treba definirati problem i, koliko je moguće, predočiti granice postojećih spoznaja, tako da se čitateljima koji se ne bave područjem o kojemu je riječ omogući razumijevanje ciljeva rada.

Materijal i metode trebaju biti što preciznije opisane da omoguće drugim znanstvenicima ponavljanje pokusa. Glavni eksperimentalni podaci trebaju biti dvojezično navedeni.

Rezultati trebaju obuhvatiti samo materijal koji se izravno odnosi na predmet. Obvezatna je primjena metričkog sustava. Preporučuje se upotreba SI jedinica. Rjeđe rabljene fizikalne vrijednosti, simboli i jedinice trebaju biti objašnjeni pri njihovu prvom spominjanju u tekstu. Za pisanje formula valja se koristiti Equation Editorom (programom za pisanje formula u MS Wordu). Jedinice se pišu normalnim (uspravnim) slovima, a fizikalni simboli i faktori kosima (italicom). Formule se susljedno obrojčavaju arapskim brojkama u zagradama, npr. (1) na kraju retka.

Broj slika mora biti ograničen samo na one koje su prijeko potrebne za objašnjenje teksta. Isti podaci ne smiju biti navedeni i u tablici i na slici. Slike i tablice trebaju biti zasebno obrojčane, arapskim brojkama, a u tekstu se na njih upućuje jasnim naznakama ("tablica 1" ili "slika 1"). Naslovi, zaglavlja, legende i sav ostali tekst u slikama i tablicama treba biti napisan hrvatskim i engleskim jezikom.

Slike je potrebno rasporediti na odgovarajuća mjesta u tekstu, trebaju biti izrađene u rezoluciji 600 dpi, crno-bijele (objavljivanje slika u koloru moguće je na zahtjev autora), formata jpg ili tiff, potpune i jasno razumljive bez pozivanja na tekst priloga.

Svi grafikoni i tablice izrađuju se kao crno-bijeli prilozi (osim na zahtjev). Tablice i grafikoni trebaju biti na svojim mjestima u tekstu te originalnog formata u kojemu su izrađeni radi naknadnog ubacivanja hrvatskog prijevoda. Ako ne postoji mogućnost za to, potrebno je poslati originalne dokumente u formatu u kojemu su napravljeni (*excel* ili *statistica* format).

Naslovi slika i crteža ne pišu se velikim tiskanim slovima. Crteži i grafikoni trebaju odgovarati stilu časopisa (fontovima i izgledu). Slova i brojke moraju biti dovoljno veliki da budu lako čitljivi nakon smanjenja širine slike ili tablice. Fotomikrografije moraju imati naznaku uvećanja, poželjno u mikrometrima. Uvećanje može biti dodatno naznačeno na kraju naslova slike, npr. "uvećanje 7500 : l". Diskusija i zaključak mogu, ako autori žele, biti spojeni u jedan odjeljak. U tom tekstu treba objasniti rezultate s obzirom na problem postavljen u uvodu i u odnosu prema odgovarajućim zapažanjima autora ili drugih istraživača. Valja izbjegavati ponavljanje podataka već iznesenih u odjeljku Rezultati. Mogu se razmotriti naznake za daljnja istraživanja ili primjenu. Ako su rezultati i diskusija spojeni u isti odjeljak, zaključke je nužno napisati izdvojeno. Zahvale se navode na kraju rukopisa. Odgovarajuću literaturu treba citirati u tekstu, i to prema harvardskom sustavu (ime - godina), npr. (Bađun, 1965). Nadalje, bibliografija mora biti navedena na kraju teksta, i to abecednim redom prezimena autora, s naslovima i potpunim navodima bibliografskih referenci. Popis literature mora biti selektivan, a svaka referenca na kraju mora imati naveden DOI broj, ako ga posjeduje (http://www.doi.org) (provjeriti na http://www.crossref.org).

#### Primjeri navođenja literature

Članci u časopisima: Prezime autora, inicijal(i) osobnog imena, godina: Naslov. Naziv časopisa, godište (ev. broj): stranice (od – do). Doi broj.

#### Primjer

Kärki, T., 2001: Variation of wood density and shrinkage in European aspen (*Populus tremula*). Holz als Roh- und Werkstoff, 59: 79-84. http://dx.doi.org/10.1007/s001070050479.

Knjige: Prezime autora, inicijal(i) osobnog imena, godina: Naslov. (ev. izdavač/editor): izdanje (ev. svezak). Mjesto izdanja, izdavač (ev. stranice od – do).

#### Primjeri

Krpan, J., 1970: Tehnologija furnira i ploča. Drugo izdanje. Zagreb, Tehnička knjiga.

Wilson, J. W.; Wellwood, R. W., 1965: Intra-increment chemical properties of certain western Canadian coniferous species. U: W. A. Cote, Jr. (Ed.): Cellular Ultrastructure of Woody Plants. Syracuse, N.Y., Syracuse Univ. Press, pp. 551-559.

Ostale publikacije (brošure, studije itd.)

Müller, D., 1977: Beitrag zür Klassifizierung asiatischer Baumarten. Mitteilung der Bundesforschungsanstalt für Forstund Holzvvirt schaft Hamburg, Nr. 98. Hamburg: M. Wiederbusch.

#### Web stranice

\*\*\*1997: "Guide to Punctuation" (online), University of Sussex, www.informatics.sussex.ac.uk/department/docs/punctuation/node 00.html. First published 1997 (pristupljeno 27. siječnja 2010).

Autoru se prije konačnog tiska šalje pdf rada. Rad je potrebno pažljivo pročitati, ispraviti te vratiti Uredništvu s listom ispravaka. Autori znanstvenih i stručnih radova besplatno dobivaju po jedan primjerak časopisa. Autoru svakog priloga također se dostavlja besplatan primjerak časopisa.

Dodatne informacije o načinu pisanja znanstvenih radova mogu se naći na web adresi:

www.ease.org.uk/publications/author-guidelines

# Instructions for authors

#### General terms

The "Drvna industrija" ("Wood Industry") journal publishes scientific papers (original scientific papers, review papers, previous notes), professional papers, conference papers, professional information, bibliographical and survey articles and other contributions related to biology, chemistry, physics and technology of wood, pulp and paper and wood products, including production, management and marketing issues in the wood industry.

Submission of a paper implies that the work has not been submitted for publication elsewhere or published before (except in the form of an abstract or as part of a published lecture, review or thesis, in which case it must be stated in a footnote); that the publication is approved by all co-authors (if any) and by the authorities of the institution where the research has been carried out. The complete content of the journal *Drvna industrija* (Wood Industry) is available on the Internet permitting any users to download, print, further distribute, read and reuse it with no limits provided that the author(s) and the original source are identified in accordance with the Creative Commons Attribution 4.0 International License (CC BY). The authors retain their copyrights.

The scientific and professional papers shall be published in English with summary in Croatian. The titles, headings and all the relevant results shall be also presented bilingually. The Editor's Office shall provide the translation into Croatian for foreign authors. Other articles are generally published in Croatian. The scientific and professional papers will be subject to a thorough review by at least two selected referees. The Editorial Board shall make the choice of reviewers, as well as the decision about the classification of the paper and its acceptance (based on reviewers' recommendations).

All contributions are subject to proofreading. The editors will require authors to modify the text in the light of the recommendations made by reviewers and language advisers, and they reserve the right to suggest abbreviations and text improvements. Authors are fully responsible for the contents of their contributions. It shall be assumed that the author has obtained the permission for the reproduction of portions of text published elsewhere, and that the publication of the paper in question does not infringe upon any individual or corporate rights. Papers shall report on true scientific or technical achievement. Authors are responsible for the terminological and metrological consistency of their contributions. The contributions are to be submitted by the link http://journal.sdewes.org/drvind

#### Details

Papers submitted shall consist of no more than 15 single-sided DIN A-4 sheets of 30 double-spaced lines, including tables, figures and references, appendices and other supplements. Longer papers should be divided into two or more continuing series. The text should be written in doc format, fully written using Times New Roman font (text, graphs and figures), in normal style without additional text editing.

The first page of the paper submitted should contain full title, name(s) of author(s) with professional affiliation (institution, city and state), abstract with keywords (approx. 1/2 sheet DIN A4).

The last page should provide the full titles, posts and address(es) of each author with indication of the contact person for the Editor's Office.

Scientific and professional papers shall be precise and concise. The main chapters should be characterized by appropriate headings. Footnotes shall be placed at the bottom of the same page and consecutively numbered. Those relating to the title should be marked by an asterix, others by superscript Arabic numerals. Footnotes relating to the tables shall be printed under the table and marked by small letters in alphabetical order.

Latin names shall be printed in italics and underlined.

Introduction should define the problem and if possible the framework of existing knowledge, to ensure that readers not working in that particular field are able to understand author's intentions.

Materials and methods should be as precise as possible to enable other scientists to repeat the experiment. The main experimental data should be presented bilingually.

The results should involve only material pertinent to the subject. The metric system shall be used. SI units are recommended. Rarely used physical values, symbols and units should be explained at their first appearance in the text. Formulas should be written by using Equation Editor (program for writing formulas in MS Word). Units shall be written in normal (upright) letters, physical symbols and factors in italics. Formulas shall be consecutively numbered with Arabic numerals in parenthesis (e.g. (1)) at the end of the line.

The number of figures shall be limited to those absolutely necessary for clarification of the text. The same information must not be presented in both a table and a figure. Figures and tables should be numbered separately with Arabic numerals, and should be referred to in the text with clear remarks ("Table 1" or "Figure 1"). Titles, headings, legends and all the other text in figures and tables should be written in both Croatian and English.

Figures should be inserted into the text. They should be of 600 dpi resolution, black and white (color photographs only on request), in jpg or tiff format, completely clear and understandable without reference to the text of the contribution.

All graphs and tables shall be black and white (unless requested otherwise). Tables and graphs should be inserted into the text in their original format in order to insert them subsequently into the Croatian version. If this is not possible, original document should be sent in the format in which it was made (excel or statistica format).

The captions to figures and drawings shall not be written in block letters. Line drawings and graphs should conform to the style of the journal (font size and appearance). Letters and numbers shall be sufficiently large to be readily legible after reduction of the width of a figure or table. Photomicrographs should have a mark indicating magnification, preferably in micrometers. Magnification can be additionally indicated at the end of the figure title, e.g. "Mag. 7500:1".

Discussion and conclusion may, if desired by authors, be combined into one chapter. This text should interpret the results relating to the problem outlined in the introduction and to related observations by the author(s) or other researchers. Repeating the data already presented in the "Results" chapter should be avoided. Implications for further studies or application may be discussed. A conclusion shall be expressed separately if results and discussion are combined in the same chapter. Acknowledgements are presented at the end of the paper. Relevant literature shall be cited in the text according to the Harvard system ("name – year"), e.g. (Badun, 1965). In addition, the bibliography shall be listed at the end of the text in alphabetical order of the author's names, together with the title and full quotation of the bibliographical reference. The list of references shall be selective, and each reference shall have its DOI number (http://www.doi.org) (check at http://www.crossref.org).:

#### Example of references

Journal articles: Author's second name, initial(s) of the first name, year: Title. Journal name, volume (ev. issue): pages (from - to). DOI number.

#### Example:

Kärki, T., 2001: Variation of wood density and shrinkage in European aspen (*Populus tremula*). Holz als Roh- und Werkstoff, 59: 79-84. http://dx.doi.org/10.1007/s001070050479.

Books:

Author's second name, initial(s) of the first name, year: Title. (ev. Publisher/editor): edition, (ev. volume). Place of publishing, publisher (ev. pages from - to).

#### Examples:

Krpan, J. 1970: Tehnologija furnira i ploča. Drugo izdanje. Zagreb: Tehnička knjiga.

Wilson, J.W.; Wellwood, R.W. 1965: Intra-increment chemical properties of certain western Canadian coniferous species. U: W.

A. Cote, Jr. (Ed.): Cellular Ultrastructure of Woody Plants. Syracuse, N.Y., Syracuse Univ. Press, pp. 551-559.

Other publications (brochures, studies, etc.):

Müller, D. 1977: Beitrag zur Klassifizierung asiatischer Baumarten. Mitteilung der Bundesforschungsanslalt für Forst- und Holzwirtschaft Hamburg, Nr. 98. Hamburg: M. Wiederbusch. Websites:

\*\*\*1997: "Guide to Punctuation" (online), University of Sussex, www.informatics.sussex.ac.uk/department/docs/punctuation/node 00.html. First published 1997 (Accessed Jan. 27, 2010).

The paper will be sent to the author in pdf format before printing. The paper should be carefully corrected and sent back to the Editor's Office with the list of corrections made. Each contributor will receive 1 copy of the journal.

Further information on the way of writing scientific papers can be found on the following website:

#### www.ease.org.uk/publications/author-guidelines



# HRVATSKA KOMORA INŽENJERA ŠUMARSTVA I DRVNE TEHNOLOGIJE

Osnovana je na temelju Zakona o Hrvatskoj komori inženjera šumarstva i drvne tehnologije.

Komora je samostalna i neovisna strukovna organizacija koja obavlja povjerene joj javne ovlasti, čuva ugled, čast i prava svojih članova, skrbi da ovlašteni inženjeri obavljaju svoje poslove savjesno i u skladu sa zakonom, promiče, zastupa i usklađuje njihove interese pred državnim i drugim tijelima u zemlji i inozemstvu.

# Članovi komore:

inženjeri šumarstva i drvne tehnologije koji obavljaju stručne poslove iz područja šumarstva, lovstva i drvne tehnologije.

# Stručni poslovi:

projektiranje, izrada, procjena, izvođenje i nadzor radova iz područja uzgajanja, uređivanja, iskorištavanja i otvaranja šuma, lovstva, zaštite šuma, hortikulture, rasadničarske proizvodnje, savjetovanja, ispitivanja kvalitete proizvoda, sudskoga vještačenja, izrade i revizije stručnih studija i planova, kontrola projekata i stručne dokumentacije, izgradnja uređaja, izbor opreme, objekata, procesa i sustava, stručno osposobljavanje i licenciranje radova u šumarstvu, lovstvu i preradi drva.

# Zadaci Komore:

- promicanje razvoja struke i skrb o stručnom usavršavanju članova,
- poticanje donošenja propisa kojima se utvrđuju javne ovlasti Komore,
- reagiranje struke na pripremu propisa iz područja šumarstva, lovstva i drvne tehnologije,
- suradnja s nadležnim institucijama i zastupanje struke u odnosu prema njima,
- organizacija stručnoga usavršavanja,
- zastupanje interesa svojih članova,
- izdavanje pečata i iskaznice ovlaštenim inženjerima,
- briga i nadzor poštivanja kodeksa strukovne etike,
- osiguravanje članova Komore za štetu koja bi mogla nastati investitorima i trećim osobama i sl.

Članovima Komore izdaje se rješenje, pečat i iskaznica ovlaštenoga inženjera. Za uspješno obavljanje zadataka te za postizanje ciljeva ravnopravnoga i jednakovrijednoga zastupanja struka udruženih u Komoru, članovi Komore organizirani su u razrede:

- Razred inženjera šumarstva
- Razred inženjera drvne tehnologije

HRVATSKA KOMORA INŽENJERA ŠUMARSTVA I DRVNE TEHNOLOGIJE Prilaz Gjure Deželića 63 10000 ZAGREB

> telefon: ++ 385 1 376-5501 e-mail: info@hkisdt.hr

www.hkisdt.hr

povezite se s prirodom

# D drzodjelac

Drvodjelac d.o.o. Petra Preradovića 14, Ivanec, Hrvatska +385 (0)42 781 922 | www.drvodjelac.hr